## Quantifying the Effects of Situationally-Induced Impairments and Disabilities on Mobile Interaction

Zhanna Sarsenbayeva

Submitted in total fulfilment of the requirements of the degree of Doctor of Philosophy

Supervised by Dr. Jorge GONCALVES, Prof. Vassilis KOSTAKOS School of Computing and Information Systems The University of Melbourne Author ORCID: https://orcid.org/0000-0002-1247-6036 February, 2020

## Abstract

Situationally-Induced Impairments and Disabilities (SIIDs), also known as situational impairments, have been shown to negatively affect mobile interaction. This is a consequence of the fact that smartphones have become an indispensable part of our everyday life, and are used under various situations, contexts, and environments. While some situational impairments have received more attention from the research community (*e.g.*, walking-, encumbrance-based SIIDs), some remain underexplored. In addition, research conducted on SIIDs has typically followed an ad-hoc approach, with studies aimed at investigating the impact of a particular SIID on a particular task.

Conversely, this thesis systematically quantifies the effects of a range of SIIDs: ambient noise, stress, and dim ambient light on mobile interaction. These findings then enable us to draw baseline comparisons between the effects of these SIIDs on mobile interaction. Furthermore, in a case study this thesis focuses on cold-induced SIIDs, and proposes a sensing mechanism to detect and respond to the onset of such effects.

Our contribution to Human-Computer Interaction (HCI) and UbiComp research is to enhance our understanding of the impact of SIIDs on mobile interaction. This knowledge is crucial to enable the development of smarter ubiquitous technology that can detect SIIDs and adapt mobile device interfaces accordingly with the purpose of improving the user experience for people of all abilities.

## Declaration

This is to certify that

- 1. The thesis comprises only my original work towards the PhD.
- 2. Due acknowledgement has been made in the text to all other materials used.
- 3. Appropriate ethics procedure and guidelines have been followed to conduct this research.
- 4. The thesis is less than 100,000 words in length, exclusive of tables, maps, bibliographies and appendices.

Zhanna Sarsenbayeva

February 2020

## Preface

This thesis has been written at the School of Computing and Information Systems, The University of Melbourne. The major parts of this thesis are in Chapters 4, 5, 6, 7 and 8. Ethics approval for the studies presented in Chapters 4, 5, 6 were obtained from The University of Melbourne Human Ethics Advisory Group. The study presented in Chapter 8 was conducted at the Finnish Institute of Occupational Health in collaboration with the colleagues from University of Oulu, Finland, under local ethical guidelines. Chapters 4, 5, 6, and 8 are based on peer-reviewed publications and I declare that I am the primary author and have > 50% contributions in each of the following publications:

 Zhanna Sarsenbayeva, Niels van Berkel, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. 2018. Effect of Distinct Ambient Noise Types on Mobile Interaction. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 2, 2, Article 82 (July 2018), 23 pages. Ethics ID: 17503501. The University of Melbourne Human Ethics Advisory

Ethics ID: 1750350.1, The University of Melbourne Human Ethics Advisory Group.

 Zhanna Sarsenbayeva, Niels van Berkel, Danula Hettiachchi, Weiwei Jiang, Tilman Dingler, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves.
2019. Measuring the Effects of Stress on Mobile Interaction. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 3, 1, Article 24 (March 2019), 18 pages.

Ethics ID: 1851997, The University of Melbourne Human Ethics Advisory Group.

 Zhanna Sarsenbayeva, Niels van Berkel, Weiwei Jiang, Danula Hettiachchi, Vassilis Kostakos, and Jorge Goncalves. 2019. Effect of Ambient Light on Mobile Interaction. In *IFIP Conference on Human-Computer Interaction* (pp. 465-475) (September 2019). Springer, Cham.

Ethics ID: 1852612, The University of Melbourne Human Ethics Advisory Group.

 Zhanna Sarsenbayeva, Niels van Berkel, Aku Visuri, Sirkka Rissanen, Hannu Rintamaki, Vassilis Kostakos, and Jorge Goncalves. 2017. Sensing Cold-Induced Situational Impairments in Mobile Interaction Using Battery Temperature. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 1, 3, Article 98 (September 2017), 9 pages.

In addition, I would like to acknowledge the following sources of funding that supported my research:

- Google PhD Fellowship in Human-Computer Interaction
- Melbourne Research Scholarship
- Diane Lemaire Scholarship
- University of Melbourne-University of Manchester Research Fund
- Google PhD travel grants
- UbiComp'18 travel grant

## Acknowledgements

I was very lucky to share my PhD journey with people to whom I am grateful and would like to express my appreciation.

First and foremost, I would like to thank my primary supervisor Dr. Jorge Goncalves. I have so many things that I am eternally grateful for, I don't even know where to start. Thank you for your wise guidance throughout my PhD journey. Thank you for being so invested in my research life and academic career. I could share with you my challenges and worries that I was going through and always received support that helped me strive through these difficulties. I honestly could not have dreamt of a better supervisor. Thank you for always being there for me (even when you were not attending my talks, but asked people to record them to watch later :)). Thank you for your sharp thinking and insightful reflections and thought-provoking questions that have expanded my mind and thanks for challenging me by involving in various research activities and projects. They definitely made me a better researcher and a better person. Thank you for being not only a great supervisor, but also a great friend.

Then of course my infinite gratitude goes to my secondary supervisor Professor Vassilis Kostakos. Thank you for the opportunity to do research with you. Thanks for taking me on board with you in your Australian adventure. Thank you for our insightful discussions and inspiring meetings. Thank you for believing in me, as it all started back in Oulu, Finland when you hired me as a Research Assistant. Thank you for your endless support and optimism. I am forever grateful.

Next, I would like to express my appreciation and gratitude to my good friend Niels van Berkel (Dr. van Berkel now). Niels, you have been my main support as well as the major source of joy and fun during my PhD journey. Thank you for always being there for me and helping to go through the challenges I encountered in my research and personal life. You have been there with me when we moved from Finland to Australia, you have been there with me through the accepts and major revisions, and you have been there with me when travelling for conferences. My PhD life would have been empty without your presence in it. Thank you from the bottom of my heart :)

I would also like to thank Dr. Eduardo Velloso, Dr. Tilman Dingler, and Dr. Benjamin Tag for their support and advice regarding my research and career. Eduardo, thank you for your optimism, kindheartedness, and for the insights I received during our discussions and ideation. Thanks for the morning maths, stats, research and other knowledge quizzes that you engaged us while getting our morning coffee. Tilman, thanks for being such an endless source of energy, enthusiasm and encouragement which motivates us all in the lab. It is impossible to be sad or lazy when you are around. Thank you, Ben, for being not only a great colleague, but a great friend with whom I could share my secrets and concerns. Your friendship and support mean a lot to me. Thank you all for your support and kindness.

Then, I am deeply grateful to Prof. Timo Ojala from the University of Oulu for his support in my career and of course for creating the International Master's Degree Programme in Computer Science and Engineering which was the start of my research career and defined my further academic path. Thank you for your kindness and support that you provided me with during my studies and research at the University of Oulu. I am honoured to have worked in your research center.

I would also like to thank my colleagues in Finland who witnessed initial steps of my academic career – Dr. Simo Hosio, Dr. Denzil Ferreira, Dr. Aku Visuri, Dr. Simon Klakegg, Dr. Susanna Pirttinkangas, Dr. Heli Koskimäki, Professor Hannu Rintamäki, and Dr. Sirkka Rissanen. Thank you, Simo, for your support, kindness and encouragement throughout my PhD. Thank you for your sincere happiness for my success. Thank you, Denzil, for your insightful discussions, enthusiasm, and motivation. Thank you, Aku and Simon, for your friendship and fun that we have shared back in Oulu. Also thanks for providing us with accommodation at UbiComp'17 :) Big thanks to Susanna and Heli for being great conference companions! Thank you Professor Hannu and Sirkka for your kindness and warm welcome to your lab during our collaborative work in the cold chamber! I love you all for your kind hearts and sincerity.

During my PhD I was lucky to work with academics from other labs outside Australia and I would like to thank them for collaborating and working with me on different projects. My thanks goes to Dr. Sarah Clinch and Prof. Simon Harper for organising such a warm and welcoming visit for me in their lab at the University of Manchester during my research visit in 2018. I would also like to thank my friends that I got to know in their lab: Olu, Julia, Alan, Aitor, Julio, Deemah, and Alaa. Thanks guys for making my stay in Manchester pleasant and fun!

I would also like to thank Prof. Hans Gellersen for hosting me at Lancaster University. It was such an inspiring experience to visit your lab and learn about your projects.

Then I would like to thank my dear friend and colleague Katerina Mangaroska. Thank you, Katerina, for your care and support you provided me with when it was needed. Thank you for always being there for me. Thank you for introducing me to Prof. Dragan Gasevic to whom I am deeply grateful for his guidance and advice regarding my career in research and academia. Thank you, Dragan, for encouraging us to believe in ourselves and strive higher.

Next, I would like to thank Dr. Garreth Tigwell for being a great co-organiser of our workshop at CHI'19 and for your help to arrange my visit and stay in the UK to be as smooth as possible. My gratitude goes to Prof. Annalu Waller OBE and Dr. Rachel Menzies for organising my visit to the University of Dundee. I would also like to thank Prof. Yeliz Yesilada, Prof. David Flatla, and Dr. Benjamin Gorman for co-organising a great workshop at CHI'19 with me and Garreth. I am happy and grateful that the workshop at CHI brought us all together.

I am also grateful to Prof. Jacob Wobbrock for not only being a great coorganiser of the same workshop at CHI'19, but also for the inspiration of my PhD topic. Your paper "The future of mobile device research in HCI" inspired me to define the topic of my PhD when I was just about to complete my Masters degree. Since then you have been my research role model and someone to look up to when I needed to seek for motivation. Thank you.

Next, I would like to thank the faculty and staff of IDL and CIS at the University of Melbourne. Thank you Prof. Frank Vetere for warm welcome in IDL when we first joined, Assoc. Prof. George Buchanan for the opportunity to serve as an SV Chair at OzCHI'18, Dr. Nicole Barbee for keeping IDL running and for taking care of everyone in the lab, Dr. Greg Wadley, Dr. Jenny Waycott, Dr. Dana McKay, Dr. Melissa Rogerson, Dr. Steven Baker, Dr. Ryan Kelly, Julie Ireland, Rhonda Smithies, Emma Russo, Imbi Neeme for a pleasure to work with you all.

Then I would like to thank my peers and friends in IDL and CIS at the University of Melbourne. Thank you Chu for also being there with me during our move from Finland to Australia. Thanks to Danula, Gabriele, Weiwei, Chaofan, Senuri, Brandon, Martin, Difeng, Kangning, Romina, Sarah, Yousef, Jing, Namrata, Henrietta, Ebrahim, Xin, Hank and many others for sharing fun times during our lunches, dinners, deadlines, trips, and parties. Special thanks to Joshua and Qiushi for helping me to move from one place to another and for assembling my furniture :)

I would also like to thank everyone that I have met at summer schools and conferences and became friends with. Thank you Daniel, Sven, Phillip, Crystal and Callum, Yomna, Florian, Hristian and Martin, Lars, Vito and many others whom I forgot to mention. Also sorry for omitting your "Dr." titles :)

I would also like to thank my friends Anna, Donia, Mitra, Nanci, Esmeralda, Kasia, Iiulia, Inkar, Bakytzhan, Gulsum, Samal, Akbota, Kuralai, Karla, Kamil, Alina, Saniul, Nat, Myran, Shanon, Bibigul, Arian, Artur, and many others for their support and joy we shared together.

Finally, I would like to thank my mom and dad, and my sisters for their unconditional love and tremendous support that they have been providing me with throughout my life. Thank you for supporting me in every step I take, even if it takes me to the other side of the world far away from you. I miss you dearly and I love you.

# Contents

List of tables							
List of figures							
1	Intr	oductio	on	1			
	1.1	Motiv	ration	1			
	1.2	Contr	ibution	3			
		1.2.1	Research Questions	3			
		1.2.2	Author's Role and Contribution	4			
	1.3	Thesis	s Outline	5			
2	Background						
	2.1	Acces	sibility of Mobile Devices	7			
		2.1.1	Visual Impairments	9			
		2.1.2	Motor Impairments	16			
		2.1.3	Hearing Impairments	18			
		2.1.4	Cognitive Impairments	20			
	2.2	Situat	ionally-Induced Impairments and Disabilities	21			
		2.2.1	Understanding	22			
		2.2.2	Sensing	27			
		2.2.3	Modelling	30			
		2.2.4	Adapting	32			
		2.2.5	Additional Design Guidelines	37			
		2.2.6	Summary	40			
3	Met	Methodology					
	3.1	System	natic Approach towards Understanding the Effects of SIIDs	45			
		3.1.1	Smartphone Tasks	46			
	3.2	Data 4	Analysis	47			
		3.2.1	Quantitative analysis	47			
		3.2.2	Qualitative Analysis	48			

	3.3	Ethical Considerations	49			
	3.4	Limitations	50			
	3.5	Conclusion	52			
4	Qua	ntifying the Effects of Ambient Noise on Mobile Interaction	53			
	4.1	Publication	54			
5	Qua	ntifying the Effects of Stress on Mobile Interaction	79			
	5.1	Publication	80			
6	Qua	Quantifying the Effects of Dim Ambient Light on Mobile Interaction				
	6.1	Publication	100			
7	Con	trasting the Effects of Different SIIDs on Mobile Interaction	111			
	7.1	Target Acquisition Task	112			
	7.2	Visual Search Task	113			
	7.3	Text Entry Task	114			
	7.4	Implications for Design and Summary	115			
8	Sen	sing Cold-Induced SIIDs	117			
	8.1	Publication	118			
9	Discussion					
	9.1	RQ1. What are the effects of different SIIDs on mobile interaction?	129			
		9.1.1 Ambient Noise	129			
		9.1.2 Stress	131			
		9.1.3 Dim Ambient Light	131			
		9.1.4 Summary	132			
	9.2	RQ2. How do these effects compare to each other?	132			
	9.3	RQ3. How can mobile devices sense the onset of cold-induced SIIDs?	?133			
	9.4	Future Directions for SIID Research	134			
10	Con	clusion	137			
Bil	Bibliography					

# List of tables

2.1	Summary Table of Situationally Induced Impairments and Disabil-					
	ities in Terms of Their Understanding, Sensing, Modelling, and					
	Adapting	41				
7.1	Comparison of the Effects of SIIDs against Respective Baseline	116				

# List of figures

3.1 Interface of the application with Target Acquisition Task (A), Visual Search Task (B-C), and Text Entry Task with user's input (D-E) . . 47

## Chapter 1

## Introduction

### 1.1 Motivation

The general term of *Situationally-Induced Impairments and Disabilities* (SIIDs), also known as *situational impairments*, refers to the relationship between the user, the surrounding situation\context\environment, the task the user is engaged in, and the technology used to complete this task [242, 243]. Wobbrock clarifies the differences between *Situation, Context*, and *Environment* by specifying that the situation refers to the "immediate circumstance" of the user; while context encompasses the broader narrative that includes the user's engaged activities, purposes, goals, and motivations behind these activities [270]. Finally, environment withholds a larger setting, including both the physical and social dimensions [270]. Based on these clarifications, Wobbrock states that the notion of SIIDs comes from the "conviction of situational impairments" – *functional limitations experienced by a user in a specific circumstance*, and from the concept of "situational disabilities" – *task or activity limitations experienced by a user in a specific circumstance* [270].

In this thesis the terms SIIDs and situational impairments are used interchangeably, and are considered in the context of *mobile interaction* on a smartphone. The ubiquitous nature of mobile devices brings new challenges to research on SIIDs as people use them under different situations, contexts, and environments. For example, it is common for modern smartphone users to interact with their devices on-the-go, under the bright sunlight or a dim environment, outside in a noisy street in winter, and even perhaps while carrying luggage or a handbag in one hand. However, this flexibility comes at a cost to users' cognitive, perceptual, motor, and social abilities [270]. The experience of situational impairments during mobile interaction applies to users of all abilities [270]. In other words, both able-bodied and permanentlyimpaired user groups are negatively affected by SIIDs during mobile interaction. Further, several researchers mention that SIIDs might have temporary detrimental effects on user abilities during mobile interaction similar to the effects of permanent health-related impairments [179, 274]. For example, Yesilada *et al.* demonstrated in their work that a situationally-impaired user without physical impairments performed a similar number of errors on a mobile device as a user with physical impairments [279].

Furthermore, several studies report that situational impairments aggravate the negative mobile interaction experience for permanently-impaired users [1, 112]. For example, visually-impaired users found it challenging to use their smartphones while walking, carrying bags, being in a moving vehicle, being exposed to bright/dim lights or inhospitable weather conditions, navigating in an unfamiliar environment, or while multitasking [1, 112]. Motor-impaired smartphone users also found it challenging to interact with their device while being situationally-impaired [175]. To be precise, motor-impaired users identified challenges to interact with their devices while "on-the-go", due to restrictive clothing, and inhospitable weather conditions [175]. Hence, finding solutions to accommodate SIIDs during mobile interaction will benefit both able-bodied users and users with permanent impairments.

In order to investigate SIIDs, research on the topic has been conducted in four main areas: Understanding, Sensing, Modelling, and Adapting [256].

- **Understanding** focuses on investigating and establishing the effects of SIIDs on mobile interaction.
- **Sensing** concerns itself with designing and building mechanisms that detect SIIDs.
- **Modelling** is directed to model the user, user behaviour, and/or the environment that might occur during SIIDs.
- Adapting aims at creating adaptive interfaces to mitigate the effects of SIIDs.

We consider *Understanding* to be the quintessential step that empowers conducting successful research in SIIDs. It is crucial to first understand the exact effects that contextual factors have on mobile interaction. This knowledge can provide the necessary evidence demonstrating if there is a need to build sensing mechanisms, and how to apply modelling and adapting techniques to accommodate these situational impairments, without wasting unnecessary resources. Not only is it important to establish the effects of various SIIDs on mobile interaction, but it is also crucial to be able to compare these effects between each other [235]. By doing so, we can contribute to accumulating knowledge within a specific research topic – an important challenge for HCI and UbiComp research [140, 141]. As a result, this research is able to provide further prospects for designers and researchers to prioritise the accommodation of different SIIDs based on the user's current task. A fair amount of research has been conducted to study and understand the impact of SIIDs on mobile interaction; however, the majority of the studies follow an ad-hoc approach and lack systematic investigation. This results in the inability of the research to draw a comparative judgement of the magnitude of the effects of different SIIDs on mobile interaction.

Furthermore, despite the established adverse effects of certain SIIDs (*e.g.*, cold ambience [231]), researchers have not proposed an adequate sensing mechanism to detect these SIIDs. The absence of such a sensing mechanism that detects the presence of SIIDs, results in the inability to further adapt the interface to accommodate these effects.

## **1.2 Contribution**

### **1.2.1 Research Questions**

Based on the research gaps identified above, we have identified three research questions addressed in this thesis:

- RQ1. What are the effects of different SIIDs on mobile interaction?
- RQ2. How do these effects compare to each other?
- RQ3. How can mobile devices sense the onset of cold-induced SIIDs?

The contributions of this thesis are three-fold and were explored via four different user studies presented in this thesis. First, we contribute to and consolidate the knowledge in understanding the effects of underexplored SIIDs on mobile interaction, namely ambient noise, stress, and dim ambient light, by conducting a systematic investigation. These SIIDs are commonly present in users' everyday settings and have recently been established as important SIIDs that lack investigation [235]. Understanding the effects of underexplored SIIDs enables creation of sensing, modelling, and adapting mechanisms to accommodate particular SIIDs avoiding misallocation of resources (*e.g.*, human, financial, technological).

Second, we provide a baseline comparison of these effects by employing a systematic methodology discussed in detail in Chapter 3. Namely, the study protocol presented in this thesis re-used the same observational, scoring, and calculation rules to measure user performance in three common mobile interaction tasks: target acquisition, visual search, and text entry that were re-used across multiple user studies (Chapters 4, 5, 6). This methodology could be used by other researchers to investigate the effects of underexplored SIIDs that were not studied within the scope of this thesis. Finally, we propose a sensing mechanism to detect the onset of one type of SIIDs – cold-induced SIIDs – using smartphone's built-in battery sensor, and demonstrate how off-the-shelf smartphone can be used to achieve this goal.

### 1.2.2 Author's Role and Contribution

Four publications are included as main contributions in this thesis presented in Chapters 4, 5, 6, and 8. These articles are published in prestigious, international and peer-reviewed conferences/journals in the field of Ubiquitous Computing and Human-Computer Interaction [233, 234, 236, 237].

I would like to clarify my contribution in the presented work. I was the lead author for the above-mentioned publications and performed the majority of the work (more than 50%). Precisely, I introduced the ideas behind the work, initiated the process, prepared the experimental designs, and handled the ethical bureaucracy, software development, participant recruitment, and data analysis. Moreover, I prepared the articles for submission and managed the revision process after receiving peer-review feedback. My co-authors provided me with feedback and suggestions on the design of the studies and data analysis techniques. Furthermore, my co-authors contributed to the write-up of the publications. To reflect this contribution and show appreciation to my co-authors, the scientific term "we" is used throughout the main chapters of this thesis.

### **1.3 Thesis Outline**

This thesis is organised as follows. Chapter 2 provides an overview of related work conducted in this research area.

Chapter 3 outlines the methodology followed throughout the studies presented in this thesis. This chapter provides the motivation behind the methodological decisions used for data collection and data analysis as well as highlights the limitations of the employed methodological approach.

Chapters 4, 5, 6 present original articles focused on quantifying the effects of ambient noise-, stress-, and ambient light-induces SIIDs on common smartphone tasks performed during mobile interaction: target acquisition, visual search, and text entry.

Chapter 7 contrasts the effects of the SIIDs presented in Chapters 4, 5, and 6 on mobile interaction. This chapter draws conclusions based on this comparison and discusses the applicability and importance of this knowledge.

Chapter 8 presents a sensing mechanism to detect cold-induced SIIDs. This chapter suggests using smartphone battery temperature as a gauge to infer user's finger temperature when user is exposed to cold environments. Hence, smartphone's built-in battery temperature can be used to detect cold-induced SIIDs.

Chapter 9 discusses the contributions presented in this thesis in relation to existing research. It also outlines the limitations of the work presented in this thesis and summarises future research directions on SIIDs with suggestions. Chapter 10 provides final remarks and concludes this thesis.

## Chapter 2

## Background

This chapter provides a review of background work of research conducted in the area of Accessibility and SIIDs with a focus on mobile devices.

### 2.1 Accessibility of Mobile Devices

Mobile devices play an important role in people's everyday life [112]. Smartphones and other personal devices are prevalently used for communication purposes, to access information on-the-go, to perform work-related activities, and to stay in touch with family and friends via social media [135–137,199,247]. Mobile devices have expanded the capabilities of users, in particular by providing them with increased independence and freedom in their daily lives [54].

Despite the fact that mobile devices have also been used to assist people with special needs [54, 112, 152], off-the-shelf smartphones might often not be accessible to people with permanent impairments (*e.g.*, motor and visual impairments in particular) due to several design constraints (*e.g.*, including inability to access the content of the screen, small buttons, limited user input methods among others) [112]. Although these accessibility problems have been established and acknowledged [152, 266], off-the-shelf smartphones are still being primarily designed for able-bodied users without sufficient consideration of the wide breadth of accessibility issues [112].

Kane and colleagues suggest that extending the accessibility features of smartphones should be highly prioritised [112, 172, 245], particularly because people with disabilities use mass-market devices to avoid the social stigma [200, 244] as well as because there will always be a need for assistive technology as individuals with severe disabilities will require assistance either from the technology or a human [172]. Shinohara *et al.* argue that we, as a society, might be able to alleviate social misperceptions of disability by making mainstream technology more accessible [244]. Moreover, Wobbrock argues that the experience of disability in mobile interaction applies to us all to some extent [270], because our abilities are not static and change according to the context [179]. This also includes but is not limited to the effects of SIIDs on both able-bodied and permanently-impaired users. Therefore, the systems should be created to fit the abilities of the users and should follow adaptable and universal design strategies [75, 274].

Adaptable design implies including modifications to a standard design with the purpose of making the design usable according to individual's needs. Universal design sometimes employs adaptable strategies, meaning that the universal design can be adaptable, but should always be accessible [172]. The goal of accessible computing is to improve independence, access, and quality of life for people with disabilities [274]. However, this goal can be expanded to also improve user experience for able-bodied people who are temporarily impaired due to SIIDs [270].

In this section we focus on visual, motor, hearing, and cognitive impairments as they are commonly present among general population and have been extensively investigated in the context of mobile interaction. Furthermore, these four permanent impairments can be mapped to commonly acknowledged situational impairments in terms of their effects on mobile interaction. For example, the effects of permanent visual impairments can be similar to the effects of bright or dim light on mobile interaction, while the effects of cognitive impairments on mobile interaction can to some extent resemble the effects of stress. Similarly, the effects of motor impairments on mobile interaction might be similar to the effects of cold-induced SIIDs, as well as the effects of hearing impairments can be similar to the effects of ambient noise in mobile interaction [268]. Therefore, solutions suggested for accommodating the effects of permanent impairments can be used to address the effects of SIIDs to benefit the users of all abilities.

### 2.1.1 Visual Impairments

Visual impairments are widely prevalent around the world with a total number of 2.2 billion people suffering from this condition [195]. In the United States alone the number of visually impaired people exceeds 8 million [4]. As smartphones are everyday ubiquitous devices, they are widely used among visually impaired or blind users [188,266]. Nevertheless, the accessibility of conventional mobile devices for visually-impaired users is restricted [86,112]. For example, Li *et al.* found that visually-impaired users often need to have access to the information (*e.g.*, calendar and contact information) during phone calls and find it problematic as mobile phones have limited accessibility to accommodate this need [133].

In order to increase the accessibility of current mobile devices for visuallyimpaired users, literature suggests several methods that include the creation of novel interaction techniques and integration of novel technology to the existing devices [86,112]. Nonetheless, Brady *et al.* in their study demonstrate that even though visually-impaired participants were, in most cases, willing to experiment with novel technology, they tend to quit and stop using the technology if their user experience was poor [27]. This example demonstrates that the usability plays a key role in adoption of the assistive technology [27].

Furthermore, Abdolrahmani and colleagues note that the experience of visuallyimpaired mobile users might be worsened by SIIDs due to cognitive load, as their attention will need to be divided in performing parallel tasks: between swiping the cane and interacting with the mobile device [1]. Leonard *et al.* argue that mobile phones can improve user experience of visually-impaired users in way-finding, memory recall and communication, but only if the effects of context and visual ability are adequately accounted for [131]. Therefore, it is important to account for situationally-induced impairments and disabilities when designing technology to assist visually impaired users [270].

Visually-impaired users encounter several challenges when using their mobile devices, including but not limited to accessing the content of the screen, inputting information on the devices (*e.g.*, text), protecting their input, and navigation [112]. To solve the aforementioned issues, researchers have suggested using voice [94, 194] and gesture [40, 111] modalities to interact with the device. Furthermore, a fair amount of research suggested several text entry techniques to improve the user experience in a text entry task [15, 25].

Moreover, researchers also considered solving the problem of privacy protection for visually-impaired users during mobile interaction [4,14]. Finally, researchers also provided solutions on mobile devices for navigation and wayfinding problems to support visually-impaired users [12,228]. The following sections summarise the existing literature in terms of voice-based interaction, gesturebased interaction, and existing text entry, privacy and security, and navigation solutions to support visually-impaired mobile device users.

#### **Voice Interaction**

One of the primary issues encountered by visually-impaired mobile device users is accessing the content of their screen [112]. Currently, the most common way for visually-impaired users to read the content of their screens is by using mobile screen reader. The use of these devices has undergone a significant growth in the past few years [63] even though voice interfaces were proposed as a solution to this issue back in the late 80s [94]. For example, Hill and Grieb developed "Touch 'n Talk" interface to assist visually-impaired users to perform computerbased tasks [94]. The authors show that "Touch 'n Talk" was well accepted by visually-impaired users and improved their performance in editing tasks and accessing menus as compared to key-based talking terminals [94]. Furthermore, O'Neill *et al.* developed a system with gestural input and voice output [194]. The researchers demonstrate that interacting with the computers without GUIs is possible and quicker than interacting with visual interfaces.

Pirhonen and colleagues took this approach one step further and implemented it on mobile devices [214] as an interface for a mobile music player – "Touch-Player". The authors show that the usability performance of their participants significantly improved when using gesture and audio based interfaces when compared to a visual interface [214]. Similarly, Zhao *et al.* proposed "Earpod" – a system for mobile devices enabling eyes-free menu selection using touch input with a synchronously linked audio feedback [282]. Their results also demonstrate the feasibility of the system for non-visual interaction with mobile devices. The system was comparable to an iPod-like visual menu in selection accuracy; however, was slower in terms of efficiency. Nevertheless, the authors argue that this issue is solved by providing the users with extensive training [282]. In addition, text-to-speech is considered to be a popular interaction method for visually-impaired users to interact with their devices [110, 176]. For example, Sánchez and Aguayo presented a mobile messenger for the blind users [227]. The authors used a custom-built keyboard for input and integrated text-to-speech functionality for output in the messenger. Their results show that visuallyimpaired participants were satisfied with system and easily adapted to the input and output modalities of the messenger [227]. However, using text-to-speech functionality for these purposes has several potential issues that currently remain unresolved. For example, the users themselves do not feel comfortable to use voice interaction with the device to avoid attracting attention, and other social and safety reasons [245]. In addition, the presence of ambient noise causes difficulties to use a talking device for visually impaired participants [112].

#### Gestures

Another issue that has been acknowledged for visually-impaired or blind users is the usage of touch screen to interact with the device, as touch screens might constrain the device's accessibility [111]. The limitations of smartphones include but are not limited to restricted input, small screen size, and non-intuitive interfaces [247]. As a result, early work in this area suggested to provide feedback for touch screen interaction [40]. In particular, it was suggested that the accessibility of touch screens can be enhanced by providing haptic feedback [259] or by creating new non-visual interaction techniques [111] including eyes-free interaction and gestures [214]. Moreover, Pirhonen *et al.* suggest that the gestures should be reliable to be performed on-the-go to avoid interaction by mistake [214].

For instance, Kane and colleagues developed "Slide Rule" – a non-visual interaction technique to provide access for blind and visually-impaired users to a custom-developed phone book, email client, and media player [111]. "Slide Rule" utilises gesture interaction to achieve goals. Their results demonstrate that visually-impaired users completed the tasks significantly faster when using "Slide Rule" compared to when using button-based mobile screen reader. Furthermore, Azenkot and colleagues introduced "DigiTaps" – an eyes-free interaction technique with minimal audio feedback that enables visually-impaired users to interact with their mobile devices with little to no auditory attention [11]. This technique enabled blind participants to use gestures to input numbers on a smartphone at a relatively high entry rate and accuracy [11].

In addition, Ruamviboonsuk and colleagues presented "Tapulator" – a nonvisual calculator to support visually-impaired users [223]. The system uses gestures instead of buttons and was universally designed to support any user who requires non-visual interaction with a calculator. In their preliminary evaluation, the authors show that the error rate was lower when using "Tapulator" as compared to a standard calculator application. Hence, the authors demonstrate that "Tapulator" can benefit blind and low-vision users in terms of speed and accuracy when compared to a traditional calculator application installed on smartphones [223].

Previous work has also suggested using devices' built-in sensors and user's spatial information to enhance the accessibility of mobile devices [132]. For example, Li *et al.* introduced "Virtual Shelves" – an interaction technique used to improve the user experience of visually-impaired users during mobile interaction [132]. The technique uses *proprioception*, the sense of position and orientation of one's body parts with respect to each other, and spatial memory for interaction [132]. The authors utilised the 3-axis accelerometer and the 3-axis gyroscope of a smartphone to determine the orientation and movement of the device for completing selection tasks and found that participants were correct in 81.8% of these tasks. Furthermore, the authors introduced shortcuts for the participants to be used for calling contacts, location-based tasks, email and weather checking tasks. They demonstrate that 88.3% of shortcuts were launched correctly. This work demonstrates how proprioception can be used to improve accessibility of mobile devices for the visually-impaired [132].

Furthermore, Ye *et al.* suggested using wearables to extend mobile interaction for visually-impaired users [278]. In their study the authors found that visually-impaired participants positively accepted the use of a wristband and a ring sensor to enable eyes-free interaction. The authors also suggest utilising a greater variety of tactile feedback to extend the interaction technique [278].

Similarly, Amar and colleagues implemented "ADVICE" – a handheld device that utilises a combination of tactile and audio feedback to assist visuallyimpaired users [6]. Their results demonstrate that the visually-impaired participants desire to use devices with the same functionality as their sighted peers [6]. Therefore, it is important to provide accessibility of commonly used mobile device's features. One such example could be photography, especially as earlier work has revealed that visually-impaired users found assistive photography for blind appealing and useful [260]. Therefore, several researchers developed assistive applications to support non-visual photography [3,88,103,260]. Adams and colleagues investigated the requirements for building an application to assist blind photography and found that visually-impaired users had difficulties to operate with traditional smartphone camera applications [3]. As a result, the authors implemented a set of gestures that acted as shortcuts to operate with the picture taking application to reduce the complexity of operation. Moreover, visually-impaired users complained that it was challenging to identify context of the images (*e.g.*, time, place, date); hence the application had to save the metadata when the picture was captured together with the ambient audio and voice memo that can be recorded by the user [3].

In addition, Jayant and colleagues introduced "EasySnaps" – a photography application assisting visually impaired that provides an audio feedback [103]. The authors show that audio feedback improved the quality of photographs. These studies helped to identify requirements for creating photo-taking applications for visually impaired users and opens opportunities for design guidelines to support such applications for visually-impaired users.

#### **Text Entry**

Text entry is another challenging task for visually-impaired users to perform on a mobile device [41, 112]. Therefore, a significant amount of research has been conducted to improve text entry for visually impaired users [13, 15, 25]. For example, Bonner and colleagues present "No-Look Notes" – an eyes-free text entry system with a gesture-based multitouch input and a voice output [25]. The system arranges the characters on an 8-segment pie menu on the screen, and when the user touches each segment, the system announces the characters audibly in that segment. The user can select a desired character by dropping their second finger on the screen. The results of the system evaluation demonstrate that the user error rate was significantly lower when using "No-Look Notes" as compared to the VoiceOver technique; however, the text entry rate was lower when using "No-Look Notes" when compared to the VoiceOver technique [25].

Furthermore, Azenkot and colleagues present "Perkinput" – touchscreen text entry technique that uses multipoint touches, where each finger touch is represented as a bit [15]. The evaluation of this technique showed that blind users were faster and more accurate when using "Perkinput" for text entry as compared to Apple's voiceover [15].

In another approach, several research projects were based on integrating Braille technique for text entry to support blind participants due to their familiarity with the technique [80]. For example, Frey *et al.* present "BrailleTouch" – eyes-free text entry application for mobile devices to assist visually-impaired users [64]. The application has 6 buttons in total and allows multitouch input using both hands. Each alphabet letter is mapped to a combination of soft buttons pressed according to the Braille alphabet. The evaluation of the system showed that BrailleTouch had a potential to be used as an input technique for touchscreen devices [64]. Mascetti *et al.* also used the Braille alphabet for touch input on touchscreen devices [156]. However, unlike Frey *et al.* [64], the authors utilised gestures instead of pressing the buttons on the screen.

Futhermore, Oliveira *et al.* also used the Braille alphabet as a base for their system – "BrailleType" – to support text entry for visually-impaired smartphone users [193]. Unlike the work presented previously [25,64], "BrailleType" does not support multitouch gestures and favours single touch gestures. The authors evaluated performance of "BrailleType" in comparison with Apple's VoiceOver system and demonstrated that although participants were slower using the system, they were significantly less error prone. Their findings suggest that "BrailleType" had a smoother adaptation than other complex methods [193].

In addition, Oliveira *et al.* in their follow-up study evaluated four different types of non-visual text entry methods – "QWERTY", "NavTouch", "MultiTap", and "BrailleType" – with blind participants [192]. The "QWERTY" method consisted of the traditional computer keyboard in tandem with a screen reading software. The "NavTouch" method was based on gestures navigating alphabet horizontally and allowing users to perform gestures on any part of the screen. The "MultiTap" keyboard was based on the keypad-based device. Finally, "BrailleType" was based on the Braille alphabet with audio feedback. The results of their study show that the efficiency of each method depends on the user's individual abilities. In particular, the authors reveal that users' spatial abilities, pressure sensitivity, and verbal IQ levels determine users' performance and can be used to determine the most suitable method for each particular user [192].

#### **Privacy and Security**

Another challenge that visually-impaired users face while interacting with their smartphones is privacy and security [4]. While sighted people are able to monitor the surrounding context and able to protect their vulnerable information during mobile interaction in public, visually-impaired users do not have such a privilege [4]. Azenkot *et al.* report in their work that visually-impaired mobile users were not even remotely aware of security and privacy threats during their interaction with smartphones [14].

To tackle this issue, Li *et al.* built "BlindSight" – an application for visuallyimpaired users to perform phone calls [133]. Unlike traditional visual in-call interface, "BlindSight" allows interaction with the keyboard and responds to auditory feedback. The authors report general preference of the visually impaired users for the application over traditional smartphone interface [133].

#### Navigation and Wayfinding

Navigation and way-finding are acknowledged as major issues that visuallyimpaired users experience [112]. For example, Kane *et al.* demonstrated that blind participants would use their smartphones while walking only if they are in a familiar area [112]. As a result, great deal of research projects have focused on improving navigation and way-finding experience of visually-impaired users with the help of a smartphone.

Hence, to assist visually-impaired users in wayfinding, Azenkot *et al.* present smartphone feedback methods "Wand", "ScreenEdge", and "Pattern" [12]. The authors used haptic feedback to give routing instructions to blind users on a predefined path. The "Wand" technique implied using the smartphone as a wand, and when the top of the phone pointed in the correct direction, the phone would vibrate. The device's built-in compass was used to determine the top of the device. The "ScreenEdge" entailed touching the edges of the device, and when the user touched the correct edge, the device would vibrate. The "Pattern" technique used the pattern of vibration pulses to indicate the correct direction. The evaluation of these techniques showed that the three methods were viable to give wayfinding instructions without demanding user's auditory attention and opens opportunities for creating new interaction techniques to support visually-impaired users [12].

Finally, Narasimhan and colleagues presented a system to enhance independence of blind smartphone users while travelling, but used text-to-speech modality to provide information to the users [176]. In addition, Sánches and Maureira have developed a system for mobile devices to navigate in the subway [228]. The program allowed users to plan the trip beforehand and the participants found it useful. Furthermore, the researchers demonstrated that the system provided the participants with an increased independence of movement in the subway network [228].

### 2.1.2 Motor Impairments

Musculoskeletal disorders of upper limbs are a common ailment that is the leading cause of disability [169]. The disorder includes but is not limited to such diseases as essential tremor and arthritis. The disorder is prevalent around the world with 7 million people in the US suffering from essential tremors [142], whereas, for example, in Australia every fifth person (age over 65) suffers from this condition [202]. The prevalence of arthritis has grown by 75% from 1990 to 2013 [30], and currently 15% of the Australian population suffers from arthritis according to Australian Bureau of Statistics [191]. As the interaction with mobile devices is prevalently designed to use hands, people who are diagnosed with musculoskeletal disorders of upper limbs and other motor impairments, experience difficulties when using their smartphones. For example, prior research has shown that upper limbs disorders are associated with missed targets, longer interaction times, lower text entry performance, which then lead to frustration and stress [119,215,275].

Research on designing and building assistive technology for motor-impaired has risen significantly in the past decades. Assistive technology started with a focus to improve usability experience when interacting with PCs including but not limited to input techniques [118, 119], text entry techniques [181], and one-handed keyboards [148]. For example, Myers *et al.* presented a software that allowed motor-impaired people to use handheld device as a substitute for a keyboard and a mouse for interacting with a computer [174]. The authors report that their system was well accepted by the motor impaired users and was considered to be less tiring as compared to the conventional keyboard and a mouse combination [174]. In another example, Trewin *et al.* show in their work that users with Parkinson disease require more time when completing target selection tasks on a stationary desktop, as well as they produce more errors compared to their able-bodies peers [257].

Nevertheless, the creation of touch devices and interfaces have benefited motor-impaired participants significantly, as touch screens do not require their users to have physical strength and/or dexterity that are needed to press physical buttons [82,174]. However, the location on the screen should be taken into account when designing technology for motor-impaired users, as they prefer selecting targets located in the center of the screen as it is easier and less error prone [203, 206]. For example, Guerreiro and colleagues investigated how tetraplegic users perform taps on a touch screen [83]. The authors found that the target size of 12mm was the most suitable for selection by motor-impaired users. The authors also found that the location of the target on the screen as well as the edge of the device had a significant effect on the performance of tetraplegic users. The researchers advice using these details when designing assistive technology for motor-impaired users [83].

In a follow-up study, Guerreiro *et al.* evaluated the most common interaction techniques – "Tapping", "Crossing targets", "Exiting", and "Directional Gestures" – with motor-impaired users [82]. The findings of their study suggest that although accuracy and precision depended on target size, overall "Tapping" was found to be most effective and preferred amongst participants with tetraplegia [82]. Moreover, Nicolau *et al.* extended this work by comparing the performance of both able-bodied and motor-impaired users on the aforementioned techniques, and show that "Tapping" and "Crossing" can be performed by both able-bodied and motor-impaired groups of users; however, the latter would have a higher error rate [187]. Nevertheless, "Directional gestures" have been shown to be suitable for able-bodied users, but completely inadequate for motor-impaired users. The authors also mention that the target size and position on the screen had an effect on performance of both user groups [187] as well as the posture when holding the device [188].

Similarly, Froelich and colleagues proposed "Barrier Pointing" – a system that uses a mobile device screen's edges and corners to provide greater stability for motor-impaired users in order to improve their accuracy [65]. The authors evaluated their system with motor-impaired participants and found that "Barrier Pointing" technique improved performance of motor impaired participants in target acquisition tasks. Furthermore, the authors suggest integrating haptic feedback into the system to further improve the performance of motor-impaired users in target selection tasks [65].

Text input on mobile devices are considered to be challenging for motorimpaired users. Therefore, there was a need to create alternative text entry techniques to enhance the accessibility of mobile devices. For example, Wobbrock and colleagues introduced "Edgewrite" – an alternative text entry method with a custom alphabet that uses unistrokes [275]. The authors have demonstrated the method's usability efficiency and effectiveness on users with motor disabilities and able-bodied users. This work then was extended to use pressure strokes entered via an isometric joystick [271]. The authors showed that this method was a comparable alternative to the existing text input techniques, empowering users to utilise it while eyes-free without a significant effect on text entry performance [271].

Furthermore, Kane and colleagues presented "TrueKeys" – a system presented as a keyboard layout that models the word frequency and error patterns to automatically correct typing errors [113]. "TrueKeys" was developed to support and improve performance of motor-impaired users in text entry tasks. The authors showed that "TrueKeys" significantly reduced error rates for both motorimpaired and non-impaired users. The authors also suggest using personalisation algorithms to improve text entry performance as they observed consistency in typing errors within individual participants [113].

Finally, bio-signals could be used to enhance the interaction with mobile devices for motor impaired users. For instance, Guerreiro and Jorge suggest using an electromyographic signals to control mobile devices in order to assist individuals with spinal cord injuries [81].

### 2.1.3 Hearing Impairments

Interaction with the mobile device can be challenging for users with hearing impairments (Deaf or Hard of Hearing – DHH) due to different factors. For instance, Glasser and colleagues show in their work that Deaf or Hard of Hearing users have challenges when using speech-controlled user interfaces [68] due to the fact that automatic speech recognition systems are commonly trained on the speech produced by hearing users [62]. The authors suggest that speech-
controlled systems should be able to give a better feedback to DHH users to improve the usability. For example, it would be ideal if the system asked the user to repeat a specific word rather than a whole phrase. The researchers also show that having a visual feedback for speech-controlled systems enhances their accessibility features [68].

Himmelsbach *et al.* show in their work that hearing-impaired users prefer touch interactions when using their mobile devices [96]. Moreover Pfeuffer *et al.* suggest extending mobile interaction by adding gaze and demonstrate that gaze complements touch and, hence, is applicable when vocal interaction is not appropriate [207]. The authors use gaze as an extension and a replacement of direct touch. Therefore, gaze can be used by DHH users in cases when hands-free interaction needs to be performed [207].

Moreover, Hong and colleagues demonstrate that the accessibility of video content (including the content played on mobile devices) can be extended by providing "Dynamic Captioning", an approach that enables mapping captions to character faces and then placing the captions in a non-intrusive area around the face [100]. The authors demonstrate that their approach was effective and well received by the hearing-impaired participants [100]. Similarly, Avvenuti and Vecchio extended the accessibility of mobile devices by presenting an interpreter to convert the interaction between the user and the application from vocal to visual [10]. In particular, their system transforms vocal dialogues into a sequence of visual screens [10].

Morever, Massaro *et al.* presented an iOS application used to supplement speechreading by displaying an animated face to the user and transforming speech into visual cues [158]. In addition, Sun *et al.* presented "Lip-Interact" – a mobile interaction technique using silent commands [253]. The authors initially designed the technique to allow hands-free interaction similar to voice interaction; however, unlike voice interaction "Lip-Interact" accounts for privacy and social norms. As the technique allows lip interaction with the mobile device, its usability could assist not only those requiring hands-free interaction but also hearing-impaired users especially given the efficiency of the technique that the authors demonstrate through the user study [253].

Nasser and colleagues suggest extending the interaction with mobile devices by adding a wearable device to enhance the usability of mobile devices [177]. In particular, the authors suggest providing a thermal feedback through the wearable device to provide feedback to DHH users. The authors demonstrate that the thermal haptic feedback has a potential of being well accepted among the users, as they were successful in differentiating between different feedback cues [177].

Finally, mobile interaction becomes even more challenging when a combination of impairments persists. For such users, Azenkot *et al.* developed "GoBraille" – an Android application built on a novel framework (named "MoBraille") that provides information on bus arrival times as well as information on bus stop landmarks [13]. This software allowed deaf-blind users to travel more independently and safely [13].

#### 2.1.4 Cognitive Impairments

Cognitive impairments usually imply having difficulties in processing information that requires attention, thinking, and memory [89]. Cognitively-impaired users with attention disorders find it challenging to manage parallel tasks [180]. Therefore, users with cognitive impairments might find it difficult to type messages, as the soft keyboards require an increased focus-of-attention as the users need to look at the buttons while typing [275]. Hence, as Himmelsbach and colleagues mention in their work, ease of use is the main factor for cognitivelyimpaired participants when using their mobile device [96].

Dawe presented requirements for mobile devices to enhance their accessibility for cognitively-impaired users [55]. The author's findings include such requirements as simplified navigation menu, or a rugged handset. The author also demonstrates the need for remote communication using mobile devices, *e.g.*, through voicemails, and the need for supporting safety check-ins with a key aspect of tracking each other [55].

Another significant problem for cognitively-impaired people is to navigate in the environment [79]. As the risks of being lost are quite high, people with cognitive-impairments depend on their caregivers when going out or finding their way [79]. For this reason, assistive mobile technology has focused largely on creating applications to support navigation for cognitively impaired users. Chang and colleagues present a wayfinding system that provides support to cognitively-impaired users by providing context triggered prompts [47]. The main idea of this wayfinding system is to tag the context with the QR codes and provide end-user specific response. The authors show that the system was friendly and reliable in providing wayfinding responses to the participants. The researchers suggest that their method can be extended to be used with RFID tags and bluetooth beacons [47].

Similarly, Poppinga and colleagues implemented "NavMem Explorer" – a system to support users with mild cognitive impairments to navigate within the environment [216]. The authors' system was efficient in wayfinding tasks and, hence, it can potentially support independence and social life of the users with mild cognitive impairments by allowing them to go out alone [216]. In addition, Liu and colleagues developed another system to support navigation and wayfinding tasks of cognitively impaired individuals [138]. The system provided different feedback modalities including images, audio, and text for users. The authors found that the choice of a modality was purely individualistic and depended on users' personal preference. In addition, authors also found that feedback should be provided to users at appropriate times to improve their user experience [138]. Finally, Chang and Wang in their research show that when creating wayfinding technology for cognitively-impaired users, it is beneficial to support the technology with video modality as participants perform wayfinding tasks better [48].

# 2.2 Situationally-Induced Impairments and Disabilities

The aforementioned permanent impairments can be mapped to situational impairments as the effects of both permanent and situational impairments might somewhat be similar, albeit to a lesser extent [268]. Situational visual impairments can be caused by bright or dim ambient light [255]; situational motor impairments can be caused by walking or cold hands [70,268]; situational hearing impairments can be caused by ambient noise, and situational cognitive impairments can be caused by stress or any other type of cognitive load [268].

A decade ago, Wobbrock anticipated the similarity of the effects of permanent impairments and SIIDs on mobile interaction [268] which has then been demonstrated in research [279]; hence, it is important to account for SIIDs when designing technology in order for it to be accessible. Prior research has shown that users consider the cost of situational context, when addressing their needs using smartphones [247]. Due to the contextual restrictions, user's information needs are delayed to be addressed later or forgotten [247]. Furthermore, people create challenging situational contexts by trying to multitask as their attention is divided between the active task and the task to address their informational need [247].

Mobile devices are as small as the palm of a hand and are used in a number of dynamic environments [71]. These factors can lead to SIIDs (also known as situational impairments) [242], which can pose significant challenges to effective interaction because our current mobile devices do not have much awareness of our environments, and thus cannot adapt to them [71]. Sohn *et al.* show in their study that 40% of the time, participants used smartphone browsing to address their information needs [247]. Moreover, 72% of their information needs were prompted by contextual factors (*e.g.*, location, activity, time, conversation) [247].

Furthermore, it is important to consider SIIDs when permanently-impaired users are interacting with their mobile devices, as they aggravate the effects of permanent impairments. For example, very bright or dim lighting can cause accessibility issues for users with low vision [112]; mobility worsens mobile interaction performance for motor-impaired [175]. Therefore, creating situationally aware mobile devices for overcoming situational impairments and disabilities will benefit the users of all abilities [270].

This section provides and overview of research on SIIDs in terms of *Understanding* their effects on mobile interaction, *Sensing* SIIDs and building detection mechanisms, *Modelling* SIIDs or user behaviour, and mobile interaction under SI-IDs, and *Adapting* the interface to accommodate the effects of SIIDs [273]. Further, it summarises the design guidelines suggested by various researchers in the field.

#### 2.2.1 Understanding

Understanding SIIDs is an essential step in conducting research in the field of SIIDs as it plays an important role to accumulate and build knowledge on the effects of SIIDs on mobile interaction. It is also an important step as it enables conducting future research on SIIDs in terms of building sensing, modelling and adapting mechanisms to accommodate SIIDs and reduce their effect on mobile interaction. As a result, this can lead to the creation of situationally aware ubiquitous mobile devices that enhance user experience [270].

A great deal of research was conducted on understanding the effects of various situational impairments on mobile interaction; however, most of the work follows an ad-hoc approach (different studies used different smartphone tasks and measure variables without following a consistent protocol) and lacks systematic investigation. Moreover, some SIIDs received more attention than others within our research community. For example, walking has been widely investigated, while the effects of ambient noise remained underexplored [235]. This section summarises the literature in terms of understanding the effects of walking, encumbrance, cold ambience, ambient light, and divided attention on mobile interaction.

#### The Effects of Walking on Mobile Interaction

Kane *et al.* investigate the effects of walking on mobile interaction performance when using soft buttons of varying sizes [115]. The participants of the study were asked to scroll through the playlist to find a given song and tap on the song to play it. The authors found a significant effect of button size and an interaction of button size and movement on participants' performance. Their findings show that the button size affects the task completion time and error rate, and its effect size depends on the movement [115].

In another example, Lin *et al.* investigated the effect of walking on tapping task performance [134]. The authors found that walking in the environment with obstacles had a negative effect on target selection accuracy, reduced the walking speed and increased the perceived workload [134]. MacKay and colleagues studied the effect of walking on scrolling techniques on a PDA using a stylus [149]. The authors demonstrated that the participants took significantly longer time to complete scrolling tasks when they were walking as compared to a static condition [149].

Schildbach and Rukzio also studied the effect of walking on selection tasks. They have additionally extended their study on understanding the effects of walking on reading performance as well [239]. The authors demonstrated that participants' error rate increased by 24% and target selection time increased by 31% when they were walking. Moreover, walking decreased their reading speed by 19% and increased cognitive load by 16%. Similarly, Mustonen and colleagues demonstrate that walking deteriorates visual performance in reading and visual search tasks [173]. In addition, Bergstrom-Lehtovirta *et al.* studied the tradeoff

between the walking speed and target acquisition performance [23]. The authors found that 40-80% of the user's preferred walking speed enabled an optimal target acquisition performance [23].

As it is common to type on the smartphone while walking, researchers investigated the effects of walking on text input tasks. For example, Mizobuchi and colleagues studied the effect of walking on text entry [168]. Their work focused on understanding how this effect varied depending on different sizes of the text entry keys. The authors found that in general walking slowed down the text entry speed and increased the error rate; however, this speed increased with larger sizes of the keys while the error rate declined [168]. Similarly, Nicolau and Jorge in their work investigate the effects of walking on text input performance [188]. The authors demonstrate that walking led to more errors during text entry; however, this effect could be somewhat compensated by increasing the target sizes. The authors suggest using predictive text entry methods, correction algorithms, and/or adaptive keyboards to eliminate the negative effect of mobility on text entry. Moreover, the authors suggest that text entry methods should be able to sense the mobility or any other hand-tremor and then adapt the interface accordingly by increasing the target sizes in order to enhance user performance [188].

Harvey and Pointon further studied the effects of walking on mobile search tasks [90]. The researchers hypothesise that walking causes attention distraction during mobile interaction and, hence, negatively affects mobile search performance. They conducted a study with 19 participants in a controlled laboratory settings and asked their participants to complete search tasks in three conditions: being seated, walking on a treadmill, and walking with obstacles. The results of their study show that the participants found it to be more difficult to complete search tasks when walking on a treadmill and with obstacles as compared to a seated condition [90].

#### The Effects of Encumbrance on Mobile Interaction

As it is common to be interacting with the mobile device while at the same time being encumbered (*e.g.*, carrying shopping bags), Ng and colleagues investigated the effect of encumbrance on mobile interaction [182]. The results of their study show that encumbrance has a significant negative effect on accuracy of target selection tasks with participants having more errors when they were carrying a

small and a medium bag, as well as a thin and a thick box. Furthermore, the authors added the effect of mobility to the encumbrance by instructing participants to walk according to a defined path while being encumbered to complete target selection tasks on a smartphone. The researchers found a significant negative interaction effect of encumbrance and mobility on mobile interaction performance. Furthermore, the authors found a negative effect of encumbrance on the walking speed with a decline of approximately 41% as compared to a static position [182].

In their follow-up work, Ng *et al.* extended this research by investigating the effects of encumbrance on different three interaction postures: two-handed mode using index finger to interact with the device, two-handed mode using both thumbs to interact with the device, and one-handed mode using a thumb to interact with the device [183]. The results of this study showed that accuracy in target acquisition tasks dropped to 48.1% in a one-handed interaction mode using index finger, while the error rate increased by 40% in one-handed interaction mode using thumb when participants were encumbered [183].

Investigating the effects of combined situational impairments is not common within the research community; however, Ng and colleagues studied the effects of both encumbrance and walking on touch-based gestures: tapping, dragging, spreading and pinching, and rotating [185]. The authors used Fitts' law measurements to quantify these effects on the above-mentioned gestures. The findings of the study showed that encumbrance and mobility had a negative effect on the performance of each gesture, except for rotational activity. In particular, the participants were less accurate when performing tapping, dragging, spreading and pinching gestures. In terms of task completion time, participants were slower when performing tapping, dragging and rotation tasks. The authors demonstrate that it is important to account for physically demanding contexts when designing interaction techniques for mobile devices [185].

#### The Effects of Cold Ambience on Mobile Interaction

Some research work focused on understanding the contextual and environmental factors that might lead to SIIDs on mobile interaction. For example, Choi investigate the effects of contextual changes on human behaviour during mobile interaction [51]. The author found a significant negative effect of walking on reading comprehension task on a mobile device as the reading time increased when the participants were walking as compared to a sitting condition. The author also found that under different lighting conditions reading comprehension performance was significantly different with a longer response time under the dark ambient light as compared to the bright ambient light condition [51].

Previous work investigated the effect of cold ambient temperature on mobile interaction from the perspectives of fine-motor movements and vigilance [231]. The findings show that cold ambience had a negative effect on fine-motor performance during mobile interaction on target acquisition tasks, but not vigilance. In particular, target acquisition time becomes longer and touch accuracy drops when participants are exposed to cold environment as compared to the warm environment [231].

Goncalves *et al.* extended the previously mentioned study [231] and investigated the effect of cold ambience on mobile interaction performance in target selection tasks [76]. The researchers show that under the cold ambience smartphone interaction performance deteriorates as the throughput drops and error rate increases. Based on these findings, the researchers suggest using ambient temperature as one of performance predictors in Fitts' Law [59].

#### The Effects of Ambient Light on Mobile Interaction

Another contextual factor that has been investigated on mobile interaction performance is ambient light. Lee and colleagues demonstrate in their study the effects of ambient light on visual search performance during reading [129]. The authors demonstrate that the search speed increased as the surrounding illuminance increased. This means that dark ambient light was associated with slow search speed, while bright ambient light improved the participants' performance in visual search tasks [129]. Furthermore, the authors also show that the size of characters in reading task had an effect on search performance of their participants: bigger characters were associated with higher search speed. The authors recommend the minimum size for character of 3.3 mm for optimal search performance on a mobile device [129].

Moreover, Liu *et al.* studied the effects of different illuminance levels on character detection task on a mobile device [139]. The researchers report that their participants' performance decreased when they completed the character detection task under bright illuminance as compared to low illuminance levels due to glare of the device [139].

The authors demonstrate the importance of accounting for ambient illuminance on mobile interaction particularly for the cases when mobile interaction comes at a cost (*e.g.*, during medical operation in need of using mobile device to control technology) [139].

#### The Effects of Divided Attention on Mobile Interaction

Finally, there was an attempt to study the effects of not only the external, but also "from within" factors on mobile interaction (*i.e.*, coming from user's internal states [273]). For instance, Oulasvirta and colleagues investigated the effect of fragmented attention caused by moving through urban settings on a mobile browsing task [196,197]. The authors report that participants attention focused for 6-16 seconds on the task with the intermittent breaks of 4-8 seconds. The authors' results show that attention resource competition is real and constrains mobile interaction.

#### 2.2.2 Sensing

Mobile device sensors have been widely used in the HCI and UbiComp communities for various purposes. For example, a fair amount of work focused on using device sensors to detect context (*e.g.*, [2,56,240]), while other researchers used smartphone sensors for activity recognition (*e.g.*, [8,252,265]). Similarly, smartphone sensors have been used to detect SIIDs and this section provides an overview of existing work in terms of sensing SIIDs.

For example, Goel and colleagues used smartphone's built-in accelerometer to detect if a user is walking [70]. Based on this sensing, the authors then adapted the keyboard to overcome the SIIDs caused by walking. In particular, the authors used the displacement and the acceleration of the device for the classification model for text entry. The authors demonstrate how successful detection of walking then can lead to creation of adapting interfaces to mitigate the effects of walking on a text entry [70].

Furthermore, smartphone sensors have been used to detect essential tremors [45, 67, 109, 225]. These works utilised smartphone's accelerometer sensor to detect hand tremors. For example, Daneault *et al.* used smartphones to collect and process the accelerometer data which they then used to correlate with the laboratory

accelerometer data [53]. The authors found that the tremors with amplitudes lower than 1mm were not detected with the smartphone; however, for high amplitude tremors (> 1mm) the correlation between the smartphone data and the laboratory accelerometer data was relatively high (r > 0.88). These above examples use accelerometer to detect both SIIDs (*i.e.*, caused by walking) and permanent impairments (*i.e.*, essential tremor); hence a detection mechanism designed to sense SIIDs can be used to also detect permanent impairments as was envisioned by Wobbrock [268].

Azenkot and Zhai [16] found that majority of users at least "sometimes" used their phones with either the thumb of their dominant hand (one-handed holding posture), their dominant index finger (two-handed holding posture), or both thumbs (two-handed holding posture). Therefore, a substantial amount of work focused on identifying the grip and device holding posture of the users. For instance, Goel *et al.* in their work present "GripSense" – a system to detect hand-posture based on the device's gyroscope, vibration motor, user touch size, and swipe shape [72]. The authors demonstrate that their system was extremely accurate when detecting if the device was on the table or held in hands with a detection accuracy of 99.7%. However, this accuracy slightly dropped when detecting between different hand postures (84.3%). Furthermore, the authors suggest their system can be useful to detect when the user is situationally-impaired due to encumbrance and can only operate the mobile device in one-handed interaction mode [72].

Similarly, Gupta and colleagues also focused on identifying users' grip strength. The authors suggested using a concept of a virtual spring to detect the strength of the grasp when holding the device with one hand in order to enable eyes-free interaction [85]. In their work they presented an electro-mechanical system called "SqueezeBlock" that can be used in various use-case scenarios, *e.g.*, changing the ringer volume by squeezing the device [85]. This concept can be applicable for situations when the user is encumbered to ease the interaction with the device by limiting to the squeezing gesture.

This feature has then later been implemented within Google smartphones. Precisely, Google has implemented "Active Edge" feature on Pixel mobile devices to detect the grip force of the user. Depending on this grip force, the smartphone launches Google Assistant, thus allowing quick interaction for the user [218]. This feature can be applicable during situational impairments when the user cannot interact with the device with both hands or requires quick activation of the Google Assistant with one hand (*e.g.*, encumbrance, diverted attention). Quinn and colleagues from Google evaluated "Active Edge" and report that the technique was found to be easy, comfortable, and reliable as an interaction technique to silence the alarm, take pictures, and voice commands [218].

Several researchers proposed using smartphone's built-in sensors to protect privacy issues; hence, address privacy-induced SIIDs [87, 143]. For example, Haque *et al.* suggested using smartphone accelerometer sensors for authentication purposes [87]. In particular, the authors recommend using accelerometer to distinguish the user's gait pattern and, hence, authenticate the user. The authors claim that this approach can be extended to a conventional smartphone users to secure authentication in vulnerable locations and situations, *e.g.*, at a night time when unlock pattern can be visible due to the screen light of the device [87].

Moreover, Mariakakis *et al.* in their recent work presented "Drunk User Interfaces (DUI)" that are used to quantify the effects of alcohol on human motor coordination and cognition during mobile interaction [153]. They utilised sensor data in combination with user input data to build prediction models to detect blood alcohol levels. The authors demonstrated that their system was highly accurate to detect users' blood alcohol levels with an absolute mean error of  $0.005\% \pm 0.007\%$  and with a strong Pearson's correlation of 0.96 with the ground-truth measurements.

Mobile device sensors have been suggested to be used to detect surrounding context. For example, Yi and colleagues utilised a single tri-axial accelerometer attached to a mobile device to sense contextual information and have shown that their system was able to successfully determine the mobile state of the user and ambient light [280].

Similarly, Mass and Madaus discuss using smartphone pressure sensors to detect environmental pressure [157]. The authors claim that the network of smartphones could provide a reliable pressure data as it would not be influenced by external factors [157]. In addition, Overeem and colleagues demonstrated in their work that smartphone temperature data can be used to predict daily average air temperature [198].

Finally, Reis and colleagues presented two context-aware prototypes that were able to automatically adjust volume depending on the contextual situation based on the detected level of ambient noise [221]. The interfaces used smartphone's microphone to sense the levels of surrounding noise. The prototypes also allowed accounting for user preferences that were collected through users' settings and modifications. The authors demonstrate that the systems were positively accepted amongst the users, and hence show that accounting for noise-induced SIIDs is important when designing interaction technology [221].

We extend this literature by presenting a sensing mechanism to detect coldinduced SIIDs on mobile interaction and present our findings in Chapter 8 of this thesis.

#### 2.2.3 Modelling

Modelling is an aspect of SIIDs research that focuses on modelling either environment or user behaviour to enhance our understanding of SIIDs. Furthermore, modelling can include creation of machine learning models that are used to predict either the effect of SIIDs on mobile interaction or human behaviour under SIIDs. In addition, several researchers have created models which were used for creation of adaptive user interfaces to accommodate the effects of SIIDs. This section summarises modelling examples in SIIDs that exist in literature and have been implemented by the researchers.

For example, Flatla and Gutwin developed and presented individual models for colour differentiation for users affected by permanent (colour vision deficiency) or situational visual impairments (*e.g.*, bright light) [60]. The authors demonstrate that their model was effective in detecting user's individualistic colour differentiation abilities and efficiently improved colour adaptation of mobile device's screen [60].

There was an element of modelling in the work by Mariakakis *et al.* when creating "Drunk User Interfaces" [153]. In particular, the authors developed an application that used machine learning models to determine the blood alcohol level of the user. The authors trained machine learning models on real human behaviour data observed and collected from interaction performance and sensor data [153].

Mott and Wobbrock presented "Cluster Touch" – a touch offset model to improve touch input accuracy for motor-impaired users. This work can be extended to support users experiencing SIIDs due to walking [171]. The general model for "Cluster Touch" is built based on touch data from multiple users and can then be personalised based on individualistic touch behaviour of the users. The authors demonstrate the efficiency of their technique, as "Cluster Touch" improved touch accuracy of motor-impaired users by 14.65%; while this number for situationally-impaired users reached 6.81% as compared to the native touch baseline. Furthermore, the touch accuracy during an offline analysis improved by 8.21% and 4.84% for permanently and situationally impaired users respectively [171].

In addition, several works have suggested using language models to adjust key press probabilities [5,78,84]. For example, Buschek and Alt presented "ProbUI" – probabilistic graphical user interface framework for mobile devices that defines touch behaviours, evaluates them probabilistically, and infers touch intentions based on the first two steps [35]. This framework could be used to improve the gesture and touch accuracy in situations resulting in a reduced touch accuracy due to the user being situationally-impaired (*e.g.*, walking).

As users perform differently when interacting with their mobile devices, it is necessary to take into account personalised characteristics of human behaviour in mobile interaction [269]. For instance, Buschek *et al.* emphasise the importance of individualistic characteristics of touch interaction [36]. The authors demonstrate that the touches performed using thumb are more individualistic than the index finger touches. Hence, the mobile devices should be able to differentiate between different holding postures and adapt accordingly depending on the interaction mode [36].

In addition, one of the examples of successful personalisation in text entry was the system named "Text Text Revolution" (TTR) presented by Rudchenko and colleagues [224]. In their user study, the authors trained target resizing on touch point collected from the participants first 10 rounds of performing tasks in TTR, and simulated personalised target resizing models in the second 10 rounds of TTR. Their results showed an improvement in text entry as the error rate reduced by 21.4% [224].

#### 2.2.4 Adapting

The adapting aspect of SIIDs research is directed to create adaptive interfaces to accommodate the effects of different SIIDs. This section presents several adaptive interfaces existing in the literature, that are used to mitigate the effects of walking, encumbrance, situational visual impairments, and situational privacy issues that arise due to the vulnerability of the environment.

#### Interface Adaptations to Accommodate the Effects of Walking

As walking was paid more attention than other SIIDs within the research community, several researchers worked on creating adapting interfaces to accommodate the effect of walking on mobile interaction. For example, to mitigate the effect of walking on song selection task, Kane *et al.* created an adaptive *walking user interface* for a music player that scales target sizes based on user motion [115]. The interface shrinks the buttons when the user is standing still; however, when the user is walking, the targets and text expand in size to improve readability and touch access [115]. Their evaluation results show that the interface adaptation to accommodate walking significantly improved task completion time; however the authors suggest that task difficulty and other individual characteristics of the users should also be taken into account to create successful adaptive interfaces to mitigate the effect of walking [115].

Another suggestion to overcome walking-induced situational impairments was to use screen stabilisation techniques to stabilise the content of the screen [219]. For example, Rahmati *et al.* present "NoShake" – a system that utilised smartphone's accelerometer to detect if the device was shaking and then compensate for shaking by shifting the screen content in the opposite direction. The results of their study show that "NoShake" improved user experience in presense of shaking. The authors also suggest that the system could improve user experience of people who are unable to hold the device steadily, *e.g.*, those suffering from Parkinson's disease [219].

Yamabe and Takahashi also proposed a user interface adaptation approach to accommodate for the effects of walking in mobile interaction [276]. In particular, the authors suggest changing the size of the screen elements based on the display movement. Furthermore, the authors suggest taking into account user's individual characteristics when designing adaptive interfaces to diminish the effects of

walking [276]. Brewster suggested accompanying small buttons with the sound feedback to enhance the usability of the interface [31] that might particularly be useful for situations when touch is inaccurate, *e.g.*, due to walking.

A fair amount of work has focused on creating adaptive interfaces to improve text entry methods as it is one of the common tasks users perform on their smartphones. For instance, Goel *et al.* introduced "WalkType" – an adaptive text entry system for smartphones that compensates for movement when users are walking [70]. The system utilises devices' accelerometer sensors to detect if the user is walking. The authors demonstrate that "WalkType" reduces errors by 45.2% and improved typing speed by 12.9% [70]. Similarly, Himberg and colleagues developed an adaptive keyboard that arranges keys according to the spatial distribution of keystrokes [95]. The results of their evaluation show that the changes in the keyboard are consistent within each user. Hence, the authors conclude that personalisation should be taken into account when designing adaptive keyboards [95].

In addition, Go and Endo introduced personalisation in adaptive keyboards by presenting "CATKey" – an adaptable keyboard for touchscreen devices with customisable functions [69]. The keyboard was able to adapt each key's centroid to the centroid of recorded keystroke points. Although "CATKey" did not have any significant advantage in terms of improved efficiency in typing during evaluation, the participants expressed their preference towards "CATKey" as compared to the traditional "QWERTY" keyboard [69]. This keyboard is another example how text entry can be improved during walking.

#### **One-Handed Interaction to Overcome Encumbrance**

Buschek and colleagues in their work presented and evaluated dynamic adaptations of mobile touch interfaces to overcome the inconvenience of using a large screen in a one-handed interaction mode [37]. The authors present three techniques – "Roll", "Bend", and "Move" – to locate GUI elements on the screen at a comfortable reach for the user. The authors show that their techniques improved users interaction due to the increased comfort of usability, decreased fatigue, and ease of grip [37]. These techniques could be used to overcome the effects of encumbrance during mobile interaction, as it is common to perform a single-handed interaction with the device when encumbered [185]. Karlson and colleagues introduced "AppLens" and "LaunchTile" – user interface techniques designed to enable one-handed interaction with the mobile device [117]. Both interfaces used the zoom technique with a difference that "AppLens" used a tabular fisheye, while "LaunchTile" used a pure zoom to be used with specific gestures to perform using a thumb [117]. The authors demonstrate that the users were more efficient when using "AppLens" as compared to "LaunchTile" in terms of task completion time and preferred using it over "LaunchTile".

Karlson and Bederson presented "ThumbSpace" – an interaction technique for one-handed interaction mode using thumb [116]. The results of their evaluation show that "ThumbSpace" was positively received by the users and improved their performance in completing target acquisition tasks on a smartphone designed for small size targets. Furthermore, the authors suggest this interaction technique could also be useful for encumbered users, as it allows to free one hand and effectively complete smartphone interaction [116].

"Twiddler" is a text-entry technique presented by Lyons and colleagues [148]. Similar to work mentioned above [37,116,117], this method could be used for text entry when a user is being encumbered as it enables single-handed text entry and proves to be effective and intuitive as the authors demonstrate in their work [148]. Another system that can be used in encumbered situations is "Unigesture" developed and presented by Sawazal *et al.* [238]. "Unigesture" – a tilt-to-write system that enables one-handed text entry [238]. "Unigesture" uses accelerometer data to determine the tilt of the device and acquire the input character that is mapped to a particular tilt. The evaluation of the system demonstrated that the individual characteristics of the user had a greater effect on their text entry performance. Hence, the authors suggest taking into account personal characteristics of users when designing adaptive technology for text entry [238].

In addition, Boring and colleagues created the "Fat Thumb" interaction technique that allows using thumb's contact size with a small size for panning and a big size for zooming [26]. After evaluating this technique, the authors demonstrate that the "Fat Thumb" interaction technique compared quite well with the existing mobile interaction techniques. "Fat Thumb" could be used efficiently in situations when the user is encumbered or cannot interact with the device in a traditional manner [26]. Furthermore, not only being encumbered can cause difficulties when interacting with a mobile device, but also small screen size [89, 279] can lead to the "Fat finger problem" [246]. Baudisch and Chu suggest using the back of the device to solve this problem during selection tasks on a mobile device [21]. The researchers show that using the back of device for mobile interaction enables higher accuracy and lower error rate in target selection tasks independent of device size [21].

Moreover, Buschek and colleagues also suggest using back of the device to allow rapid and efficient typing as it engages all ten fingers of the users [39]. The authors' technique [39] can be used to solve the problem of a small screen that is also stated to cause situational impairments to mobile device users [31,279].

Finally, as research has shown that hand posture can either improve or deteriorate the performance on text entry, some work took into account the holding posture of the device in text entry [71]. For example, "ContextType", a system that infers users' hand postures to improve text entry on mobile touch screen devices. "ContextType" supports typing with four hand postures: two thumbs, just the left thumb, just the right thumb, and either index finger [71]. "ContextType" switches between underlying touch-models based on inference about how the user is holding the device while typing, without changing the visual layout of the keyboard and leverages previous work by Goel *et al.* on "GripSense" based on detecting holding posture for the device [72] that was mentioned earlier in this chapter [71].

#### Non-visual Interaction to Accommodate Situational Visual Impairments

In some situations it is important than the mobile device allows eyes-free interaction, *i.e.*, when the user cannot interact with the device due to *e.g.*, mobility, divided attention, or situational visual impairments. To address this requirement, Jain and Balakrishnan developed a bezel gesture-based text entry method to allow eyes-free text entry on a mobile device [102]. Their system was easy to adapt to, as the novice users transitioned to being an expert after one hour of training. The authors also demonstrate that the bezel-based text entry was comparable in terms of speed, accuracy, ease of learning and use with existing traditional text entry methods [102]. This example showcases how eyes-free text entry can be enabled using gestures. Similarly, Chen and colleagues presented "Swipeboard" – an eyes-free text entry method that uses swipe gestures for inputting text [49]. The technique utilises two swipes: 1) to specify the region of the character location, and 2) to select the necessary character. The authors demonstrate that with an extensive training users perform text entry 15% faster using "Swipeboard" than the conventional baseline text entry method [49].

In addition, several works have been shown to use device's accelerometer for text entry to enable eyes-free interaction [107, 204, 238, 267]. For example, Jones *et al.* utilised mobile device accelerometers for text entry using mid-air gestures [107]. The authors showed that accelerometer-based text entry gestures had a significantly higher words per minute rate and a lower error rate. In addition, subjective ratings of the study's participants showed their preference for accelerometer-based text entry technique [107]. Partridge and colleagues also suggested using accelerometer for text entry, however unlike Jones *et al.* [107], they used accelerometer together with the buttons [204]. This interface was suggested to be used for smartwatches. The combination of the device tilt and pressed button were mapped to a particular character [204].

Moreover, Pielot *et al.* present "PocketMenu" that enables non-visual interaction with the mobile device by locating the menu items on the edge of the touch screen [210]. The system provided a vibrotactile feedback and speech input to enable non-visual interaction. When evaluating "PocketMenu" in comparison with iPhone's VoiceOver, the authors found that "PocketMenu" outperformed VoiceOver as it was quicker to complete tasks with and had lower selection error rate with higher subjective usability values. The authors propose using "PocketMenu" in situations when the non-visual interaction is needed, such as situational visual impairments, walking, hiking, cycling [210].

Finally, Mariakakis and colleagues implemented "SwitchBack" – a method allowing users to resume tasks in a more efficient manner [154]. The authors used mobile device's front-facing camera to determine the gaze of the user and based on this information indicated an area on the screen. The method improved user reading speed by 7.7% when the user was distracted. This system enabled easy return of user attention back to a mobile device in presence of distractions of the surrounding environment [154].

#### **Overcoming Privacy Issues**

It has been shown in literature that users can be situationally-impaired due to privacy issues [269]; hence, it is important to account for privacy when creating adaptive interfaces to accommodate SIIDs. As such, Lyons and colleagues suggest using dual-purpose speech interaction to overcome privacy challenges when using voice for interacting with mobile devices [147].

A dual-purpose speech serves two roles: 1) it is socially appropriate and meaningful in the context of human conversation; 2) it provides input to a computer. Hence, only the user can be aware of the commands being executed on the device [147].

Buschek and colleagues present "SnapApp" – an unlock concept to reduce authentication overhead by enabling time-constrained quick-access to the mobile device [38]. The technique can be used in situations when the user needs a quick access to the device while having difficulties to unlock it, *e.g.*, while walking or being encumbered. Azenkot and colleagues suggest using tapping sequence to authenticate the user [14]. Precisely, the authors present "PassChords" – an eyes-free touchscreen authentication method based on tap patterns. The authors show that their technique was 3 times faster than the iPhone's standard Passcode Lock with VoiceOver [14]. This technique can be used to enable eyes-free unlock in situations where the visual attention of the user is occupied with another task.

#### 2.2.5 Additional Design Guidelines

This section presents additional design guidelines proposed but not implemented by the respective researchers to potentially overcome different SIIDs.

For example, to diminish the effect of walking on target selection task, Schildbach and Rukzio suggested increasing the button size by a range of 20% and 40% [239]. Moreover, Parhi *et al.* conducted a study aimed at examining the effect of different button sizes on mobile interaction [201]. The study shows that the target selection time and accuracy improve with larger target sizes. The authors argue that target sizes of 9.2 - 9.6 mm is an optimal tradeoff between the target selection time and target size. Nevertheless, the authors suggest taking into account the screen size of the device [201]. Moreover, to decrease the effects of hearing impairments on mobile interaction, both permanent and temporal (caused by SIIDs such as ambient noise), the literature suggests enlarging the cursor [34], using highly visible visual feedback (*e.g.*, flash) to grab user attention [50], and vibrotactile haptic feedback [32]. To reduce the effects of hands motion or tremor on mobile interaction, caused both by permanent impairments of upper limbs and temporal SIIDs caused by *e.g.*, walking, the literature suggests using alternative keyboards, audio input (*e.g.*, voice commands), and eye tracking [151].

Qian *et al.* suggest using tactile notifications to support situationally-impaired users in order to free their visual and auditory attention for other tasks [217]. Furthermore, Yatani and Truong showcased that haptic feedback is effective in supporting eyes-free interaction, as users in their study could distinguish between ten different patterns of tactile feedback with an accuracy of 90% [277]. Similarly, Bragdon *et al.* suggest using Bezel gestures to overcome SIIDs as in their study the authors demonstrate that the environmental factors (*e.g.*, mobility and divided attention) did not have an effect on bezel gestures; hence, the users performed faster and more accurately when interacting with a mobile device [28].

Research has shown that the location on the screen also needs to be taken into account when designing adaptive interfaces to accommodate for different SIIDs. For instance, Wobbrock and Gajos suggest a target selection technique called "goal-crossing" that considers the area around the target as a selection area. As a result it improves target selection accuracy for motor-impaired users and motion-induced SIIDs [272]. Although the authors suggest using the technique for desktop computers, we argue that the method could also be adapted for mobile interaction.

Meanwhile, Kane *et al.* found in their study that the blind users prefer using the edges of the smartphone for the gestures to interact with their mobile device [114]. It is also important to remember that gestures are more preferred when interacting with mobile device as compared to other interaction methods [222]. For example, Reyal and colleagues in their study show the feasibility of gesture-based input keyboard, as their findings demonstrate that the users have a tendency to switch from text-based input keyboard to gesture-based input keyboard [222]. In addition, it is also important to account for different ways to present information as it can benefit interaction when adapting for SIIDs. For instance, Brewster *et al.* suggest using a 3D audio radial pie menu and 2D gestures to facilitate eyes-free interaction [33]. The results of their evaluation show that users were more accurate when provided with an audio-feedback and, hence, improve the usability of mobile devices under eyes-free interaction mode [33].

It is also crucial to provide feedback when designing adaptive interfaces to reduce the effects of SIIDs. Findlater *et al.* formulated design guidelines for touch-screen keyboards that support touch-typing with limited tactile feedback [58]. After investigating typing patterns of 20 professional typists, the authors conclude that creation of effective solutions for non-visual keyboard designs requires personalisation as the key presses were consistent within individual participant [58]. Moreover, Hoggan *et al.* studied the effect of tactile feedback on mobile interaction performance and also showed that tactile feedback significantly improved text entry [98]. Moreover, Hoggan *et al.* suggest taking into account a surrounding environment when providing feedback to the user [99]. The authors empirically show that audio feedback becomes ineffective at noise levels of 94*dB*, while tactile feedback become ineffective at vibration levels of 9.18*g*/*s* [99]. These findings show the importance of the mobile device being able to sense the context to appropriately choose the feedback modality for the user.

In addition, Pielot and colleagues suggest supporting navigation systems with a tactile feedback to free users attention for other tasks from navigating within the environment [208,209]. Qian *et al.* also suggested using tactile feedback to support users during SIIDs in order to free their visual and auditorial channels to complete other tasks that demand their attention [217]. The authors suggest extending their work on to wearable devices to extend the smartphone capabilities [217].

Furthermore, Harper *et al.* suggest using multimodal input that includes combination of oral, visual, audio, and haptic feedback to increase the efficiency of mobile interaction when the participants are experiencing visual impairments [89]. Similar to Harper *et al.* [89], Hoggan and Brewster also suggest using multimodal and crossmodal interaction with mobile devices to substitute visual feedback for mobile notifications [97]. The authors show that participants trained to understand audio alerts could recognise corresponding tactile alerts without any training and vice versa. Therefore, crossmodal features added to the mobile interface elements and hence enable non-visual information about these elements on mobile interface [97]. Pielot and colleagues also suggest using multimodal feedback, including tactile and visual, to improve navigation systems on handheld devices [211]. The authors evaluated their system with 21 participants and have shown combination of the two modalities improved the participants' navigation performance. The authors also show that in the presence of the tactile feedback is used, the user distraction levels decrease [211,212]. Furthermore, in their follow-up work, Pielot and colleagues demonstrate that tactile feedback can be successfully adopted to reduce the effects of SIIDs that appear when using navigation systems (*e.g.*, bright sun light) and, hence, reduce the distraction levels of users [213].

Finally, Barnard *et al.* suggest considering personal characteristics of interaction behaviour, as SIIDs do not affect people in a uniform way [20]. The authors suggest using a combination of hardware sensors, software, and design to overcome the challenges introduced by SIIDs on mobile interaction [20].

#### 2.2.6 Summary

Table 2.1 summarises the findings from section 2.2 in terms of understanding SIIDs including the smartphone tasks used to quantify the effects of SIIDs on mobile interaction, sensing mechanisms used to detect different SIIDs, existing modelling and adapting mechanisms that have been implemented, and other distilled design guidelines recommended by the researchers.

As can be observed from the detailed literature review and stated in [235], the effects of some SIIDs remain underexplored. The literature review also demonstrates that the studies conducted in the field of SIIDs followed an adhoc approach, and there were no standard measures and rules to quantify the effects of SIIDs on mobile interaction performance; hence, creating an additional challenge of comparing these effects. We extend this literature by investigating the effects of ambient noise, the effects of stress, and the effects of dim ambient light on mobile interaction in Chapters 4, 5, 6 of this thesis respectively. We also provide a sensing mechanism to detect cold-induced SIIDs in Chapter 8. Finally, we showcase that by following a systematic approach in studying the effects of SIIDs to each other. It is important to note that this table is limited to the SIIDs that have been studied within the research community and includes established findings from the literature. However, a more extensive list of SIIDs that were not yet investigated or acknowledged can be found in [269, 273].

SIID	Understanding			Modelling and	Design
	Smartphone	Effects of	- Sensing	Adapting Interfaces	Guidelines
	Task	SIID			
Walking	Scrolling [115, 149];	Increased error rate [115]; Increased task completion time [115,149];	"Cluster Touch" [171]; "ProbUI" [35]; Accelerometer [53,70,278]; Bezel gesture-based text entry method [102]; "SwipeBoard" [49];	"Cluster Touch" [171]; "Probl II" [35]:	Sound feedback for buttons [31]; Change the size of the screen
	Target acquisition [23, 134, 239];	Increased error rate [134,239]; Increased workload [134]; Increased target access time [239];		elements [201,239]; Alternative keyboards using audio input and eye-tracking [151];	
	Reading (on the device) [51, 173, 239	Decreased reading speed [173,239]; ]; Increased cognitive load [239]; Increased task completion time [51];		"WalkType" [70]; "ContextType" [71]; "Walking user interfaces" [115]; "Text Text Revolution" [224];	Bezel gestures [28]; Increase target selection area [272]; Tactile feedback [211,212];
	Visual search [90, 173];	Decreased visual search speed [173]; Increased task completion time [173]; Increased cognitive load [90];		"CATKey" [69]; "NoShake" [219]; Adaptive text entry [95];	Multimodal feedback [89,97,211];
	Text entry [168,188];	Decreased text entry speed [168]; Increased error rate [188];		Accelerometer-based gestures [107,202	];
Encumbrance	e Target acquisition [182, 184];	Increased error rate [182, 184];		Bezel gesture-based text	
Encumbrance	Target acquisition [182, 185];	Decreased walking speed due to encumbrance [182]; Decreased accuracy [185]; Increased target access time [185];	Gyroscope, vibration motor, user touch, swipe shape [72]; "Virtual spring" [85]; "Google Active Edge" [216];	entry method [102]; "Text Text Revolution" [224]; Dynamic adaptive touch interfaces [37];	Bezel gestures [28]; Alternative touch interfaces [37];
+ Walking	Dragging [185];	Decreased accuracy [185]; Increased task completion time [185];		Back-of-device interaction [21,39]; "Fat Thumb" [26];	
	Spreading [185];	Decreased accuracy [185]; Increased task completion time [185];		"ThumbSpace" [116]; "AppLens", "LaunchTile" [117];	5pace" [116]; ns", "LaunchTile" [117];
	Rotating [185]; Pinching [185];	Increased task completion time [185]; Decreased accuracy [185];		"Twiddler" [148]; "Unigesture" [238];	
Cold ambience	e Target acquisition [76,231]; Visual search [231];	Increased error rate [76,231]; Increased target access time [76,231]; Decreased throughput [76]; No significant effect [231];	Battery sensor [Chapter 8];	n/a	Change the size of the screen elements [201,231,239];

**Table 2.1:** Summary Table of Situationally Induced Impairments and Disabilities in Terms of Their Understanding, Sensing, Modelling, and Adapting

				Dual-purpose speech	
Privacy	n/a	n/a	Accelerometer [87];	interaction [147];	Speech recognition [147];
				"SnapApp" [38];	Non-visual touch interfaces [14] ;
_				"PassChords" [14];	
Alcohol	Text entry, Swiping, Simple and Choice reaction tasks [153];	n/a * The authors used these tasks for building predictive models but not to understand alcohol-induced SIIDs on mobile interaction;	Accelerometer, touch-screen [153],	"DUI" to model blood ; alcohol levels [153];	n/a
Fragmented attention	Browsing task [196,197];	Decreased attention time to the task [197];	n/a	"SwitchBack" [154];	Bezel gestures [28]; Tactile feedback [208,209,211,212,217];
	Visual search [129];	Increased search speed [129];		Colour differentiation models [60];	Tactile notifications [217];
Bright ambient			Accelerometer, ambient light	bezel gesture-based text	Tactile feedback [58,98,213,277];
light	Character detection [139]	Increased error rate [139];	sensor [278];	entry method [102]; "SwipeBoard" [49];	Using smartphone edges [114];
				"PocketMenu" [210];	3D radial pie menu [33];
					2D gestures [33];
					Multimodal feedback [89,97,211];
Ambient noise	Target acquisition [Chapter 4] Visual search [Chapter 4] Text entry [Chapter 4]	Increased target access time under music and urban noise [Chapter 4]; Decreased accuracy under music [Chapter 4]; Decreased target memorisation time under urban indoor noise [Chapter 4]; Increased error rate under urban outdoor noise [Chapter 4]; Increased time per character entry under urban outdoor noise and meaningful speech [Chapter 4];	Microphone [219];	n/a	Enlarge a cursor [34]; Provide highly visible visual feedback [50]; Haptic feedback [32];
Stress	Target acquisition [Chapter 5] Visual search [Chapter 5] Text entry [Chapter 5]	Increased target access time [Chapter 5]; Decreased accuracy [Chapter 5]; Decreased target memorisation time [Chapter 5], No effect	n/a	n/a	n/a

	Target acquisition [Chapter 6] t	Increased target access time [Chapter 6];			Tactile notifications [217]:
Dim ambien		Decreased accuracy [Chapter 6];	Accelerometer, ambient light	n/a	Tactile feedback $[58, 98, 213, 277]$
light	Visual search [Chapter 6]	Increased target memorisation time [Chapter 6]	]; sensor [278];		Using smartphone edges [114]:
	Text entry [Chapter 6]	No effect			2D re diel mie menue [22]
					3D radial pie menu [33];
					2D gestures [33];
					Multimodal feedback [89,97,211];

# **Chapter 3**

# Methodology

This chapter outlines the methodological approach that was followed in the research presented in this thesis. In particular, it provides the motivation behind particular decisions that were employed in the methodology of the experiments presented in Chapters 4, 5, 6, and 8. The purpose of the studies conducted and presented in Chapters 4, 5, and 6 was to provide an *understanding* of the effects of different situational impairments on mobile interaction; whereas, the purpose of the study presented in Chapter 8 was to introduce a *sensing* mechanism to detect cold-induced SIIDs. All of the experiments conducted within the framework of this thesis were conducted as laboratory studies with strictly controlled settings to avoid causing potential harm to participants and exclude the effect of potential confounding factors.

The focus of the experimental methodology described in this section was to investigate the effects of different SIIDs in a systematic way which allows a fair comparison of these effects between each other. Being able to compare different SIIDs is a crucial step for enhancing SIIDs research [235], as this knowledge is a stepping stone for building appropriate *sensing*, *modelling*, and *adapting* mechanisms that would accommodate the most prominent SIID while at the same time potentially addressing other accompanying SIIDs [256].

# 3.1 Systematic Approach towards Understanding the Effects of SIIDs

In our systematic approach to investigate the effects of different SIIDs on mobile interaction we have developed an experimental protocol that was reused across multiple studies (Chapters 4, 5, 6). In particular, we re-used the same smartphone tasks to quantify the effects of different SIIDs on mobile interaction. We have also re-used the same observational, scoring, and calculation rules to construct and record dependent variables measuring user performance in mobile interaction under different SIIDs.

We have always followed a within-subject experimental design, in that participants were assigned to all of the conditions in each of the studies, and we also counter-balanced the order of the presentation of the conditions to avoid potential learning effects. In addition, upon the start of the experiment, we briefed our participants about the purpose of the experiment to ensure that each participant was equally informed about the study. In the experiment presented in Chapter 5 we deceived our participants about the true purpose of the study to avoid participant bias and observe natural reaction of participants to stress.

Furthermore, in all of the studies presented in Chapters 4, 5, and 6 we collected baseline measurements of mobile interaction performance upon which the effect of SIIDs were measured. Due to the fact that participant samples in the abovementioned studies were different, direct comparison of the effects of SIIDs was not possible. However, by contrasting the improvement and/or deterioration in mobile interaction performance for different SIIDs conditions as compared to their relative baseline, we provide a fair comparison of the prominence of SIIDs' effect on mobile interaction. Moreover, we restrict our participants posture for holding smartphone device to perform two-handed interaction: holding a smartphone in a non-dominant hand, while interacting with the smartphone with an index finger of a dominant hand. The holding posture is controlled in order to eliminate the effect of the posture on mobile interaction. At the end of each experiment we held semi-structured interviews with our participants to collect qualitative data to support or contrast collected quantitative data. Hence, based on this methodology we provide a comparison of the magnitudes of the effects of different SIIDs (ambient noise, stress, and ambient light) on mobile interaction in Chapter 7.

#### 3.1.1 Smartphone Tasks

We quantified the effects of different situational impairments on mobile interaction throughout the studies presented in this thesis on three common smartphone tasks (Figure 3.1):

- Target Acquisition Circular targets of similar size (r = 135px) randomly appear on different parts of the screen according to 4x6 grid [92] with a clearly indicated centre one at a time. Participants were asked to press the centre of the targets as quick and as precise as possible. This task measures participants' reaction time (target access time) in milliseconds (ms) and accuracy of the touch (offset size) in pixels.
- Visual Search Application icon is selected at random and shown to the participant for memorising purposes, and then it needs to be found among other 24 icons randomly distributed on the smartphone screen according to 4*x*6 grid [92]. This task measures participants' memorisation time in milliseconds (ms), visual search time in milliseconds (ms), and error rate.

• **Text Entry** — A text snippet is displayed on the top of the smartphone screen and participants are asked to type in the text following the case and punctuation. This task measures participants' time per character entry in milliseconds (ms), and total error rate [248].

We argue that these three tasks are the most fundamental in mobile interaction. Before the start of the data collection, our participants underwent extensive training to familiarise themselves with the smartphone tasks. Moreover, the order of the tasks was randomised each time when presented to participants. These steps were necessary to reduce any potential learning effects. We used these tasks to quantify the effects of ambient noise, stress, and dim ambient light on mobile interaction presented in Chapters 4, 5, and 6 respectively. However, in the study presented in Chapter 8 we only used target acquisition task as the purpose of the study was not focused on quantifying the effects of cold ambience on mobile interaction, but to propose a sensing mechanism to detect cold-induced situational impairments.



**Figure 3.1:** Interface of the application with Target Acquisition Task (A), Visual Search Task (B-C), and Text Entry Task with user's input (D-E)

## 3.2 Data Analysis

We applied a mixed-methods approach, which implies that we utilised both quantitative and qualitative techniques to analyse the data collected in our studies.

### 3.2.1 Quantitative analysis

Below we provide an overview of the quantitative methods utilised in the data analysis of the research conducted as part of this thesis. We use confidence level of 95% (*i.e.*,  $p - value \le 0.05$ ) to declare the statistical significance and to reject null hypothesis. This is a common convention used in the discipline of Human-Computer Interaction [127].

In Chapter 4 regarding the effect of ambient noise on mobile interaction, we apply statistical modelling. Statistical modelling in data analysis is used to achieve three main purposes [124]:

- Predict an outcome variable;
- Extract information;
- Describe stochastic structures.

In this chapter we utilise statistical modelling to predict the outcome variable and explain the effect of each of the explanatory variables (predictors) on the dependent variable. To be precise, we built generalised linear mixed effects models to predict variables quantifying user performance (*e.g.*, target acquisition time) based on the predictor variables: fixed (*e.g.*, conditions and other contextual factors) and random effects (*e.g.*, participant). Random effects in generalised linear mixed effects models allows us to account for individual variation among participants, thus allowing us to apply the model to a broader inference about the larger population of participants. Furthermore, statistical modelling allows us to consider many variables that may have an effect on participants' performance during mobile interaction. For example, participant being a native English speaker could have an effect on their mobile interaction behaviour when listening to English speech. Similarly, participants' habit to listen to music while interacting with their smartphones could have an effect on their performance when performing the experiment under the music conditions.

In Chapters 5 and 6 we adopt methods from frequentist statistical inference, namely Analysis of variance, commonly known as ANOVA. In particular, we employ repeated measures ANOVA, as we follow within-subjects experimental design in both of the studies presented in Chapters 5 and 6. Frequentist statistical inference methods are broadly used in HCI research to evaluate the differences in outcome variables under varying conditions [127].

In Chapter 8 we use correlation analysis to understand the relationship between two variables: finger temperature and smartphone battery temperature to evaluate how they behave under the cold ambience. Correlation analysis is also widely employed within the HCI research to describe the type of the relationship and the degree of the dependence between a pair of variables [127]. As the data collected in the study presented in Chapter 8 was normally distributed, we used Pearson product-moment correlation to examine the relationship between the observed and manipulated variables [120].

#### 3.2.2 Qualitative Analysis

Together with the quantitative analysis, we applied qualitative analysis to understand the data collected during semi-structured interviews that we held with participants in the studies presented in Chapters 4, 5 and 6. This approach was applicable to our data analysis as participants' perception and opinions about their performance under various conditions throughout the studies presented in Chapters 4, 5 and 6 was important to us in order to either support or contrast results derived from the quantitative analysis. We did not collect any qualitative data in the study presented in Chapter 8 because we were interested in observing a natural physiological reaction of human finger temperature and smartphone battery temperature to the variations in ambient temperature.

In our qualitative analysis we used two main methods – In Vivo Coding [165] (Chapters 4 and 6) and Thematic Analysis [29] (Chapter 5) – to explain our qualitative data. In Vivo Coding is one of the most common methods used in HCI research to analyse qualitative data [165]. It utilises words and phrases taken from participants' own language and is appropriate for studies that honour participants' voice [165].

The process of performing In Vivo Coding is the following. First, we process the raw data by directly transcribing participants' answers and remarks collected during the interview. Then we identify relevant and repeating concepts and label them with a word or a phrase describing the concept. This allows us to identify clusters and segments related to particular research question, hypothesis, or theme [165].

In Chapter 5 to further enhance our qualitative data analysis, we employ Thematic Analysis following the methodology suggested by Braun and Clarke [29]:

- 1. We, first, familiarised ourselves with the data.
- 2. Then we generated codes using In Vivo Coding approach.
- 3. After that, we identified the themes from the codes.
- 4. Next, we reviewed these themes.
- 5. Afterwards, we defined the themes.
- 6. Finally, we presented the themes as part of the results of the study.

We involved more than one researcher to perform In Vivo Coding and Thematic Analysis to increase the confidence and the validity of the qualitative results [22].

## 3.3 Ethical Considerations

We accounted for multiple potential issues with our experiments by employing several precautions to manage risks that might have occurred during the studies presented in Chapters 4, 5, 6, and 8. First of all, we assign each participant with an anonymous ID and the collected data is associated with the participant ID only. We ensure that the participants are not directly identifiable. Furthermore, the soft-

ware used for data collection does not store any sensitive personal information (*e.g.*, name, ethnicity). When storing the data, we use secure servers protected with firewall and control the access to the data through an authentication mechanism. The authentication mechanism requires having a secure password and requests change of the password every 6 months. This is how we solve the issue related to the anonymity of the participants and their data storage.

In each of the studies presented in Chapters 4, 5, 6, and 8 we provided our participants with information in a written Plain Language Statement. Furthermore, each participant's consent was established by signing and returning a Consent Form. We held individual intake sessions with each of the participants where we explained the details of the study and potential risks associated with it. Moreover, participants had the opportunity to read both the Plain Language Statement and a Consent Form before agreeing to participate in the experiment. We commenced the experiment and data collection only after obtaining participant's agreement. Moreover, we excluded any risks that might be harmful to the participants' mental or physical well-being. Precisely, in the work presented in Chapter 4 we limited the sound levels to 55 decibels – an acceptable sound level for everyday life; in the study presented in Chapter 5 we induced mild stress on the participants and excluded involving physical stressors (*e.g.*, electrical shock); in the experiment described in Chapter 8 we exposed our participants to -10°C while providing them with a winter attire. Furthermore, our participants had the right to withdraw from the studies at any point if they wished to do so or felt any discomfort caused by the study protocol.

Finally, we have also accounted for possible dependent relationships between the researchers and participants (*e.g.*, students enrolled in the subjects instructed by any of the research confederates involved in the study) by stating in the Plain Language Statement and in the Consent Form that unwillingness to participate and withdrawal from the study at any point would not have an effect on the students' grades.

## 3.4 Limitations

The methodology presented in this thesis has several limitations. First, the study settings were strictly controlled. In particular, we ran the experiments under the laboratory settings to provide fair comparison of the effects of SIIDs on mobile interaction performance. Nevertheless, we acknowledge that the effects of SIIDs under the naturalistic conditions could be more prominent [231]. For example, participants in our lab studies experienced certain levels of noise (55 decibels), stress, dim light (20 lux), and cold ( $-10^{\circ}$ C); however, in the real world scenario the levels could have been higher (*e.g.*, louder noise, more stress, darker, and colder environment).

Furthermore, we conducted our studies under a limited number of conditions. For example, in Chapter 4 we used such types of ambient noise as music with fast and slow tempo, indoor and outdoor urban noise, and meaningful and meaningless speech. It is entirely possible that in a naturalistic environment ambient noise types would be more diverse, and multiple noise types could occur simultaneously.

Furthermore, in Chapter 5 we strictly controlled the levels of stress induced on our participants. It is possible that in a real world scenario, participants can be exposed to stronger levels of stress. Nevertheless, it was necessary to strictly control stress induction by following the established protocol to avoid causing harm to our participants, and even a mild incidence of stress was sufficient to adversely affect participant's mobile interaction.

We also acknowledge that across the four user studies presented in this thesis, the number of participants did not exceed 28 people. This sample size might be considered low in other disciplines; however, we followed the local guidelines for the HCI discipline. According to HCI user study standards, the total number of participants above 20 is sufficient to perform statistical analysis [42]. Furthermore, due to the fact that we employed a within-subject experimental design, the statistical power of our analysis increases despite the limited number of participants.

Further to this, we used the same smartphone across the studies presented in Chapters 4, 5, and 6. It is possible that using a different device could have led to a higher variability of the results due to different specifications of the device (*e.g.*, screen contrast, screen size). Nevertheless, we argue that controlling for the device was necessary in order to draw fair comparisons between the effects of different SIIDs on mobile interaction performance.

In addition, in Chapter 6 we examined only two levels of ambient illumination – normal and the dim light, and did not study the effect of bright ambient light (*e.g.*, outdoor illuminance) on smartphone interaction performance. However, this exclusion was necessary to eliminate the effect of additional external factors (*e.g.*, glare) on smartphone interaction performance.

In chapter 8 we limited the room temperature to -10° due to safety concerns. Furthermore, we instructed our participants to not wear gloves or warm their hand by any other means, unlike in naturalistic conditions in order to: 1) avoid touch inaccuracies during mobile interaction; and 2) observe a steady finger temperature drop.

Moreover, the types of smartphone tasks presented in the methodology of this thesis were limited to target acquisition, visual search, and text entry. However, in a more naturalistic setting, users may perform more complex tasks, requiring more cognitive demand. Nevertheless, we argue that these tasks are basic for smartphone interaction, and on a more complex tasks, the effects of SIIDs might be more profound. Finally, we restricted our participants to only one interaction mode with the smartphone – using the index finger. We argue that controlling the interaction mode was necessary to draw a fair comparison between the effects of ambient noise, stress, and ambient light, while at the same time not making the individual experiments overly long.

## 3.5 Conclusion

This chapter summarised the primary methodological methods used in the studies presented in this thesis. Overall, it can be concluded that we quantified the effects of different SIIDs on mobile interaction in the laboratory environment with strictly controlled settings to provide fair conditions and comparisons of the effects of different SIIDs on mobile interaction. The aim of the thesis is to quantify the effects of different SIIDs on mobile interaction and contrast their effects as well as to introduce a sensing mechanism to detect cold-induced SIIDs. The following Chapters 4, 5, 6, and 8 present scientific articles on this research topic published at peer-reviewed flagship venues.

# **Chapter 4**

# **Quantifying the Effects of Ambient Noise on Mobile Interaction**

In this chapter we present our work that quantifies the effects of different ambient noise types on mobile interaction. Namely, we asked the participants to complete smartphone tasks (target acquisition, visual search, and text entry) under 4 types of ambient noise conditions:

- Classical music with slow and fast tempo;
- Urban noise outdoor and indoor;
- Speech meaningful (in English) and meaningless (in Kazakh);
- Silence, which acted as a baseline condition.

These are four common ambient noise types that might be present during mobile interaction (*e.g.*, , interacting with the smartphone while listening to music, while standing outside near a noisy road, while being involved in a conversation, or in complete silence). Furthermore, these types of ambient noise have previously been shown to influence human behaviour: music tempo can define the speed of human actions [43, 162, 167], urban noise [18, 46] and speech [155, 226] can adversely affect human memory and cognitive performance. However, their effect on mobile interaction has not been previously investigated.

This article shows that when exposed to music with both fast and slow tempo, participants were significantly quicker and less accurate in completing the target acquisition tasks, compared to the silent condition. Our results also demonstrate that when exposed to urban noise conditions, participants were significantly quicker when completing the target acquisition tasks; however, unlike music conditions, urban noise did not affect participants' accuracy in performing the target acquisition tasks. In addition, there was no statistically significant effect of speech conditions on the target acquisition task.

In regard to the visual search task, our findings show that music and speech did not have a statistically significant effect on participants' performance. However, when exposed to the indoor urban noise condition, participants were significantly faster to memorise icons as compared to the baseline condition.

Finally, with respect to the text entry task, our work shows that the meaningful speech condition, dissimilar to music and urban noise, significantly impaired participants' performance. In particular, participants were slower to type in the provided text when listening to meaningful speech when compared to the baseline.

Overall, this chapter investigates the effects of noise-induced SIIDs. Therefore, our work contributes towards accumulating knowledge in SIIDs research. The detailed motivation, analysis, results and discussion of our approach are presented in the attached publication in Section 4.1.

### 4.1 Publication

©Copyright is held by the authors. Publication rights licensed to ACM. This is the authors' version of the work. It is posted here for your personal use. Not for redistribution. The definitive Version of Record was published in:

Zhanna Sarsenbayeva, Niels van Berkel, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. 2018. Effect of Distinct Ambient Noise Types on Mobile Interaction. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 2, 2, Article 82 (July 2018), 23 pages. DOI: https://doi.org/10.1145/3214285.

Ethics ID: 1750350.1, The University of Melbourne Human Ethics Advisory Group.
# Effect of Distinct Ambient Noise Types on Mobile Interaction

ZHANNA SARSENBAYEVA, The University of Melbourne, Australia NIELS VAN BERKEL, The University of Melbourne, Australia EDUARDO VELLOSO, The University of Melbourne, Australia VASSILIS KOSTAKOS, The University of Melbourne, Australia JORGE GONCALVES, The University of Melbourne, Australia

The adverse effect of ambient noise on humans has been extensively studied in fields like cognitive science, indicating a significant impact on cognitive performance, behaviour, and emotional state. Surprisingly, the effect of ambient noise has not been studied in the context of mobile interaction. As smartphones are ubiquitous by design, smartphone users are exposed to a wide variety of ambient noises while interacting with their devices. In this paper, we present a structured analysis of the effect of six distinct ambient noise types on typical smartphone usage tasks. The evaluated ambient noise types include variants of music, urban noise and speech. We analyse task completion time and errors, and find that different ambient noises affect users differently. For example, while speech and urban noise slow down text entry, being exposed to music reduces completion time in target acquisition tasks. Our study contributes to the growing research area on situational impairments, and we compare our results to previous work on the effect of cold-induced situational impairments. Our results can be used to support smartphone users through adaptive interfaces which respond to the ongoing context of the user.

 $\label{eq:CCS Concepts: Human-centered computing $\rightarrow$ Empirical studies in HCI; $\bullet$ Human-centered computing $\rightarrow$ Ubiquitous and mobile computing; $\bullet$ Human-centered computing $\rightarrow$ Smartphones;$ 

Additional Key Words and Phrases: Smartphones, ambient noise, situational impairments, mobile interaction, performance

#### **ACM Reference Format:**

Zhanna Sarsenbayeva, Niels van Berkel, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. 2018. Effect of Distinct Ambient Noise Types on Mobile Interaction. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 2, 2, Article 82 (June 2018), 23 pages. https://doi.org/10.1145/3214285

## **1 INTRODUCTION**

Research in Ubiquitous Computing has been largely driven by the realisation that the context of use has a substantial influence on the interaction with a system. When non-conventional contextual and environmental factors negatively affect the use of the system, they lead to *situational impairments* [13, 16, 17, 45, 53]. As constant companions for most users, smartphones are frequently used in situations where the user is situationally impaired, *e.g.*, due to cold environments [17], motion [14], or encumbrance [37, 38]. Therefore, one of the key challenges in our community is to enhance the capability of mobile devices to detect situational impairments and to adapt the interaction accordingly.

Authors' addresses: Zhanna Sarsenbayeva, The University of Melbourne, Melbourne, Australia, z.sarsenbayeva@student.unimelb.edu.au; Niels van Berkel, The University of Melbourne, Melbourne, Australia, n.vanberkel@student.unimelb.edu.au; Eduardo Velloso, The University of Melbourne, Melbourne, Australia, eduardo.velloso@unimelb.edu.au; Vassilis Kostakos, The University of Melbourne, Melbourne, Australia, vassilis.kostakos@unimelb.edu.au; Jorge Goncalves, The University of Melbourne, Melbourne, Australia, jorge.goncalves@unimelb.edu.au.

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than the author(s) must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

© 2018 Copyright held by the owner/author(s). Publication rights licensed to ACM. 2474-9567/2018/6-ART82 \$15.00 https://doi.org/10.1145/3214285

#### 82:2 • Z. Sarsenbayeva et al.

In this paper we investigate the effects of one such situational impairment on mobile interaction that remains relatively underexplored [46]—*ambient noise*. Ambient noise is defined as any noise the user is exposed to while their attention is directed at some task or activity [25]. This includes disruptive sounds like the hammering of construction work, the barely perceptive humming of a fridge, the pleasant music playing on headphones, and the disruptive conversation of co-workers. Even though people commonly use their mobile devices in environments where they are exposed to some level of ambient noise (*e.g.*, public places, cafeterias) [25], little is known about their effect on mobile interaction. This paper aims to fill this gap.

Research in Cognitive Science and related fields has highlighted the negative impact that noise has on human behaviour [22, 39], cognitive performance [5, 44, 54], and emotional state [25]. Banbury *et al.* suggest that this effect is mainly caused by the spread of attention while completing a primary task [6]. They state that even if the attention is directed elsewhere, sound is perceived and processed by the brain and, hence, diverts the attention from the main task, causing degradation in performance. Therefore, it is likely that ambient noise can also have an adverse effect on mobile interaction in certain situations. However, due to limited research, it is unclear what is the magnitude of this effect, if there are any differences between different types of noises, and how ambient noise compares to other situational impairments in the context of mobile interaction.

To explore these issues, we investigate how different categories of common ambient noises — *music, urban noise*, and *speech* — affect mobile interaction while performing typical smartphone activities: target acquisition, visual search and text entry. To further unpack the effects of the different types of noise, we explore two variants of each. We presented users with *slow* and *fast* tempo music, as previous work has shown that different music tempos have varying effects on cognitive performance [22, 39]. We also presented users with two types of urban noise: *outdoor* and *indoor*. Outdoor urban noise has been shown to have a significant adverse effect on memory tasks. However, there is no consensus regarding the effect of indoor urban noise: some studies report its distracting effect during memory tasks [5, 54], while others did not observe any effect on cognitive performance [44]. Finally, we presented users with two types of speech: *meaningful* and *meaningless* speech. Meaningful speech is speech is speech is a familiar language, whereas meaningless speech is defined as the speech presented in a language unfamiliar to the listener [32]. To contextualise our study within the body of research on situational impairments, we compare our findings to previous work on cold-induced situational impairments [45], which collected similar data to the one described in our study.

In summary, this paper advances the state-of-the-art on situational impairment research in Ubiquitous Computing through three main contributions. First, we investigate how different types of ambient noise affect the interaction with mobile devices during three typical activities: target acquisition, visual search, and text entry. Second, we contribute towards the situational impairments research agenda by comparing our findings to the effects of cold-induced situational impairments. Third, we discuss and provide recommendations on the detection of certain types of ambient noise that can affect mobile interaction and how to accommodate a situationally impaired user in such cases.

# 2 RELATED WORK

# 2.1 Impact of Situational Impairments on Mobile Interaction

Previous research has shown that interaction with mobile devices can be adversely affected by implicit environmental and contextual factors, subjecting the user to what is known as a *situational impairment* [46, 49]. A number of causes for situational impairment have been studied within the HCI/UbiComp community, including the effects of ambient temperature [17, 45], motion [14], and encumbrance [37, 38]. Regarding cold-induced situational impairments, Goncalves *et al.* and Sarsenbayeva *et al.* showed a negative effect of cold temperatures on smartphone input performance [17, 45]. The authors found that colder temperatures are associated with lower throughput and accuracy in tapping tasks. Other studies investigated the effect of user motion on interaction

with mobile phones. For example, walking has been found to adversely affect mobile interaction in completing typing tasks [14, 36], and target acquisition tasks [48]. Furthermore, Ng *et al.* [37, 38] demonstrated the negative impact of encumbrance on target acquisition tasks during mobile interaction. In particular, encumbrance has been shown to decrease accuracy, while increasing error rate and target selection time [38]. Given the effect situational impairments have on mobile interaction, previous work was conducted to explore ways of detecting situational impairments and develop design solutions to accommodate them. For example, Sarsenbayeva *et al.* [47] suggest using smartphone's battery temperature to detect ambient temperature drop. Similarly, Goel *et al.* [14] successfully utilise a smartphone's accelerometer sensor to detect walking. In terms of design solutions, researchers have shown that increasing target size [48], providing audio guidance [52] and adaptive text entry [14] can compensate the negative effect of situational impairments.

In comparison to other causes of situational impairment, the effect of ambient noise on mobile interaction lacks appropriate investigation [46]. Previous work emphasises the importance of studying unexplored causes of situational impairment in order to broaden the research scope in this area [46]. There is little prior work that has studied the effect of ambient noise on mobile interaction. A study by Hoggan *et al.* [21] showed that loud noise levels had an adverse effect on participants' performance when completing text entry tasks on mobile phones. However, the aim of their study was simply to identify the noise threshold at which the audio and tactile feedback become ineffective. Harvey & Morgan [18] demonstrated that noisy environments have a negative effect on user performance during web search tasks. They showed an effect of noise on task performance using questionnaire data only and did not provide any quantitative metrics regarding performance. To the best of our knowledge, this is the first study to investigate and quantify the effect of ambient noise as a potential cause of situational impairment during mobile interaction.

# 2.2 Effect of Ambient Noise on Human Behaviour and Cognitive Performance

Previous work has shown that ambient noise has an effect on human daily activities in terms of behavioural [29, 33, 39], emotional [25], and cognitive [5, 54] performance. As mobile interaction is now an essential part of people's everyday activities, it is important to investigate its impact on mobile interaction. Next, we discuss related work on the impact of music, urban noise and speech on human behaviour and cognitive performance.

2.2.1 Effect of Background Music on Human Behaviour and Cognitive Performance. Previous works have investigated the effect of music on people's physical activities [33], cognitive performance [26], and mood [25]. Cassidy *et al.* showed a significant negative effect of heavy metal music on immediate recall, free recall, delayed recall and performance in Stroop tasks [15] compared to a silent condition [10]. Similarly, Wen *et al.* demonstrated that performance in recall is significantly higher under classical music in contrast to rock music [55]. However, there is no consensus regarding the effect of background music on reading [25]. Whereas some studies report an increase on reading performance [41], other studies failed to replicate this effect [32]. Another study showed that slow tempo music resulted in longer reading time and poorer reading efficiency in contrast to fast tempo music [25]. Henderson *et al.* 's results suggest that what is detrimental is not the music itself, but the speech contained in it [19]. The authors found that participant performance in completing reading tasks significantly decreased when they were exposed to popular music containing lyrics, but found no significant decrease with classical music or silence [19]. Given these results, we evaluate the effects of music and speech in separate study conditions.

The most widely studied musical element in terms of the effect of music on human behaviour and cognitive performance has been the *tempo*, *i.e.*, the speed of the underlying beat of the music. Related work suggests that the faster the tempo, the faster people complete tasks. Milliman showed that faster tempo increased the walking speed of customers, decreasing the average time customers spent in a store [34]. Research in restaurants and cafeterias has shown that faster music also caused people to eat and drink faster [9, 33, 35]. Conversely, slower

#### 82:4 • Z. Sarsenbayeva et al.

music was shown to slow down people's eating and drinking [9, 33, 35] as well as to prolong the perception of elapsed time spent in casinos [40].

Despite accelerating task completion, previous research suggests that music with a faster tempo is also more cognitively demanding. For instance, Holbrook showed that fast tempo music demanded more cognitive resources to be processed compared to slow tempo music [22]. According to North *et al.*, this phenomenon occurs because more information is perceived and processed by the listener's brain when fast tempo music is played than when slow tempo music is played [39]. Given the different effects of the tempo of the music, we evaluated the effect of music on mobile interaction with fast and slow tempo music.

An interaction effect between the tempo of the music and the gender of the listener has also been identified. For example, Kallinen showed that female and male participants perceived information differently under slow and fast music conditions [25]. In this study, the author found an inverse effect of the tempo of the music on the perception of news articles — a slow tempo led male participants to evaluate the news as negative, whereas it led female participants to evaluate the same set as positive. Further, men read more slowly when slow tempo music was playing, whereas women read more slowly when no music was playing. This suggests that it is important to ensure appropriate gender balance in the participant sample to avoid any gender bias in the results.

Taking into consideration the effect of music tempo on other aspects of human behaviour and cognitive performance, we chose to investigate its effect on mobile interaction. We hypothesise that fast tempo music will result in a faster completion of tasks when compared to the silent condition, particularly those that require fine-motor movements (*e.g.*, target acquisition tasks).

2.2.2 Effect of Urban Noise on Human Behaviour and Cognitive Performance. As with background music, urban ambient noise is also known to influence human behaviour and cognitive performance. In this study, we focus on two types of urban noise: outdoor and indoor. We define outdoor urban noise as noise containing sound as sampled in an urban area: a combination of street, construction, and traffic sounds. We define indoor urban noise as the noise coming from an office or cafeteria: working sounds, people's murmur, and chatter.

A negative effect of urban outdoor noise on human performance has been shown in several studies. For example, Cassidy *et al.* showed a significant negative effect of outdoor urban ambient noise on free, immediate, and delayed recall and Stroop tasks [15] when compared to no noise [10]. Furthermore, Stansfeld *et al.* showed that communities exposed to lower traffic noise had a lower rate of psychiatric hospitalisation [50]. Previous work has also shown that an exposure to a broadband noise at 100 dB for 30 minutes can lead to significant impairments when performing memory tasks [2].

Regarding urban indoor noise, Banbury and Berry demonstrated that students' performance in completing mathematical and recall tasks was significantly worsened with its presence [5]. Furthermore, a two-year lon-gitudinal study by Cohen and Weinstein showed that children in noisy schools with sound levels of 60-80 dB performed worse in solving puzzles and mathematical tasks [11], as reported by Holmberg and Coon [23].

This literature coupled with the fact that it is common practice for mobile device users to interact with their devices while being exposed to urban outdoor (*e.g.*, streets) and indoor (*e.g.*, office, cafeteria) noise, suggests that these types of noises can have an effect on mobile interaction, which we investigate in this paper.

2.2.3 Effect of Speech on Human Behaviour and Cognitive Performance. Previous research has made a distinction between the effects of two types of speech on human behaviour and cognitive performance [32]: meaningful — speech that the listener can understand — and meaningless — speech that the listener does not understand. For instance, Martin *et al.* found a detrimental effect of continuous meaningful speech on reading performance when compared to silence [32]. The authors compared the effect of meaningful and meaningless speech on reading comprehension and showed that reading performance is significantly worse under both speech conditions as compared to silence, with a stronger effect for meaningful speech [32].

Previous work has also shown that both meaningful and meaningless speech have an equally distracting effect on memory recall [51]. Tremblay *et al.* found that natural speech was more disruptive than sine wave speech (*i.e.*, a form of artificially degraded speech) on tasks requiring memory recall [51]. Moreover, in a study reported in [43] meaningless speech was shown to have a negative effect on memory tasks. Salame *et al.* showed that immediate memory was disrupted the most by the the unattended foreign language speech, when compared to other conditions such as instrumental music, urban noise and silence [44]. These results contradict previous findings that did not find any effect of meaningless speech on cognitive performance in phonological judgement tasks [3]. Nevertheless, the authors argue that the phenomenon needs further investigation [3]. Finally, speech is also known to reduce performance in completing arithmetic tasks and memory for prose tasks [4, 54].

In this paper we explore the effect of both meaningful (English) and meaningless (Kazakh) speech on mobile interaction. Based on the literature, we hypothesise that the performance in tasks with higher cognitive requirements (*e.g.*, text entry) is likely to deteriorate while hearing background speech.

# 3 STUDY

We operationalise mobile interaction in terms of three common activities conducted on smartphones: target acquisition, visual search, and text entry. For this study we used two software applications as experimental tasks, *TapCircle* and *FindIcon*, previously used to investigate the effect cold-induced situational impairments on mobile interaction [45]. This allows for direct comparison of our results to those previous findings. Additionally, we developed a new custom software called *TypeMe*. The tasks presented by TapCircle measures users' fine-motor performance during target acquisition, the task in FindIcon measures users' cognitive performance during visual search [45], and the task in TypeMe measures users' text entry performance. The three tasks were combined into one Android application and were presented in a random order. Details for each task are presented below, the application interface is shown in Figure 1.



Fig. 1. Interface of the application with TapCircle task (A), FindIcon task (B-C), and TypeMe task with user's input for easy and difficult texts (D-E)

# 3.1 Tasks

3.1.1 Task 1: Target Acquisition. Circular targets with a radius of 135 pixels randomly appear one at a time in each position of a 4×6 grid at least once. A similar task was used by Henze *et al.* [20]. Every target has an indicated centre and participants were instructed to tap these circles as quickly and as precisely as possible. The

# 82:6 • Z. Sarsenbayeva et al.

application logged the coordinates of the target's centre and participant's touch point, the elapsed time, and the position of the circle on the grid. Figure 1-A shows a screen of the TapCircle task interface.

3.1.2 Task 2: Visual Search. In the FindIcon task, participants were first presented with an icon which they subsequently had to locate within a grid of other icons. Participants were free to look at the target icon for as long as they wanted to memorise it. In the following screen, participants were required to locate and tap the target icon amongst a set of 24 icons ( $100 \times 100$ px) in a 4×6 grid. To minimise any possible learning effects, each of the 24 icons was designated as a possible target icon in random order in each condition. The application also ensured that every grid position hosted a target icon in a random order. To make our results comparable, we used the same set of application icons as described by Sarsenbayeva *et al.* [45]. Figures 1-B-C show the interface of the target icon, grid position of the icon, and the coordinates of the centre of the icon and of the participant's touch point.

3.1.3 Task 3: Text Entry. In the text entry task, the application presented some text at the top of the screen and participants were asked to type it verbatim in the text-box below it. In order to have varying complexity, we distinguished between easy and difficult texts. Easy texts consisted of only one sentence and contained common words that are used on a daily basis. Difficult texts were selected from Shakespeare's sonnets and consisted of more than 1 sentence. We quantified the difficulty of the sentences using the Flesch-Kincaid readability test [28]. The easy texts had an average Flesch-Kincaid grade level of 1.6, whereas the difficult texts had an average Flesch-Kincaid grade of 5.1. Participants completed both an easy and a difficult text during the task. Figures 1-D-E show the task as presented to the participants, including examples of an easy and a difficult text.

# 3.2 Hardware

Participants completed all tasks using a Samsung Galaxy S7 smartphone running Google's Android 7.0 (Nougat). The smartphone has a 5.1-inch screen with a resolution of 1080×1920px. The smartphone was selected due to identical screen size and resolution to the smartphone's screen parameters used in Sarsenbayeva *et al.* 's evaluation of cold-induced situational impairments [45]. This allowed us to directly compare the effects of noise-induced situational impairments to cold-induced situational impairments on mobile interaction.

# 3.3 Experimental Conditions

We selected the following experimental ambient noise conditions for this study given their demonstrated effect on human behaviour and cognitive performance in the literature:

- music (fast and slow tempo)
- urban ambient noise (indoor and outdoor)
- speech (meaningful English, meaningless Kazakh)
- silence, which acted as a control condition.

In the music conditions we used the same composition sampled at fast and slow rates to avoid other characteristics of the music, other than tempo (*e.g.*, pitch, timbre), to affect the participants performance. This is because tempo is considered to be the main factor in music to affect human behaviour and performance [25]. We chose music that did not contain any lyrics in order to avoid overlap with the speech condition. Given these guidelines, we selected Bach's "Brandenburg Concerto No. 2". As the original tempo of the composition is considered to be fast (92 beats per minute) [25], we used it as is for the fast tempo music condition. For the slow tempo music condition we sampled the same composition at 60 beats per minute, below the 70 beats per minute considered in the literature to be slow [25], without any noticeable decrease in sound quality.

As for the urban outdoor ambient noise condition, we chose a clip of street noise containing road traffic, vehicles' motor sounds, honking, and indistinguishable crowd speech. For the urban indoor condition we chose a

cafeteria noise composed of coffee machine, cutlery sounds, and indistinguishable people's murmur. Both of the sounds were obtained from YouTube (urban outdoor  $^1$ , urban indoor  $^2$ ).

For the meaningful speech condition, we selected a weather forecast presented in English. For the meaningless speech condition we selected TV news presented in a language that participants did not understand (Kazakh). Both clips were narrated by a female voice and did not contain any other sounds (*e.g.*, background noise, music, speech of different people).

The volume level for all of the ambient noise conditions was kept in the range of 55–60 dB following Iwanaga and Ito's design guidelines, to avoid discomfort to our participants [24]. All audio files were long enough for participants to complete all tasks before the audio clip ended.

# 3.4 Participants and Procedure

We recruited twenty-four participants (12 male, 12 female) through our university's mailing lists. We balanced gender as the literature suggests men and women can react differently to the same noise conditions [25]. Participants were aged between 19 and 54 years (M = 31.67, SD = 8.95) and had a diverse range of educational backgrounds (*e.g.*, Accounting, Geology, Linguistics, Biochemistry, Computer Science, Elderly Care). All participants were fluent in English and used it as their main language of communication at work. The foreign language presented in this study ('meaningless speech') was not known to any of the participants. Each participant was assigned a unique anonymous ID (participant ID) in our study.

Our study had a within-subjects experimental design. Condition acted as an independent variable and the order of conditions presented to the participants was counter-balanced. This way, we minimised the impact of any potential fatigue or learning effects. Unlike the study protocol presented in [45], participants completed all tasks in only one smartphone holding posture as opposed to two different holding postures. We instructed our participants to complete all tasks in two-handed interaction mode while standing (*i.e.*, interacting with the phone with index finger of the dominant hand while holding the phone in the non-dominant hand [45]). Figure 2 shows the participant completing a task during the experiment. As each participant was required to complete three tasks in each of seven conditions, including a second smartphone holding posture (and essentially doubling study duration) would have likely introduced considerable fatigue and irritation among our participants. Each participant was rewarded with a \$10 voucher for participation.

We collected data for several dependent variables to measure performance while conducting the tasks. In the target acquisition task, we recorded the time taken to tap a circular target and the size of the tap offset. In the visual search task, we recorded the time taken to find the correct icon, the time taken to memorise an icon, and the number of incorrectly selected icons. Finally, in the text entry task, we recorded the time taken to complete typing a text, the number of characters per text, and the number of errors within the text that were not corrected by participants. We measured the typing performance with the time taken per character entry by dividing the total typing time by the number of characters per text.

Upon arrival at our usability lab, participants were briefed about the purpose of the study and asked to sign a consent form. We then collected their personal details (age, gender, background, native language, dominant hand) and performed a training procedure in order for the participants to get acquainted with the tasks. During the training, participants completed all three study tasks in a random order, until they were comfortable with each one, in order to minimise any potential learning effects.

Once the training was completed, participants started the experiment. As the study contained four main conditions (silence, music, urban noise, speech), we created 24 combinations of the conditions to ensure proper counter-balancing. This means that no participant had a repeated order of conditions presented to them. Three

<sup>&</sup>lt;sup>1</sup>https://www.youtube.com/watch?v=cDWZkXjDYsc

<sup>&</sup>lt;sup>2</sup>https://www.youtube.com/watch?v=lSlOXT3AKBo

#### 82:8 • Z. Sarsenbayeva et al.





(a) Laboratory setting of the experiment

(b) Smartphone holding posture

Fig. 2. Participant completing tapping circles task

of the conditions included sub-conditions (music fast/slow tempo, urban noise outdoor/indoor, speech meaningful/meaningless). We randomly allocated the participants into two groups. Within each of these conditions, the first group was exposed to fast music tempo, indoor urban noise, and meaningful speech conditions first. The second group, opposite to the first group, was exposed to slow music tempo, outdoor urban noise, and meaningless speech first. Both groups experienced both sub-conditions.

During the experiment, participants had to complete the same set of tasks in each of seven conditions. Each round of tasks consisted of: target acquisition, visual search, and text entry. The order of the tasks was randomised by the application, and the participants had to complete each task one after another without taking any breaks. Once the sound of a condition started playing, the researcher started a timer. After being exposed to the sound for one minute, the participant was then instructed to begin that round of tasks. Listening to the sound for one minute ensured participants were more accustomed to that particular ambient noise condition [17, 45].

After the experiment was completed, we conducted a short semi-structured interview. Participants were asked about their personal perception of their performance during the conditions, whether any of the sounds were experienced as particularly distracting, and whether they usually listen to music or any other sounds while working, reading, or performing tasks that demand concentration. Finally, we enquired if they listen to music or other type of audio when they interact with their smartphones. The experiment lasted approximately 70 minutes per participant, including briefing, training, data collection, and final interview.

# 4 RESULTS

From our 24 participants, we collected 15,556 and 4,094 target hits for the tapping the circles and finding icons task respectively, and 336 typed sentences for the typing task. Data collected from left-handed participants (N = 2) was mirrored relative to the X-axis of the screen for the tapping a circle and finding an icon tasks.

To investigate whether mobile interaction was affected by ambient noise, we built generalised linear mixedeffect models to describe participant performance in the three tasks (target acquisition, visual search, and text entry). Apart from the ambient noise conditions (discussed in Subsection 3.3), we also considered other factors that may affect participants' performance during mobile interaction. We provide a list of these variables below. Unless otherwise stated, the variable was included for all conditions (silence, music, urban noise, and speech).

- Gender A binary variable indicating gender of the participant.
- Age A numeric variable indicating the age of the participant.
- **Music listener** A binary variable indicating whether the participant typically listens to music or other sounds while working, reading, or performing tasks that require cognitive demand. We only used this predictor in models predicting the effect of music on mobile interaction.

- Native English speaker A binary variable indicating if the participant is a native English speaker. We only used this predictor in models predicting the effect of speech on mobile interaction.
- X centre coordinate X-axis coordinate of the centre of target (circle or icon for tapping a circle and finding an icon task respectively).
- Y centre coordinate Y-axis coordinate of the centre of target (circle or icon for tapping a circle and finding an icon task respectively).
- **Participant ID** Participant ID was treated as a random effect in order to control for individual differences in our models.

We applied a backfitting algorithm using AIC to all our models for predictor selection. AIC penalises the inclusion of additional predictors, which could lead to overfitting of a model. Finally, to ensure the validity of the models we checked for the presence of multicollinearity. All of the predictors for each of the presented models had a variance inflation factor between 1 and 1.38, well below the often used threshold of 5 to 10 to detect multicollinearity.

# 4.1 Results: Background Music Condition

The tested background music conditions are fast tempo music, slow tempo music, and silence. The conditions are tested across all three defined tasks. We discuss the results per task below.

4.1.1 Target Acquisition. Our first model describes the time taken to hit a circle. The final prediction model retained 3 of the 6 predictors, as shown in Table 1. The model is statistically significant ( $\chi^2(4) = 56.05$ , p < 0.01) and describes 7% of variance of the time taken to hit the circle (*Marginal*  $R^2 = 0.07$ , *Conditional*  $R^2 = 0.28$ ). The results indicate that participants were significantly quicker to tap circles in the fast and slow music conditions as compared to the silent condition. However, we did not find a significant difference between the fast and slow music condition; Wilcoxon signed-rank test (V = 1.17e+06, p = 0.21). A boxplot of the time taken to complete the tapping circles task across the different conditions is shown in Figure 3 (a).

	Estimate	Std. Error	t value	Pr (> t )
(Intercept)	4.63e+02	4.55e+01	10.19	7.68e-10 ***
Condition (Fast music)	-1.51e+01	3.41	-4.43	9.61e-06 ***
Condition (Slow music)	-2.01e+01	3.41	-5.91	3.71e-09 ***
Y centre coordinate	8.76e-03	2.62e-03	3.35	0.001 ***
Age	3.98	1.38	2.89	0.008 **

Table 1. Effects of model factors on predicting time taken to hit a circle in music condition

Significance: '\*\*\*' <0.001, '\*\*' <0.01, '\*' <0.05

Mean values for the offset size per condition for the tapping task are visualised in Figure 3 (b). We built a model to describe the mean offset size when tapping a circle. The final model contains both conditions and the X centre coordinate as predictors, and is summarised in Table 2. The model was significant ( $\chi^2(4) = 63.43$ , p < 0.01, *Marginal*  $R^2 = 0.01$ , *Conditional*  $R^2 = 0.10$ ). Participants were significantly less accurate when tapping the circles in the slow music condition as compared to the silent condition. A Wilcoxon signed-rank test did not show a significant difference between the offset size in the fast tempo music condition and the slow tempo music conditions (V = 1183600, p = 0.21). From Table 2, we observe that the offset is smaller for the circular targets located closer to the edge of the screen on which participants held their phone.





Fig. 3. Mean values for time taken to tap a circle and offset size per condition

Table 2. Effects of model factors on predicting the offset size when tapping a circle in music condition

	Estimate	Std. Error	t value	Pr (> t )
(Intercept)	4.47e+01	2.22	20.17	<2e-16 ***
Condition (Fast music)	1.58	1.01	1.57	0.12
Condition (Slow music)	2.74	1.01	2.72	0.007 **
X centre coordinate	-1.01e-02	1.34e-03	-7.52	6.22e-14 **

Significance: '\*\*\*' <0.001, '\*\*' <0.01, '\*' <0.05

4.1.2 Visual Search. Next, we built a model to describe the time taken to memorise an icon. The final prediction model contained only age as a predictor. The direction of age was negative, indicating that younger participants took less time to memorise an icon. The model was statistically significant ( $\chi^2(1) = 12.26$ , p < 0.01) and described 9% of variance of time taken to find an icon (*Marginal*  $R^2 = 0.09$ , *Conditional*  $R^2 = 0.22$ ).

We built another model to describe the time taken to find an icon. The final model again contained only age as a predictor, and direction was again negative (indicating that younger participants took less time). The model was statistically significant ( $\chi^2(1) = 18.22$ , p < 0.01) and described 2% of variance of the time taken to find an icon (*Marginal*  $R^2 = 0.02$ , *Conditional*  $R^2 = 0.08$ ).

Finally, we built a model to describe the number of errors made by participants when finding an icon. However, none of the predictors sufficiently described this dependent variable. We were thus unable to create a model for this variable.

*4.1.3 Text Entry.* The last task for the background music condition is the typing task. We built a model to predict the time per character entry. However, we were unable to create a model for this variable as none of the predictors sufficiently described it.

Following this, we build a model to describe the number of errors made by participants when typing a message. The final model contained one predictor (participant age). Participants of younger age made fewer errors. The model was statistically significant ( $\chi^2(1) = 8.55$ , p < 0.01) and described 12% of variance of the number of uncorrected errors (*Marginal*  $R^2 = 0.12$ , *Conditional*  $R^2 = 0.31$ ).

# 4.2 Results: Urban Noise Condition

The tested urban noise conditions are outdoor, indoor, and silence. The conditions are tested across all three defined tasks. We discuss the results per task below.

4.2.1 Target Acquisition. We first built a model to describe the time taken to tap a circle. The final model was described with three predictors: condition, Y centre coordinate of the circle, and participant age. We provide a summary of the factors in Table 3. The model was statistically significant ( $\chi^2(4) = 54.73$ , p < 0.01) and described 8% of variance of the time taken to tap a circle (*Marginal*  $R^2 = 0.08$ , *Conditional*  $R^2 = 0.27$ ). Participants took significantly less time to tap a circle in both indoor and outdoor noise conditions, with the stronger effect of urban outdoor noise. Wilcoxon signed-rank test did not show a significant difference between the indoor and outdoor noise conditions for time taken to tap circle (V = 1.26e+06, p = 0.81). We visualised mean values for the time taken to tap a circle per condition in Figure 4. Table 3 shows that older participants required more time to tap a circle, similar to our observation in the music conditions. Furthermore, we show that the further from the top left corner the target is, the longer the target acquisition time is.

Table 3. Effects of model factors on predicting time taken to tap a circle in urban noise condition

	Estimate	Std. Error	t value	Pr (> t )
(Intercept)	4.68e+02	4.03e+01	11.61	6.42e-11 ***
Condition (Indoor noise)	-7.89	3.09	-2.55	0.011 *
Condition (Outdoor noise)	-19.4	3.12	-6.21	5.65e-10 ***
Y centre coordinate	6.33e-03	2.37e-03	2.67	0.007 **
Age	3.85	1.22	3.14	0.005 **

Significance: '\*\*\*' <0.001, '\*\*' <0.01, '\*' <0.05



Fig. 4. Mean values for time taken to hit a circle per condition

We then built a model to predict the offset size. The final model contained the X centre coordinate of a circle as its only predictor. The direction of the predictor is negative, indicating that the closer to the right edge of the screen the circle is located, the smaller the offset of the touch is. The model was statistically significant ( $\chi^2(1) = 33.77$ , p < 0.01) describing 0.5% of variance of the offset size (*Marginal*  $R^2 = 0.005$ , *Conditional*  $R^2 = 0.07$ ).

#### 82:12 • Z. Sarsenbayeva et al.

4.2.2 Visual Search. To check the effect of the urban noise conditions on performance during icon search task, we built a model to predict the time taken to memorise an icon. The final model contained two predictors: condition and participant age. The model was significant ( $\chi^2(3) = 21.33$ , p < 0.01) and described 8% of variance of time taken to memorise an icon (*Marginal*  $R^2 = 0.08$ , *Conditional*  $R^2 = 0.22$ ). The coefficients of the predictors are summarised in Table 4. From the table, we observe that participants spent significantly less time memorising an icon in the urban indoor noise condition compared to the silent condition. However, the effect of urban outdoor condition did not significantly affect the time taken to memorise an icon. Mean values for time taken to memorise an icon per condition can be found in Figure 5.

Table 4. Effects of model factors on predicting time taken to memorise an icon in urban noise condition

	Estimate	Std. Error	t value	Pr (> t )
(Intercept)	510.38	69.36	7.36	2.01e-07 ***
Condition (Indoor noise)	-36.48	12.33	-2.96	0.003 **
Condition (Outdoor noise)	-5.18	12.32	-0.42	0.67
Age	7.55	2.10	3.593	0.002 **

Significance: '\*\*\*' <0.001, '\*\*' <0.01, '\*' <0.05



Fig. 5. Mean values for time taken to memorise an icon per condition

Next, we built a model to describe the total time taken to find an icon in the urban noise conditions. The final model contained the X centre coordinate of the icon as its only predictor. The predictor was again negative, indicating that the icons were quicker to find the closer they were located to the right edge of the screen. The model was statistically significant ( $\chi^2(1) = 9.32$ , p < 0.01) and described 0.5% of variance of time taken to find an icon (*Marginal*  $R^2 = 0.005$ , *Conditional*  $R^2 = 0.09$ ).

Finally, we built a model to describe the number of errors made by participants in the finding an icon task. The model was described with three predictors; condition and both the X and Y centre coordinates of the icons. The model was statistically significant ( $\chi^2(4) = 16.45$ , p < 0.01) and explained 1% of variance of the number of errors participants performed during icon search task (*Marginal*  $R^2 = 0.01$ , *Conditional*  $R^2 = 0.04$ ). The model is summarised in Table 5.

Proc. ACM Interact. Mob. Wearable Ubiquitous Technol., Vol. 2, No. 2, Article 82. Publication date: June 2018.

Table 5. Effects of model factors on predicting errors in finding an icon in urban noise condition

	Estimate	Std. Error	t value	Pr(> t )
(Intercept)	1.89e-02	1.63e-02	1.16	0.25
Condition (Indoor noise)	6.44e-03	1.19e-02	0.54	0.59
Condition (Outdoor noise)	2.60e-02	1.19e-02	2.18	0.03 *
X centre coordinate	-4.42e-05	1.71e-05	-2.59	< 0.01 **
Y centre coordinate	2.08e-05	9.90e-06	2.10	0.04 *

Significance: '\*\*\*' <0.001, '\*\*' <0.01, '\*' <0.05

The mean values for the number of errors in the icon search task are visualised in Figure 6. According to our findings, participants were significantly less accurate in the urban outdoor noise condition when compared to silent condition. The effect of X and Y centre coordinates is again in line with our previous findings.



Fig. 6. Mean values for errors in find an icon task per condition

4.2.3 Text Entry. We then built a model describing the time required per character entry. The final prediction model contained both condition and gender as predictive variables. The model was statistically significant  $(\chi^2(3) = 10.79, p = 0.01)$  and explained 14% of variance of the time per character entry participants spent during the typing task (*Marginal*  $R^2 = 0.14$ , *Conditional*  $R^2 = 0.49$ ). We presented the summarised model in Table 6. We then visualised the values of the mean time per character entry in a boxplot in Figure 7. Finally, we built a model to predict the number of errors made by participants during the typing task. None of the included predictors sufficiently describe this variable.

Table 6. Effects of model factors on predicting time per character entry in a typing task in urban noise condition

	Estimate	Std. Error	t value	$\Pr(> t )$
(Intercept)	376.59	47.16	7.99	5.17e-09 ***
Condition (Indoor noise)	26.57	32.67	0.81	0.42
Condition (Outdoor noise)	69.50	32.67	2.13	0.04 *
Gender (M)	155.59	61.13	2.55	0.02 *

#### 82:14 • Z. Sarsenbayeva et al.



Fig. 7. Mean values for time per character entry in urban noise condition

# 4.3 Results: Speech Condition

4.3.1 Target Acquisition. First, we built a model to describe the time taken to tap a circle. The final model contained the Y centre coordinate of the circle and participant age as its predictors. The relationship of the Y centre coordinate was positive, indicating that the lower the circle was from the top left corner of the screen, the quicker it was to tap. For age, we see again the same trend: young participants were quicker in tapping the circles. The model was statistically significant ( $\chi^2(1) = 105.39$ , p < 0.01) describing 7.6% of variance of time taken to tap a circle (*Marginal*  $R^2 = 0.08$ , *Conditional*  $R^2 = 0.26$ ).

We then built a model to describe the offset size of the touch. The final model contained only the X centre coordinate of the circle as a factor. As before, the predictor was negative, indicating that the icons were quicker to find the closer they were located to the right edge of the screen. The model was statistically significant ( $\chi^2(1) = 41.85$ , p < 0.01) and described 0.6% of the variance of the offset size (*Marginal*  $R^2 = 0.006$ , *Conditional*  $R^2 = 0.06$ ).

4.3.2 Visual Search. We created a model to describe the time taken to memorise an icon. The final model contained only participant age as a predictor, again in the same direction - indicating that young participants were quicker to find the icons. The model was statistically significant ( $\chi^2(6) = 12.31$ , p < 0.01) and explained 7% of variance of time taken to memorise an icon (*Marqinal*  $R^2 = 0.07$ , *Conditional*  $R^2 = 0.17$ ).

Finally, we construct a model to describe the time taken to find an icon and the corresponding number of errors. However, none of the predictors sufficiently described this dependent variable.

4.3.3 Text Entry. To investigate the effect of speech condition on a typing task, we built a model describing the time needed per character entry. The model was statistically significant ( $\chi^2(3) = 9.30, p < 0.03$ ) and described 9% of variance of time taken to type a character (*Marginal*  $R^2 = 0.09$ , *Conditional*  $R^2 = 0.40$ ). Coefficients of the model are summarised in Table 7. As can been seen from Table 7, participants were significantly slower when typing a character under meaningful speech condition, compared to silent condition. We drew a boxplot to visualise the values of the mean time per character entry for each condition in Figure 8.

We then built a model describing the number of uncorrected errors in a typing task. However, none of the predictors were descriptive enough to predict the number of uncorrected errors during the typing task under the speech condition.

Table 7. Effects of model factors on predicting time per character entry in speech condition

	Estimate	Std. Error	t value	Pr(> t )
(Intercept)	396.01	47.56	8.33	1.12e-09 ***
Condition (Speech meaningful)	85.44	36.53	2.34	0.02 *
Condition (Speech meaningless)	53.09	36.76.20	1.44	0.15
Gender (M)	116.75	60.31	1.94	0.07

Line being the speech meaningful meaningless Condition

Significance: '\*\*\*' <0.001, '\*\*' <0.01, '\*' <0.05

Fig. 8. Time per character entry in speech condition

# 4.4 Results: Interviews

The qualitative analysis presented in this section reflects on participants' answers to the short interview session that we held after they finished the experiment. Several participants (4 out of 24) mentioned that they felt their "overall performance was worse towards the end of the experiment". In contrast to this opinion, 3 out of 24 participants mentioned that their performance was better towards the end of the experiment as they were "more used to the tasks".

4.4.1 Perceived Task Performance in Music Condition. Participants felt the effect of music was most prominent in the tapping task. 12 out of 24 participants claimed that they followed the tempo of the music when tapping circles. The most popular comment we received from the participants was "I was tapping circles in rhythm with the music tempo". Some of the participants thought that the music improved their performance: "Classical music improved my performance" (P24), "I was performing better in finding icons under both music conditions" (P21). Interestingly, participants mentioned that in both of the music conditions they tapped on circles faster when compared to the silent condition: "I was fast in both music conditions. But even with the slow music I was performing faster than in silence" (P14, P20, P24). The participants also mentioned that the music made their task completion experience more enjoyable: "Music made the task easier psychologically, I was not stressed or worried" (P01), "Slow music was relaxing" (P06), "Clicking the circles was fun with the music" (P14), "I was enjoying the music and focused on the task better" (P24). Overall, interview remarks demonstrate that the participants noticed the influence of music tempo on completing target acquisition task. However, there were exclusive remarks from our participants who believed that music was annoying and distracting (P08, P10, P17) and, hence, might have deteriorated their performance: "Slow music was annoying and distracting. I could not concentrate on finding an icon and was slower when typing a message" (P10). However, upon checking the quantitative data we found that mean time taken to

#### 82:16 • Z. Sarsenbayeva et al.

find an icon on the visual search task during the slow music condition for participants P08 (M = 1166.21 ms, SD = 257.99) and P17 (M = 1359.12 ms, SD = 462.27) was less than the mean time for all participants (M = 1753.19 ms, SD = 1166.21). Only participant P10 took longer time finding an icon under the slow music condition (M = 1860.92 ms, SD = 1098.26) compared to the total mean time for the visual search task. We also found that all three participants took less time per character entry in the text entry task (M = 431.56 ms, SD = 59.26; M = 379. 44 ms, SD = 67.38; and M = 424.65 ms, SD = 72.02 for participants P08, P10, P17 respectively) compared to the mean time of all participants (M = 485.14 ms, SD = 192.48). This shows that participants' perception does not always align with their actual performance.

4.4.2 Perceived Task Performance in Urban Noise Condition. There was a lack of consensus in the remarks regarding the urban noise condition. More than half of the participants agreed that they were annoyed and distracted with the urban outdoor noise (13 out of 24). Several participants mentioned that both urban noise conditions were distracting. Five participants complained about the outdoor urban noise condition: "In urban outdoor noise I was stressed and alert. I had to correct the texts a lot under outdoor urban noise condition" (P01), "Urban outdoor noise was distracting" (P11, P14, P19), "I made several mistakes when finding icons in outdoor noise condition" (P19). One participant mentioned that they performed quicker in all of the tasks during the outdoor noise as they tried to "tune out the noise" (P06). Meanwhile other participants complained about urban indoor noise condition: "Ambient indoor noise was distracting, as it made feel like I was in a cafe and I felt as if I needed to be involved in a conversation with a friend" (P06), "Indoor noise put me almost asleep" (P13). Surprisingly, one participant claimed that they liked the urban outdoor noise condition and it did not affect their performance: "Traffic noise was good, I am used to it, I was quicker in tapping circles and typing tasks." (P13). These comments are mostly in line with our quantitative findings, as our results show that participants took less time in tapping circles and memorising an icon under both indoor and outdoor urban conditions. However, time per character entry was higher for these conditions.

4.4.3 Perceived Task Performance in Speech Condition. Regarding the speech condition, 13 out of 24 participants found the speech condition to be distracting and claimed their overall task performance was worse in both of the speech conditions. The participants emphasised that speech condition deteriorated their performance particularly in visual search and text entry tasks. Some of the insights from the participants are: *"English speech was distracting. When listening to it I made more mistakes while typing"* (P13, P14, P17, P22, P23), *"I made several mistakes when finding icons in English speech condition"* (P19). Regarding the target acquisition task only a few participants felt that speech affected their performance. For example, one participant said that they were slower in tapping circles under English speech condition but could not explain the reason behind (P01). However, another participant mentioned that they tapped on circles quicker to avoid "listening to the foreign language" (P06). When we examined their data, we found that P01 did in fact take less time (M = 574.42 ms, SD = 112.69) tapping circles in English speech condition compared to the mean time of all participants (M = 592.81 ms, SD = 119.60). The quantitative data of the participant P06 is in line with their comment, as they took less time tapping circles (M = 542.75 ms, SD = 117.19) compared to the total mean time taken to tap a circle (M = 609.38, SD = 132.66). Overall, the participants claimed that the speech condition affected their performance in visual search and text entry tasks.

# 4.5 Summary of Results

We summarise the effects of each condition on the different dependent variables and present them in Table 8. As can been seen from the table, the target acquisition time decreased in both music conditions, as well as in both urban noise conditions. Offset size was significantly larger in the music with slow tempo condition. Further, participants spent significantly less time on memorising an icon in the urban indoor noise condition. In addition,

participants produced significantly more errors in urban outdoor noise condition when searching for an icon. Finally, the time per character entry was significantly longer in the urban outdoor noise and meaningful speech conditions when compared to the silent condition.

Table 8. Table summarising the effect of ambient noise conditions on predicted variables compared to silent condition

	Mu	sic	Urba	n noise	Speech		
	Slow Tempo	Fast Tempo	Indoor	Outdoor	Meaningful	Meaningless	
Time to tap a circle	↓ *	↓ *	↓ *	$\downarrow$ *	-	-	
Offset size	↑*	ſ	-	-	-	-	
Time to memorise an icon	-	-	↓ *	$\downarrow$	-	-	
Time to find an icon	-	-	-	-			
Errors in icon search			Î	↑*			
Time per character entry			Î	↑*	↑*	ſ	
Errors in typing	-	-					

'↓' - decreased, '↑' - increased (relative to baseline) '\*' - the effect was statistically significant, '-' - no effect was observed Empty cells indicate a failed attempt to describe the variable with provided factors

# 4.6 Comparison of Cold- and Ambient Noise-Induced Situational Impairments

In this section we compare our findings to the effect of cold ambience on mobile interaction. We compare our mean values for time taken to tap a circle, offset size of the tap, time taken to memorise an icon, and the time taken to find an icon with the values presented in Sarsenbayeva *et al.* [45] and summarise them in Table 9. The comparison suggests that the effect of cold was more pronounced than all our ambient noise conditions in terms of the time taken to memorise an icon and time taken to find an icon. The effect of cold on offset size was similar for the fast tempo music, slow tempo music and meaningful speech conditions (approx. 43 pixels). In terms of time to tap a circle, the effect of cold was larger than the conditions of fast tempo music, slow tempo music, urban indoor noise, and meaningful speech conditions, except for the meaningless speech condition. The mean values for the silent condition are slightly smaller for offset size, time to memorise an icon, and time to find an icon compared to the warm condition. Meanwhile, the time to tap a circle is approximately equal for both of the baseline conditions.

# 82:18 • Z. Sarsenbayeva et al.

	Base	line	Situation				pairments		
Mean (SD)	Warm	Silent	Cold	Music Fast	Music Slow	Urban Indoor	Urban Outdoor	Speech (Meaning- ful)	Speech (Meaning- less)
Time to tap	593	591.84	603	577.76	574.15	583.16	573.56	592.81	609.38
a circle, ms	(137.89)	(125.07)	(144.62)	(135.85)	(123.45)	(117.38)	(110.39)	(119.60)	(132.66)
Offset size, px	41.34 (40.89)	39.38 (35.26)	42.66 (33.04)	42.95 (29.72)	42.11 (37.48)	41.06 (32.70)	40.91 (38.47)	42.90 (49.86)	39.47 (30.55)
Time to	815	748 76	854	737 71	745 15	719 11	743 56	753 20	738 00
memorise an icon, ms	(150.15)	(253.89)	(196.47)	(233.54)	(294.29)	(194.40)	(253.03)	(319.57)	(254.39)
Time to find an icon, ms	1632.24 (1235.27)	1587.74 (871.95)	1942.46 (2750.85)	1564.70 (830.81)	1753.19 (2263.27)	1543.12 (700.54)	1633.15 (1191.98)	1637.54 (1072.61	1520.99 (767.07)

Table 9. Comparison of the effect of situational impairments against the baseline

# 5 DISCUSSION

# 5.1 Effect of Ambient Noise on Smartphone Interaction

A large body of scientific work highlights the effect of ambient noise on human behaviour [22, 39], cognitive performance [5, 44, 54], and emotional state [25]. Our findings show that participants were quicker in the target acquisition task in both of the fast and slow music conditions. Our results are partly in line with the literature, as it has been shown that fast tempo music accelerates human performance in drinking [33], eating [9], or walking [34]. However, previous work has not identified a positive effect of slow tempo music on performance. In our qualitative data, participants mentioned that music in general helped them get into the rhythm of the task, which can explain why both music conditions led to a decrease in time taken to tap a circle. However, a faster completion of the task also resulted in a larger offset size. This agrees with previous work that demonstrated that while music increases the task performance speed, it also reduces overall accuracy [12]. As for the remaining tasks (visual search and text entry), we did not observe an effect of music on performance.

Another important factor to consider is that, even though literature has identified tempo as the main factor influencing human performance when listening to music [34], different types of music (*e.g.*, vocal music, rock) can also have a different effect on mobile interaction. For instance, Wen *et al.* demonstrated that performance in recall is significantly higher under classical music in contrast to rock music [55]. In our study, we manipulated the tempo of a classical music piece. While both slow and fast tempo classical music affected performance in a similar way, further research is needed to investigate the effect of a multitude of different music genres, a wider tempo range, and other musical elements such as pitch and timbre on mobile interaction.

Our results demonstrate an overall positive trend regarding the effect of urban noise on the target acquisition time and the time taken to memorise an icon. However, urban noise was negatively perceived by participants, with many participants commenting on its distracting nature. As such, there was an incentive for participants to perform quicker in these two tasks in order to reduce time spent in this unpleasant condition. However, in a real-world scenario a user would not be able to "escape" these unpleasant background noises by simply completing a task on their mobile device. Moreover, under the urban outdoor noise condition, participants made significantly more errors when finding an icon and took longer to type each character. These results are in line

with the findings presented in previous work [5, 22] and show the negative effect of urban noise as reported by participants in the interviews.

The effect of speech was limited to the text entry task. Our findings show that participants took significantly longer to type a text when listening to meaningful speech. These results correspond with previous work where meaningful speech was shown to have a negative effect on cognitive performance [4, 27, 50, 54]. Participants mentioned that they were listening to the English speech (meaningful), which resulted in longer completion times when typing a text. The effect of meaningless speech was much smaller. As participants did not understand the Kazakh speech (meaningless), it was likely easier for participants to ignore the spoken text.

It is important to note that factors outside the ambient noise conditions also affected participants' performance during the tasks. For example, in the target acquisition task, the Y centre coordinate of the circle played a significant role for predicting the time to tap a circle. Furthermore, the X centre coordinate of the circle significantly affected the offset size of the tap. The offset size of touch was smaller for the targets located closer to the right edge of the screen. These findings are in line with previous work which showed that screen coordinates have a significant effect on time to tap as well as tap accuracy [20, 45]. Another common factor influencing results was age, influencing performance for time taken to tap a circle, to memorise an icon, and to find an icon. This is in line with previous work which shows that age is an important factor affecting both memory and errors during time based tasks [42].

# 5.2 Contrasting Situational Impairments in Mobile Interaction

In addition to ambient noise, cold ambience has been identified as a situational impairment that can affect mobile interaction. By comparing these situational impairments to each other, we were able to establish a benchmark measurement and determine what strategy should be prioritised and in which situations. To allow for a fair comparison between previously reported situational impairments, we used the same target acquisition and visual search tasks as presented in Sarsenbayeva *et al.* [45]. We then compared the results of our participants against the results obtained by those authors, in which participants were exposed to cold as a source of situational impairment.

Our results show that the mean values for time taken to tap a circle, memorise an icon and find an icon were larger in a cold environment compared to the ambient noise conditions. Cold ambience affects people physiologically [7], with a decrease in body temperature when exposed to cold — albeit at different rates per individual. As a result, the cold affects the dexterity of the fingers and leads to deteriorated fine-motor performance. Our comparison shows that the effect of ambient noise is more nuanced. Some people are more accustomed to certain types of noise than others, and the effect of these noises can be in opposite directions between people. For instance, people with a certain preference in music might be less tolerant to other types of music. These preferential differences between people cause the effect of noise to be less homogeneous than situational impairments that affect everyone similarly (cold reduces our body temperature).

Previous work has highlighted the need for accumulated knowledge in the HCI/UbiComp community [30, 31]. Liu *et al.* [31] argue that accumulated knowledge contributes to the formation of important research themes in the field. In our work we have obtained new results that are directly comparable to a previous study, and therefore allow us to benchmark mobile interaction under different situational impairments.

#### 5.3 Detecting and Accounting for Noise-Induced Situational Impairments

Our findings demonstrate that mobile interaction can be affected by ambient noise in certain situations. Previous work has argued that detecting situational impairments is a fundamental step towards the successful adaptation of mobile interfaces [46]. The automatic detection of situational impairments during mobile interaction opens the way to appropriate interface adaptations; thus, enabling the interaction with mobile devices to be more

# 82:20 • Z. Sarsenbayeva et al.

appropriate to the user's context. Furthermore, solutions should ideally leverage the built-in sensors of the mobile device [47] as opposed to requiring additional instrumentation (*e.g.*, external temperature and humidity sensors to receive climatic parameters [1]). Following this suggestion, we argue that the built-in microphone of mobile devices can be used to detect noise-induced situational impairments, as previously suggested by Kanjo *et al.* [27]. In addition, a classifier could be used to distinguish between different noise types, as our results show that they can have different effects. Smartphone interfaces would then accommodate accordingly based on the condition detected. For example, if exposure to urban noise or speech is detected (shown to adversely affect text entry) while a user is typing, the phone can present an alternative interface to mitigate the effect of potential noise-induced situational impairments (*e.g.*, "WalkType" interface by Goel *et al.* [14], "Fat thumb" technique by Boring *et al.* [8]).

# 5.4 Limitations

Our study had several limitations. First, the study settings were strictly controlled. It is possible that in a naturalistic environment ambient noise types would be more diverse, and multiple noise types could occur simultaneously (*e.g.*, urban outdoor noise might contain music in a touristic part of the city). However, controlling for individual noise types was necessary as our goal was to systematically compare the effect of specific ambient noises on performance. Second, we were limited in the number of sounds included and did not include other types of music (*e.g.*, varying music genres, music with lyrics), urban noise (*e.g.*, markets, public performances), or speech (*e.g.*, second language of the participant). These are potential research directions that can be explored in future studies. We also did not run our study under different volume levels, but kept the noise level constant at between 55-60 dB to avoid participant discomfort.

Finally, we restricted our participants to use only their index finger to interact with the smartphone. We argue that this restriction was necessary to draw a fair comparison between the effect of ambient noise and cold ambience [45]. Moreover, by restricting the interaction technique we created a more comparable setting between participants.

# 6 CONCLUSION

In this study we investigate the effect of ambient noise on mobile interaction performance in target acquisition, visual search, and text entry tasks. We found that participants were significantly quicker in completing the target acquisition task in music conditions (both fast and slow) compared to the silent condition. However, they were significantly less accurate while listening to slow music. During the visual search task, participants took significantly less time to memorise an icon while listening to urban noise, but made more errors when finding an icon under urban outdoor noise. Participant performance during the text entry task was significantly affected by the urban outdoor noise and meaningful speech conditions. The comparison of cold-induced and noise-induced situational impairments on mobile interaction showed that the effect of cold ambience was more prominent on tasks requiring cognitive skills. Our findings enhance the understanding of noise-induced situational impairments research. Furthermore, detection of ambient noise and noise-induced situational impairments may be used for sensing user context and adapting the interface accordingly to mitigate their effect on mobile interaction.

# ACKNOWLEDGMENTS

This work is partially funded by the Academy of Finland (Grants 285459-iSCIENCE), the European Commission (Grant 6AIKA-A71143-AKAI), and Marie Skłodowska-Curie Actions (645706-GRAGE).

# REFERENCES

- S. Aram, A. Troiano, and E. Pasero. 2012. Environment sensing using smartphone. In 2012 IEEE Sensors Applications Symposium Proceedings. 1-4. https://doi.org/10.1109/SAS.2012.6166275
- [2] Amy FT Arnsten and Patricia S Goldman-Rakic. 1998. Noise stress impairs prefrontal cortical cognitive function in monkeys: evidence for a hyperdopaminergic mechanism. Archives of general psychiatry 55, 4 (1998), 362–368.
- [3] Alan Baddeley and Pierre Salame. 1986. The unattended speech effect: perception or memory? Journal of Experimental Psychology: Learning, Memory, and Cognition 12, 4 (1986), 525.
- [4] Simon Banbury and Dianne C Berry. 1997. Habituation and dishabituation to speech and office noise. Journal of Experimental Psychology: Applied 3, 3 (1997), 181.
- [5] Simon Banbury and Dianne C Berry. 1998. Disruption of office-related tasks by speech and office noise. British journal of psychology 89, 3 (1998), 499–517.
- [6] Simon P. Banbury, William J. Macken, Sébastien Tremblay, and Dylan M. Jones. 2001. Auditory Distraction and Short-Term Memory: Phenomena and Practical Implications. *Human Factors* 43, 1 (2001), 12–29. https://doi.org/10.1518/001872001775992462 arXiv:https://doi.org/10.1518/001872001775992462 PMID: 11474757.
- [7] ML Bartelink, A De Wit, H Wollersheim, A Theeuwes, and Th Thien. 1993. Skin vascular reactivity in healthy subjects: influence of hormonal status. Journal of Applied Physiology 74, 2 (1993), 727–732.
- [8] Sebastian Boring, David Ledo, Xiang 'Anthony' Chen, Nicolai Marquardt, Anthony Tang, and Saul Greenberg. 2012. The fat thumb: using the thumb's contact size for single-handed mobile interaction. In Proceedings of the 14th international conference on Human-computer interaction with mobile devices and services. ACM, 39–48.
- [9] Clare Caldwell and Sally A Hibbert. 2002. The influence of music tempo and musical preference on restaurant patrons' behavior. Psychology & Marketing 19, 11 (2002), 895–917.
- [10] Gianna Cassidy and Raymond AR MacDonald. 2007. The effect of background music and background noise on the task performance of introverts and extraverts. Psychology of Music 35, 3 (2007), 517–537.
- [11] Sheldon Cohen and Neil Weinstein. 1981. Nonauditory effects of noise on behavior and health. Journal of Social Issues 37, 1 (1981), 36–70.
- [12] Brian H Dalton and David G Behm. 2007. Effects of noise and music on human and task performance: A systematic review. Occupational Ergonomics 7, 3 (2007), 143–152.
- [13] Anind K. Dey. 2001. Understanding and Using Context. Personal Ubiquitous Comput. 5, 1 (Jan. 2001), 4–7. https://doi.org/10.1007/ s007790170019
- [14] Mayank Goel, Leah Findlater, and Jacob Wobbrock. 2012. WalkType: Using Accelerometer Data to Accomodate Situational Impairments in Mobile Touch Screen Text Entry. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '12). ACM, New York, NY, USA, 2687–2696. https://doi.org/10.1145/2207676.2208662
- [15] Charles J Golden and Shawna M Freshwater. 1978. Stroop color and word test. (1978).
- [16] Jorge Goncalves, Vassilis Kostakos, Simo Hosio, Evangelos Karapanos, and Olga Lyra. 2013. IncluCity: Using Contextual Cues to Raise Awareness on Environmental Accessibility. In Proceedings of the 15th International ACM SIGACCESS Conference on Computers and Accessibility (ASSETS '13). ACM, New York, NY, USA, Article 17, 8 pages. https://doi.org/10.1145/2513383.2517030
- [17] Jorge Goncalves, Zhanna Sarsenbayeva, Niels van Berkel, Chu Luo, Simo Hosio, Sirkka Risanen, Hannu Rintamäki, and Vassilis Kostakos. 2017. Tapping Task Performance on Smartphones in Cold Temperature. *Interacting with Computers* 29, 3 (2017), 355–367. https://doi.org/10.1093/iwc/iww029
- [18] Morgan Harvey and Matthew Pointon. 2018. Noisy Signals: Understanding the Impact of Auditory Distraction on Web Search Tasks. In Proceedings of the 2018 Conference on Human Information Interaction&Retrieval. ACM, 241–244.
- [19] Mack T Henderson, Anne Crews, and Joan Barlow. 1945. A study of the effect of music distraction on reading efficiency. *Journal of Applied Psychology* 29, 4 (1945), 313.
- [20] Niels Henze, Enrico Rukzio, and Susanne Boll. 2011. 100,000,000 Taps: Analysis and Improvement of Touch Performance in the Large. In Proceedings of the 13th International Conference on Human Computer Interaction with Mobile Devices and Services (MobileHCI '11). ACM, New York, NY, USA, 133–142. https://doi.org/10.1145/2037373.2037395
- [21] Eve Hoggan, Andrew Crossan, Stephen A Brewster, and Topi Kaaresoja. 2009. Audio or tactile feedback: which modality when?. In Proceedings of the SIGCHI conference on human factors in computing systems. ACM, 2253–2256.
- [22] Morris B Holbrook. 1981. Integrating compositional and decompositional analyses to represent the intervening role of perceptions in evaluative judgments. *Journal of Marketing Research* (1981), 13–28.
- [23] Sharon K Holmberg and Sharon Coon. 1999. Ambient sound levels in a state psychiatric hospital. Archives of psychiatric nursing 13, 3 (1999), 117–126.
- [24] Makoto Iwanaga and Takako Ito. 2002. Disturbance effect of music on processing of verbal and spatial memories. *Perceptual and motor skills* 94, 3\_suppl (2002), 1251–1258.

#### 82:22 • Z. Sarsenbayeva et al.

- [25] Kari Kallinen. 2002. Reading news from a pocket computer in a distracting environment: effects of the tempo of background music. Computers in Human Behavior 18, 5 (2002), 537 – 551. https://doi.org/10.1016/S0747-5632(02)00005-5
- [26] Juliane Kämpfe, Peter Sedlmeier, and Frank Renkewitz. 2011. The impact of background music on adult listeners: A meta-analysis. Psychology of Music 39, 4 (2011), 424–448.
- [27] Eiman Kanjo. 2010. Noisespy: A real-time mobile phone platform for urban noise monitoring and mapping. Mobile Networks and Applications 15, 4 (2010), 562–574.
- [28] J Peter Kincaid, Robert P Fishburne Jr, Richard L Rogers, and Brad S Chissom. 1975. Derivation of new readability formulas (automated readability index, fog count and flesch reading ease formula) for navy enlisted personnel. Technical Report. Naval Technical Training Command Millington TN Research Branch.
- [29] Hannu Kukka, Jorge Goncalves, Kai Wang, Tommi Puolamaa, Julien Louis, Mounib Mazouzi, and Leire Roa Barco. 2016. Utilizing Audio Cues to Raise Awareness and Entice Interaction on Public Displays. In Proceedings of the 2016 ACM Conference on Designing Interactive Systems (DIS '16). ACM, New York, NY, USA, 807–811. https://doi.org/10.1145/2901790.2901856
- [30] Yong Liu, Jorge Goncalves, Denzil Ferreira, Simo Hosio, and Vassilis Kostakos. 2014. Identity Crisis of Ubicomp?: Mapping 15 Years of the Field's Development and Paradigm Change. In Proceedings of the 2014 ACM International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp '14). ACM, New York, NY, USA, 75–86. https://doi.org/10.1145/2632048.2632086
- [31] Yong Liu, Jorge Goncalves, Denzil Ferreira, Bei Xiao, Simo Hosio, and Vassilis Kostakos. 2014. CHI 1994-2013: Mapping Two Decades of Intellectual Progress Through Co-word Analysis. In Proceedings of the 32Nd Annual ACM Conference on Human Factors in Computing Systems (CHI '14). ACM, New York, NY, USA, 3553–3562. https://doi.org/10.1145/2556288.2556969
- [32] Randi C Martin, Michael S Wogalter, and Janice G Forlano. 1988. Reading comprehension in the presence of unattended speech and music. Journal of memory and language 27, 4 (1988), 382–398.
- [33] Heather McElrea and Lionel Standing. 1992. Fast music causes fast drinking. Perceptual and Motor skills 75, 2 (1992), 362-362.
- [34] Ronald E Milliman. 1982. Using background music to affect the behavior of supermarket shoppers. *The journal of Marketing* (1982), 86–91.
- [35] Ronald E Milliman. 1986. The influence of background music on the behavior of restaurant patrons. Journal of consumer research 13, 2 (1986), 286–289.
- [36] Sachi Mizobuchi, Mark Chignell, and David Newton. 2005. Mobile Text Entry: Relationship Between Walking Speed and Text Input Task Difficulty. In Proceedings of the 7th International Conference on Human Computer Interaction with Mobile Devices & Amp; Services (MobileHCl '05). ACM, New York, NY, USA, 122–128. https://doi.org/10.1145/1085777.1085798
- [37] Alexander Ng, Stephen A. Brewster, and John H. Williamson. 2014. Investigating the Effects of Encumbrance on One- and Two- Handed Interactions with Mobile Devices. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '14). ACM, New York, NY, USA, 1981–1990. https://doi.org/10.1145/2556288.2557312
- [38] Alexander Ng, John H. Williamson, and Stephen A. Brewster. 2014. Comparing Evaluation Methods for Encumbrance and Walking on Interaction with Touchscreen Mobile Devices. In Proceedings of the 16th International Conference on Human-computer Interaction with Mobile Devices & Services (MobileHCI '14). ACM, New York, NY, USA, 23–32. https://doi.org/10.1145/2628363.2628382
- [39] Adrian C North, David J Hargreaves, and Sarah J Heath. 1998. Musical tempo and time perception in a gymnasium. Psychology of Music 26, 1 (1998), 78-88.
- [40] Theodore J. Noseworthy and Karen Finlay. 2009. A Comparison of Ambient Casino Sound and Music: Effects on Dissociation and on Perceptions of Elapsed Time While Playing Slot Machines. *Journal of Gambling Studies* 25, 3 (01 Sep 2009), 331–342. https: //doi.org/10.1007/s10899-009-9136-x
- [41] Michael D Oliver. 1996. The effect of background music on mood and reading comprehension performance of at-risk college freshmen. Ph.D. Dissertation. Wayne State University.
- [42] Denise C Park, Christopher Hertzog, Daniel P Kidder, Roger W Morrell, and Christopher B Mayhorn. 1997. Effect of age on event-based and time-based prospective memory. *Psychology and aging* 12, 2 (1997), 314.
- [43] Pierre Salame and Alan Baddeley. 1982. Disruption of short-term memory by unattended speech: Implications for the structure of working memory. Journal of Verbal Learning and Verbal Behavior 21, 2 (1982), 150 – 164. https://doi.org/10.1016/S0022-5371(82)90521-7
- [44] Pierre Salame and Alan Baddeley. 1989. Effects of background music on phonological short-term memory. The Quarterly Journal of Experimental Psychology 41, 1 (1989), 107–122.
- [45] Zhanna Sarsenbayeva, Jorge Goncalves, Juan García, Simon Klakegg, Sirkka Rissanen, Hannu Rintamäki, Jari Hannu, and Vassilis Kostakos. 2016. Situational Impairments to Mobile Interaction in Cold Environments. In Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp '16). ACM, New York, NY, USA, 85–96. https://doi.org/10.1145/2971648.2971734
- [46] Zhanna Sarsenbayeva, Niels van Berkel, Chu Luo, Vassilis Kostakos, and Jorge Goncalves. 2017. Challenges of Situational Impairments During Interaction with Mobile Devices. In Proceedings of the 29th Australian Conference on Computer-Human Interaction (OZCHI '17). ACM, New York, NY, USA, 477–481. https://doi.org/10.1145/3152771.3156161
- [47] Zhanna Sarsenbayeva, Niels van Berkel, Aku Visuri, Sirkka Rissanen, Hannu Rintamäki, Vassilis Kostakos, and Jorge Goncalves. 2017. Sensing Cold-Induced Situational Impairments in Mobile Interaction Using Battery Temperature. Proc. ACM Interact. Mob. Wearable

Ubiquitous Technol. 1, 3, Article 98 (Sept. 2017), 9 pages. https://doi.org/10.1145/3130963

- [48] Bastian Schildbach and Enrico Rukzio. 2010. Investigating Selection and Reading Performance on a Mobile Phone While Walking. In Proceedings of the 12th International Conference on Human Computer Interaction with Mobile Devices and Services (MobileHCI '10). ACM, New York, NY, USA, 93–102. https://doi.org/10.1145/1851600.1851619
- [49] Andrew Sears, Min Lin, Julie Jacko, and Yan Xiao. 2003. When computers fade: Pervasive computing and situationally-induced impairments and disabilities. In HCI International, Vol. 2. 1298–1302.
- [50] Stephen Stansfeld, John Gallacher, Wolfgang Babisch, and Martin Shipley. 1996. Road traffic noise and psychiatric disorder: prospective findings from the Caerphilly Study. *Bmj* 313, 7052 (1996), 266–267.
- [51] Sébastien Tremblay, Alastair P Nicholls, David Alford, and Dylan M Jones. 2000. The irrelevant sound effect: Does speech play a special role? Journal of Experimental Psychology: Learning, Memory, and Cognition 26, 6 (2000), 1750.
- [52] Kristin Vadas, Nirmal Patel, Kent Lyons, Thad Starner, and Julie Jacko. 2006. Reading On-the-go: A Comparison of Audio and Hand-held Displays. In Proceedings of the 8th Conference on Human-computer Interaction with Mobile Devices and Services (MobileHCI '06). ACM, New York, NY, USA, 219–226. https://doi.org/10.1145/1152215.1152262
- [53] Niels van Berkel, Chu Luo, Theodoros Anagnostopoulos, Denzil Ferreira, Jorge Goncalves, Simo Hosio, and Vassilis Kostakos. 2016. A Systematic Assessment of Smartphone Usage Gaps. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems (CHI '16). ACM, New York, NY, USA, 4711–4721. https://doi.org/10.1145/2858036.2858348
- [54] N Venetjoki, A Kaarlela-Tuomaala, E Keskinen, and V Hongisto. 2006. The effect of speech and speech intelligibility on task performance. *Ergonomics* 49, 11 (2006), 1068–1091.
- [55] We Wen and Kanachi Michihiko. 2006. The effects of music type and volume on short-term memory. *Tohoku Psychol Folia* 64 (2006), 68–76.

Received February 2018; revised April 2018; accepted April 2018

# **Chapter 5**

# Quantifying the Effects of Stress on Mobile Interaction

In this chapter we quantify the effect of an internal contextual factor – stress – on mobile interaction. Stress is defined as a mental state often experienced on a regular basis that has been shown to negatively influence people's everyday activities [104, 126, 205]. As the nature of some stressors is temporal [145, 251] and caused by a specific situation (*e.g.*, work related, family related) [128], in this work we study the effects of stress on mobile interaction from the perspective of SIIDs. Despite the acknowledged negative effects of stress on human cognitive performance (*e.g.*, impaired working [146,281] and declarative memory [122,178]), its' effect on mobile interaction has yet to be explored.

We induce stress on our participants using the *Trier Social Stress Test* [121] – a protocol commonly used in Psychology research to induce stress. Stress was induced in two steps: first, the participants delivered a five-minute speech explaining why they are the best candidate for their dream job. Then the participants had to perform consecutive mathematical subtraction. Both of the tasks were performed in front of the panel of judges composed by our research confederates. We asked our participants to complete smartphone tasks upon their arrival (baseline measurements) and under the induced stress. We validated the occurrence of stress using HRV data, derived from physiological data collected with an Empatica E4 [57] wearable sensor. In addition, we also used State-Trait Anxiety Inventory [250] self-report questionnaire to validate the presence of stress in participants.

Our findings show that stress significantly affected participants' performance in the target acquisition and visual search tasks; however, our results did not provide any evidence on the effect of stress on the text entry task. Precisely, our participants took significantly less time to complete the tap on the targets when they were stressed.

Moreover, their tapping accuracy was significantly lower when they were stressed as compared to the baseline condition. Regarding, the visual search task, our results indicate that participants were significantly faster to memorise icons when they were stressed. These findings are in line with prior research showing that people rush to complete the tasks to "escape" the anxiety and discomfort caused by stress [163].

The results of this study enhance our understanding of the effects of stress on mobile interaction and contribute to accumulation of knowledge on the SIIDs research. The details of our approach can be found in the attached publication in Section 5.1.

# 5.1 Publication

©Copyright held by the authors. Publication rights licensed to ACM. This is the authors' version of the work. It is posted here for your personal use. Not for redistribution. The definitive Version of Record was published in:

Zhanna Sarsenbayeva, Niels van Berkel, Danula Hettiachchi, Weiwei Jiang, Tilman Dingler, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. 2019. Measuring the Effects of Stress on Mobile Interaction. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 3, 1, Article 24 (March 2019), 18 pages. DOI: https: //doi.org/10.1145/3314411.

Ethics ID: 1851997.1, The University of Melbourne Human Ethics Advisory Group.

# Measuring the Effects of Stress on Mobile Interaction

ZHANNA SARSENBAYEVA, The University of Melbourne, Australia NIELS VAN BERKEL, The University of Melbourne, Australia DANULA HETTIACHCHI, The University of Melbourne, Australia WEIWEI JIANG, The University of Tokyo, Japan TILMAN DINGLER, The University of Melbourne, Australia EDUARDO VELLOSO, The University of Melbourne, Australia VASSILIS KOSTAKOS, The University of Melbourne, Australia JORGE GONCALVES, The University of Melbourne, Australia

Research shows that environmental factors such as ambient noise and cold ambience can render users situationally impaired, adversely affecting interaction with mobile devices. However, an internal factor which is known to negatively impact cognitive abilities – stress – has not been systematically investigated in terms of its impact on mobile interaction. In this paper, we report a study where we use the Trier Social Stress Test to induce stress on participants, and investigate its effect on three aspects of mobile interaction: target acquisition, visual search, and text entry. We find that stress reduces completion time and accuracy during target acquisition tasks, as well as completion time during visual search tasks. Finally, we are able to directly contrast the magnitude of these effects to previously published effects of environmentally-caused impairments. Our work contributes to the growing body of literature on situational impairments.

 $\label{eq:CCS} Concepts: \bullet \textbf{Human-centered computing} \rightarrow \mbox{Empirical studies in HCI;} \bullet \textbf{Human-centered computing} \rightarrow \mbox{Ubiquitous and mobile computing; Smartphones.}$ 

Additional Key Words and Phrases: Smartphones, situational impairments, mobile interaction, stress, performance, Trier Social Stress Test

# **ACM Reference Format:**

Zhanna Sarsenbayeva, Niels van Berkel, Danula Hettiachchi, Weiwei Jiang, Tilman Dingler, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. 2019. Measuring the Effects of Stress on Mobile Interaction. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 3, 1, Article 24 (March 2019), 18 pages. https://doi.org/10.1145/3314411

Authors' addresses: Zhanna Sarsenbayeva, The University of Melbourne, Melbourne, Australia, z.sarsenbayeva@student.unimelb.edu.au; Niels van Berkel, The University of Melbourne, Melbourne, Australia, n.vanberkel@student.unimelb.edu.au; Danula Hettiachchi, The University of Melbourne, Melbourne, Australia, dhettiachchi@student.unimelb.edu.au; Weiwei Jiang, The University of Tokyo, Tokyo, Japan, wjiang@akg.t.u-tokyo.ac.jp; Tilman Dingler, The University of Melbourne, Melbourne, Australia, tilman.dingler@unimelb.edu.au; Eduardo Velloso, The University of Melbourne, Melbourne, Australia, eduardo.velloso@unimelb.edu.au; Vassilis Kostakos, The University of Melbourne, Melbourne, Australia, vassilis.kostakos@unimelb.edu.au; Jorge Goncalves, The University of Melbourne, Australia, jorge.goncalves@unimelb.edu.au.

© 2019 Copyright held by the owner/author(s). Publication rights licensed to ACM. 2474-9567/2019/3-ART24 \$15.00 https://doi.org/10.1145/3314411

Permission to make digital or hard copies of all or part of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than the author(s) must be honored. Abstracting with credit is permitted. To copy otherwise, or republish, to post on servers or to redistribute to lists, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

# 24:2 • Z. Sarsenbayeva et al.

# 1 INTRODUCTION

Due to the ubiquitous nature of smartphones, users interact with them under varying environmental (*e.g.*, in noisy environments) and internal (*e.g.*, under stress, in different moods) conditions. Previous research has shown that environmental and contextual factors, such as cold ambience [12], ambient noise [44], encumbrance [38], and walking [11] negatively affect mobile interaction and lead to *situational impairments*. This has given rise to a growing body of work in HCI that aims to enable smartphones to detect when the user is situationally impaired and subsequently adapt the interface to mitigate the effects of such impairments during mobile interaction [43, 46, 57].

Although prior research has emphasised the importance of understanding the effects of situational impairments [43, 49, 63], these studies mostly focus on environmental factors, such as cold ambience. In contrast, relatively little work has investigated the effect of an important internal factor on mobile interaction: *stress*, a mental state often experienced on a daily basis [1], which has been shown to negatively impact performance in everyday activities (*e.g.*, [16, 22, 39]).

In this paper, we investigate the effect of stress on task performance in smartphones. According to Lazarus *et al.* [23], stress can be defined in terms of the relationship between an individual and an environment or situation [51]. Given this notion [23] and the temporal effect of certain stressors (personal or work-related) on human behaviour [26, 55], stress has been identified as a potential cause of situational impairment that is likely to have an impact during mobile interaction [43].

In our study, we use the *Trier Social Stress Test* (TSST) [20] to induce stress in our participants in order to measure its effect on three mobile interaction tasks: target acquisition, visual search, and text entry [44]. We show that stress significantly reduces target access time and accuracy during target acquisition tasks compared to the baseline, as well as completion time during visual search tasks. Further, we directly contrast the magnitude of stress-induced situational impairments to other situational impairments based on results in the literature, namely cold ambience [42] and ambient noise [44].

The main contributions of this paper are to quantify the effect of stress on mobile task performance, compare its effect size to that of cold- and noise-induced situational impairments, and, hence, to contribute to the growing body of research on situational impairments.

# 2 RELATED WORK

Previous work has highlighted that different environmental and contextual factors, such as cold ambience [12], ambient noise [44], mobile state of the user [11], and encumbrance [38] can negatively affect mobile interaction. These factors can cause users to experience situational impairments and, thereby, impact their performance during mobile interaction [43]. For example, cold ambient temperature has been shown to adversely affect smartphone input performance [12, 42]. In particular, throughput and accuracy drop significantly when performing acquisition tasks under cold temperatures. Other studies have shown a negative effect of walking on text entry [11, 35] and target acquisition tasks [48]. User encumbrance also negatively affects mobile interaction, resulting in decreased accuracy, longer target acquisition times, and increased error rate [38].

Sarsenbayeva *et al.* [44] measured the effect of background noise on smartphone interaction performance. The types of background noise presented in that study included music with fast and slow tempo, urban indoor and outdoor noise, and meaningful and meaningless speech. Similar to our study, mobile input performance was measured in terms of three common smartphone activities: target acquisition, visual search, and text entry. Performance under ambient noise conditions was benchmarked against the silent condition. The authors found that music reduced completion time during target acquisition tasks, while urban noise and speech increased the text entry rate. Although a number of previous studies have investigated the effect of stress on interaction with stationary technology (*e.g.*, keyboard and mouse), the effect of stress on mobile interaction remains under-explored. For example, Karunaratne *et al.* [18] reported stronger keyboard taps amongst participants exposed to

stressful conditions. Furthermore, Rodrigues *et al.* [40] investigated the effect of stress on mouse movements. The authors found that stressed students, when presented with challenging questions, carried out significantly more mouse movements [40].

In our study, we contribute to the growing body of research on situational impairments [43] by quantifying the effect of stress on interaction with smartphones, an internal factor that has been shown to adversely impact performance when using other technologies, such as desktop computers [18, 40, 62]. We achieve this by using the TSST protocol [20] to induce stress in our participants, and measure changes to their performance while completing typical smartphone tasks.

# 2.1 Cognitive and Physiological Effects of Stress

Behavioural and psychological scientists study the effect of stress on humans' daily lives for a variety of reasons. For example, researchers have aimed to identify the events leading to stress and how negative outcomes of cumulative stress can be avoided [14].

Early work, showed a curvilinear relationship between arousal and task performance, particularly with very high and very low levels of arousal inhibiting task performance, whereas moderate levels facilitate it [65]. More recently, research has shown the negative effect of stress on human cognitive performance [67]. For example, stress negatively affects working memory [27, 67] and verbal declarative memory [21, 37]. In addition, studies by Payne *et al.* [39] and Jelici *et al.* [16] report adverse effects of stress on human cognition resulting in memory impairments. Lupien *et al.* [26] also show that under stressful conditions, declarative memory is significantly impaired compared to a non-stressful condition in a word-pairs recall task. Moreover, Kuhlmann *et al.* [22] show that acute stress significantly delays memory retrieval. They utilise the TSST, a widely used protocol for inducing stress in participants through two tasks: public speaking and arithmetic subtraction [20]. They then asked their participants to recall pairs of words memorised at the beginning of the experiment. The results of this study show that stressed participants have significantly reduced memory retrieval as compared to the control condition [22]. Wolkowitz *et al.* [64] report similar results showing that stress has a detrimental effect when recalling words from a previously learnt list.

Further, Marquart [29] observed that stress decreased participants' learning ability and results in non-adaptive behaviour (*e.g.*, longer reaction time when completing cognitive tasks [23]). Verville *et al.* [61] found that participants who were exposed to a stressful condition required significantly longer time for recognising pictures that were flashed on a screen as compared to a control group. This, however, is a contested finding as other studies have shown that stress sped up participants' performance during an image sorting task (*e.g.*, [33]). In addition, a reduction of task completion time resulted in a greater number of errors [33]. These results are similar to a study by McKinney [32], where stress increased the number of errors in completing multiplication problems and a learning syllables task.

Based on this rich literature we hypothesise that stress will hinder user performance in smartphone tasks. In particular, participants under stress will be less accurate when compared to a baseline measurement.

#### 2.2 Stress Detection Methods

Stress induces biological responses, which can thus be physiologically measured. Examples include changes in skin conductivity [50], heart rate variability [56], muscle tension [56], and heart rate [47]. Several studies have focused on the development of wearable technology that allows for the unobtrusive tracking of physiological measures. As a result, a variety of wearables exists that track physiological measures (*e.g.*, Empatica E4 wristband<sup>1</sup>, Oura ring<sup>2</sup>).

<sup>&</sup>lt;sup>1</sup>https://www.empatica.com/research/e4/

<sup>&</sup>lt;sup>2</sup>https://ouraring.com/

#### 24:4 • Z. Sarsenbayeva et al.

Several studies have attempted to detect stress from people's interaction behaviour using sensor-based technologies [3, 14, 41]. For example, Hernandez *et al.* use a pressure-sensitive keyboard and a capacitive mouse to detect the effect of stress during computer interaction. They found that more than half of their participants increased their typing pressure under the stressful condition when compared to the control condition. They also showed that the majority of participants (75%) covered more of the surface of the mouse when they were stressed [14]. Furthermore, Exposito *et al.* [10] suggest using iPhone's built-in keyboard pressure sensor to detect stress. The authors show that under stressed conditions, participants typing pressure was higher when compared to non-stressed conditions [10]. Prior research has also shown that the combination of wearable sensors and smartphone interaction patterns can detect stress more accurately. For instance, Sano and Picard [41] achieved a higher accuracy rate for stress detection (75%) compared to previous results (53%) that only used smartphone interaction patterns [3]. Other stress detection methods are based on self-reports and questionnaires. For example, Cohen *et al.* [9] suggest the use of perceived stress as an objective stress measure. Further, Spielberger and colleagues recommend asking participants to self-report their anxiety levels using the State-Trait Anxiety Inventory (STAI) to identify the occurrence of stress [54].

In our study, we collect both sensor and self-report data to validate that we are indeed inducing stress in our participants throughout the experiment. Namely, we use heart rate variability (HRV) derived from the data of an Empatica E4 and collect self-report data using the STAI questionnaire.

# 3 METHOD

In this study, we investigate the effect of stress on performance on three common mobile interaction tasks: target acquisition, visual search, and text entry. We use the experimental tasks developed in the study by Sarsenbayeva *et al.* [44]. This allows us to directly compare the effects of stress-induced situational impairments with those previously reported for cold-induced and noise-induced situational impairments. We describe each task in detail below.

# 3.1 Smartphone Tasks

We used a Samsung Galaxy S7 smartphone (Android 7.0) with a 5.1-inch screen (1080×1920px) in this experiment. This smartphone model was chosen in order to have an identical screen size and screen resolution to the smartphone used in previous studies by Sarsenbayeva *et al.* [42, 44] on situational impairments. The three tasks were presented to participants in a random order to avoid sequence effects. Participants first completed extensive training until they were comfortable with each of the presented tasks, thus minimising any potential learning effects. Both the training and actual tasks were completed while standing, and participants were instructed to use only the index finger of their dominant hand for interaction, while holding the smartphone with their non-dominant hand.

*3.1.1 Target Acquisition Task.* In the target acquisition task participants are asked to tap circular targets (radius = 135 px) which appear one at a time at random locations on the screen. Each circle has an indicated centre and participants are asked to tap the indicated center of the targets as precisely and quickly as possible. We log the coordinates of the target's centre, the participant's touch point, and the elapsed time. The interface of the task is shown in Figure 1-A.

3.1.2 Visual Search Task. In the visual search task, participants must find a target icon amongst 24 other icons  $(100 \times 100 \text{ px})$ , arranged according to a 4 × 6 grid [13]. The target icon is first shown, participants can then take as much time as needed to memorise it (Figure 1-B). In the subsequent screen they must find and tap the memorised target among 24 icons (Figure 1-C). To minimise any potential learning effects, the application ensures that the target is randomly selected and remaining icons are randomly distributed across the grid.

# Measuring the Effects of Stress on Mobile Interaction • 24:5

** .at 65	2% # 926 pm B 41 ut 62% # 925 pm Once you remember the icon, press Continue	n Contraction	NE .al	62% 🖹 9:25 pm	Th Da fly in the	¥F "af 62% ∰ 9 car with us.	24 pm	Yee y worst old T in my verse ever live y	K at 6 ime despite thy wrong oung.	2% <b>≡</b> 9:25 pm My love shall
		Viber ChatON	Skype	Drive	There is a fly ir	n the car with us.		Yet do thy worst My love shall in	old Time despite my vers3 ever live	thy wrong young.
	Drive	YouTube	Adobe Acr	Gmail		SUBMIT			SUBMIT	
		Candy Cru	er Angry Birds	Facebook	1 2 3	4 5 6 7 8 9	0	1 2 3 4	5 6 7 8	9 0
	CONTINUE	Fruit Ninja	G+ Google+	Chrome	q w e a s d	rtyuio fghjk	р I	q w e r a s d	tyui fghj	o p k l
_		Dropbox Maps	Instagram	WhatsApp	<ul> <li></li></ul>	cvbnm English(US)	() ()	<ul> <li>☆ z x</li> <li>!@# ĝi</li> </ul>	cvbn English(US)	m ⊗ . ₽

Fig. 1. Interface of the application with Target Acquisition Task (A), Visual Search Task (B-C), and Text Entry Task with user's input for easy and difficult texts (D-E).

*3.1.3 Text Entry Task.* In the text entry task, our participants are presented with a snippet of text (displayed on the top-half of the screen), which they must re-type in a text box. The texts are of varying difficulty: 1) easy – consisting of only one sentence with common words widely used on a daily basis (see Figure 1-D), 2) difficult – consisting of several sentences chosen from Shakespeare's sonnets (see Figure 1-E). In total we have 10 easy sentences and 10 difficult sentences, and at each round of tasks a sentence is randomly selected by the application. We validated the difficulty of the sentences with the Flesch-Kincaid readability test [19]. Easy sentences had an average grade of 1.6 and difficult sentences had an average grade of 5.1.

# 3.2 Participants

We recruited 24 participants between 20 and 55 (M = 31, SD = 9.5) years old through our university's mailing lists and snowball recruitment with equal number of male and female participants to avoid gender bias [59]. Participants had a diverse range of educational backgrounds (*e.g.*, Electrical, Mechanical, Infrastructure and Civil Engineering, Agriculture, Entomology, Architecture, and Computer Science).

# 3.3 Procedure

To induce stress in our participants, we followed the TSST protocol [20]. TSST is a validated protocol used to induce stress in study participants that has been widely used in Psychology research. The original TSST protocol [20] describes three main stress points in the experiment: following a relaxation period (20 minutes), following stress-inducing tasks (speech and arithmetic), and following post-stress recovery. Hence, we ask participants to complete the smartphone tasks at these three points. The overview of the experimental procedure is presented in Figure 2.

In order to validate whether our participants experienced stress throughout the experiment, we calculated their heart rate variability (HRV) from the Empatica E4 wristband data, as HRV has been shown to correlate with a person's stress levels [56]. We also collected self-reported anxiety levels of the participants via the STAI questionnaire [54], a tool commonly used by researchers conducting studies with the TSST to validate fluctuations in stress levels. Sensor data was then cross-referenced with smartphone data using timestamps.

In line with the TSST protocol, participants were not aware that the study investigated the effect of stress on mobile interaction, and were not informed that we would induce stress on them to avoid triggering anxiety-related

# 24:6 • Z. Sarsenbayeva et al.



Fig. 2. Experimental Procedure.

behaviours. Again, according to the TSST protocol, this is done to reduce participant bias and to let us observe the natural reaction of participants to stress. The experimental design was approved by the Ethics Committee of our university. The experiment duration was approximately 90 minutes per participant, including briefing, training, data collection, and final interview. Each participant received a \$30 gift voucher for participation.

*3.3.1 Participant Briefing.* The experiment took place at our institution's usability lab, with an experimental setup consisting of two adjacent rooms, as recommended by the TSST protocol. One room works as a desensitisation room, where participants can rest and relax, while the other room is used for the stress-inducing tasks. Upon arrival, we welcomed our participants to the desensitisation room. We then informed them that we are investigating their cognitive performance as measured by their presentation skills and arithmetic skills, as well as their performance in a range of smartphone tasks. We then asked participants to sign a consent form agreeing to participate in the study and collected their demographic data (age, gender, background, dominant hand). Next, we instrumented our participants with an Empatica E4 wristband, and instructed them to wear it throughout the experiment on their non-dominant hand. Following this setup we asked our participants to complete all smartphone tasks for training purposes. Participants trained with all three study tasks in random order until they were comfortable with each one.

*3.3.2 Baseline Measurements.* After training, participants were left alone for 20 minutes, so that they could rest and stabilise. During this period we provided them with reading material with neutral content to keep them occupied and not cause any anxiety (descriptions and photographs of plants growing in different regions of the world), as recommended in the TSST protocol [20]. We also asked them not to use their smartphone during this time. Following, the participants were asked to complete a round of tasks on a smartphone. The purpose of this stage was to create a baseline measurement of performance while not stressed. After finishing the tasks, the participants completed the STAI questionnaire [54] to measure their perceived anxiety state.

*3.3.3* Inducing Stress — Step 1. We guided our participants to the experiment room. Here, we provided them with a scenario according to which they were applying for their dream job and needed to attend a hiring interview. Participants were given 5 minutes to prepare a speech to convince a panel as to why they were the best candidate for their dream job. We also informed them that their performance was going to be video-recorded and reviewed by judges trained in public speaking. Once the preparation time was over, the participants were asked to deliver their speech in front of the panel of three judges, who were, unbeknownst to the participants, confederates of the research team. Participants had to speak for the entire 5 minutes, during which the panel members did not give any feedback and maintained neutral facial expressions.

3.3.4 Inducing Stress — Step 2. Upon completing step 1, participants were asked to perform an arithmetic task. We asked them to subsequently decrement 1022 by 13 and to say the number sequence out loud. The task is inspired by the 'Serial Sevens' tasks [52], which is used in clinical tests to assess mental functions, and lasts for 5 minutes. In case of miscalculation, a researcher informed the participants of their error and they were asked to restart the task from 1022. The purpose of the arithmetic task was to induce further stress on participants.

After completing the arithmetic task, the participants completed a round of tasks on a smartphone under stress, followed by the STAI questionnaire (Round 1).

*3.3.5 Post-Stress Recovery.* At this stage, we brought participants back to the desensitisation room and left them alone for 10 minutes to rest and recover. We then asked them to complete a final round of smartphone tasks and complete the STAI questionnaire (Round 2). According to the TSST protocol [20], cortisol level reaches its peak during this moment of the experiment and, therefore, we wanted to assess their performance during mobile interaction at this point.

*3.3.6 Participant Debriefing.* Finally, we debriefed participants about the real purpose of the study and explained that it was expected to experience stress during the experiment. Further, we clarified that the tasks performed during the experiment were unreasonably difficult and did not reflect upon their aptitude or ability. We also informed them that we did not video-record their performance. We then conducted a short, semi-structured interview with our participants, asking them to report on their subjective perception of the experiment and the effect of stress on their performance.

# 4 RESULTS

In this section, we report our results regarding the effect of stress on our participants' performance across target acquisition, visual search, and text entry tasks. We also report participants' subjective assessments of their performance in the aforementioned smartphone tasks.

# 4.1 Validation of Stress Occurrence

We used HRV to validate the presence of stress in our participants and followed the process, described by McDuff and colleagues [31]. To generate HRV data from the bio-signals collected with the Empatica E4 sensor, we performed the generic HRV analysis method as follows. First, we interpolated and re-sampled inter-beat interval (IBI) data at 4Hz, to align the signals in a uniform time interval. Then we obtained an HRV power spectrum in time series from the detrended interpolated IBI data, by applying the Fast Fourier Transform (FFT) algorithm with a 512-sized slide window (i.e., a 512-sample or 128-second segment of a signal). We calculated the HF (High-Frequency) powers of the HRV as the summation of the discrete points corresponding to the power spectrum under 0.15 - 0.40Hz. As HF HRV was previously shown to be a reliable indicator of stress occurrence [4], we used it for further validation of presence of stress in our participants. We acknowledge that HRV can be a result of multiple factors apart from stress; however, we argue that in combination with the established stress-inducing protocol and participants' self-reports, we were able to reliably validate the occurrence of stress during our experiment. In addition, we did not collect EDA data due to the following reasons. First, EDA data collected from the wrist is shown to describe thermoregulatory relevant electrodermal phenomena, rather than the psychophysiological nature of the response [8]. Second, EDA is better suited for validating stress caused by discrete events (e.g., electrical shock) [7, 8, 51], whereas in our study participants experienced longitudinal cumulative stress.

We applied a one-way repeated measures ANOVA, which yielded a statistically significant effect of stress on heart rate variability (F(2, 24) = 6.54, p < 0.01). A Tukey HSD post-hoc comparison test (with Bonferroni corrections) reveals that HRV values during Stress Induction (M = 32.84, SD = 20.89) and Post-stress Recovery periods (M = 37.24, SD = 18.69) are significantly lower (p < 0.01) compared to HRV values during the baseline measurements (M = 40.81, SD = 26.05). Since low values of HF HRV indicate a physiological presence of stress [15], we can conclude from the HRV results that the participants were stressed after undergoing both the speech and arithmetic subtraction tasks of the protocol. The mean values for HF HRV are visualised in Figure 3.

#### 24:8 • Z. Sarsenbayeva et al.



Fig. 3. HRV (HF) Mean Values (95% Confidence Interval (CI))

Next, we examined participants' self-reported anxiety values as measured by the STAI questionnaire as collected following completion of each round of smartphone tasks. A one-way repeated measures ANOVA yielded a statistically significant effect of stress on self-reported anxiety values (F(2, 24) = 10.23, p < 0.01). Post-hoc comparisons using the Tukey HSD test (with Bonferroni corrections) indicated that participants were significantly more anxious (p < 0.01) during Stress Induction period (M = 37.08, SD = 11.77) of the experiment compared to the baseline values (M = 30.71, SD = 6.37). Furthermore, participants reported feeling significantly less anxious (p < 0.01) during Post-stress Recovery period (M = 31.79, SD = 8.64) when compared to Stress Induction period. The mean values for self-reported anxiety levels are presented in Figure 4.



Fig. 4. Self-Reported Anxiety Mean Values (95% CI)

From the sensor data and self-reported anxiety values we can conclude that the study protocol performed as intended in terms of inducing stress in participants.

# 4.2 Target Acquisition Task

We measured the target acquisition performance in terms of target acquisition time (ms), offset (px) between the target centre and the touch point, and effective throughput<sup>3</sup>. First, we applied a one-way repeated measures ANOVA to investigate the effect of stress on target acquisition time. The result showed a statistically significant

 $<sup>^{3}</sup>$ A measure of human performance in completing target selection tasks that describes the relationship between the difficulty of the task and target acquisition time [28]

effect of stress on time taken to hit a target (F(2, 24) = 17.05, p < 0.01). Post-hoc comparisons using the Tukey HSD test (with Bonferroni corrections) indicate that participants take significantly less time to tap a circle in Stress Induction (M = 534.77, SD = 124.43, p < 0.01) and Post-stress recovery periods (M = 520.18, SD = 126.52, p < 0.01) as compared to the baseline (M = 538.68, SD = 127.58). However, we found no statistically significant difference between time taken to hit a circle between Stress Induction and Post-stress Recovery periods. Mean target acquisition time values are presented in Figure 5.



Fig. 5. Mean Target Acquisition Time (95% CI)

Then, we investigated the effect of stress on the touch accuracy during target acquisition. A one-way repeated measures ANOVA showed a statistically significant effect of stress on the offset size (F(2, 24) = 20.11, p < 0.01). Post-hoc comparisons using the Tukey HSD test (with Bonferroni corrections) indicate that touch offset size is significantly larger in Stress Induction (M = 49.14, SD = 25.94, p < 0.01) and Post-stress Recovery periods (M = 50.20, SD = 26.51, p < 0.01) as compared to the Baseline value of the offset size (M = 46.47, SD = 25.57). However, we found no statistically significant difference in offset size between Stress induction and Post-stress Recovery measurements (p > 0.05). Mean offset size values are visualised in Figure 6.



Fig. 6. Mean Offset Size (95% CI)

Finally, we studied the effect of stress on effective throughput. Effective throughput was calculated as suggested by Soukoreff and MacKenzie [53]. A one-way repeated measures ANOVA did not show a statistically significant effect of stress on the effective throughput during target acquisition tasks (p > 0.05).

# 24:10 • Z. Sarsenbayeva et al.

# 4.3 Visual Search Task

We measured the performance during the visual search task in terms of the time taken to memorise the target icon, and the time taken to find the target icon. A one-way repeated measures ANOVA showed a statistically significant effect of stress on the time taken to memorise an icon (F(2, 24) = 31.77, p < 0.01). Post-hoc comparisons using the Tukey HSD test (with Bonferroni corrections) indicated that participants take less time to memorise an icon in Stress Induction (M = 714.52, SD = 229.27, p < 0.01) and Post-stress Recovery periods (M = 702.31, SD = 200.85, p < 0.01) when compared to the baseline measurement (M = 800.51, SD = 331.15). However, we found no statistically significant difference in the time taken to memorise an icon between the values measured during Stress Induction and Post-stress Recovery (p > 0.05). Mean values for time taken to memorise an icon are visualised in Figure 7.



Fig. 7. Mean Time to Memorise an Icon (95% CI)

Then, we conducted a one-way repeated measures ANOVA to investigate the effect of stress on time taken to find a target. The results of the test did not show a statistically significant effect of stress on visual search time (p > 0.05). Finally, we investigated the effect of stress on the number of errors during visual search tasks. A one-way repeated measures ANOVA did not show an effect of stress on the number of errors.

#### 4.4 Text Entry Task

We measured text entry performance in terms of character entry rate, number of total errors, and total error rate as suggested by Soukoreff and MacKenzie [53]. Character entry rate was calculated as the time taken to enter a character, while number of total errors was calculated as the sum of uncorrected and corrected errors. Finally, total error rate was the ratio between the number of total errors and total entered characters. We conducted a one-way repeated measures ANOVA to investigate the effect of stress on time per character entry. The results did not reveal a statistically significant effect of stress on time per character entry (p = 0.056), with participants tending to take less time per character entry after post-stress recovery time. Furthermore, a one-way repeated measures ANOVA did not show a statistically significant effect of stress on the number of total errors (p > 0.05), nor on total error rate (p > 0.05).

As we presented our participants with two types of texts: easy and difficult, we conducted a two-way repeated measures ANOVA, accounting for the effect of difficulty and stress on the number of total errors and total error rate. We did not consider this difference for the character entry rate, as this is already accounted for with text difficulty. The results of the ANOVA showed a statistically significant effect of the difficulty of the text on the number of total errors (F(1, 24) = 34.22, p < 0.01). However, we did not find an interaction effect between

Proc. ACM Interact. Mob. Wearable Ubiquitous Technol., Vol. 3, No. 1, Article 24. Publication date: March 2019.
the difficulty of the text and stress (p > 0.05). Finally, a two-way repeated measures ANOVA did not show a statistically significant effect of the difficulty of the text on the total error rate (p > 0.01).

#### 4.5 Interviews

To familiarise ourselves with the data collected during the semi-structured interviews, we read through each response and analysed it using the following process. One researcher printed participant responses and completed initial coding, utilising a data-driven approach based on our aforementioned results. Then, three researchers compared the initial codes and merged these according to their similarity (*e.g., "I rushed the task"* and "*I got progressively slower*" merged to *"Task completion time"*). The three researchers then independently coded participants' responses for each question as based on the initial codebook. We reviewed the final coding to identify similarities that allowed thematic grouping. The main themes are described below.

4.5.1 Psychological Effects of Stress on Perceived Performance. The majority of the participants (N = 19) stated that stress affected their behaviour. For instance, participants ended up rushing through the tasks when stressed. Rushing the tasks affected their performance in terms of accuracy during target acquisition tasks. Example quotes are: "I was more annoyed and just wanted to get the tasks done, so I rushed" (P13), "I think because I rushed the task I got less accurate" (P17), "When I'm stressed I get adrenaline, and tap faster, but when I make a mistake I go slower, but still not very accurate" (P14). These findings are in line with our quantitative data, that showed that participants took significantly less time to tap circles when they were stressed as well as they were less accurate.

Participants also reported that stress had an effect on their ability to concentrate and focus. However, the comments regarding concentration were not unanimous. While some participants (N = 8) claimed that stress impaired their concentration, others claimed that stress sharpened their focus (N = 5). This concords with the fact that, for some participants their perceived performance during visual search and text entry tasks was improved, for example, "I felt I messed up during first few trials as I wasn't paying attention, but stress made me focus better" (P07), "After speech I couldn't concentrate" (P20). While for others, their perceived performance deteriorated: "I forgot the icons as I was stressed" (P04), "I was thinking about the icons, focused on them, but after the arithmetic task I wasn't focused on the icon, and 60% of the time i just pressed continue and was thinking what was the icon I needed to find" (P17), "After presentation I couldn't focus, when I was doing the icons task I was thinking about my presentation and it took longer time to focus on the icon" (P18).

4.5.2 Physical Effect of Stress on Perceived Performance. We found that stress affected participants not only psychologically, but also physically. Several participants (N = 5) claimed that stress had a physical effect on them. All of these participants reported that they felt jittery under stress, and as a result they were less accurate when completing target acquisition tasks. We recorded the following quotes from our participants: "I was more jittery after the speech, my fingers were more shaky" (P09), "My fingers were shivering, especially after the presentation. My head was cloudy, and I was not focused" (P14). These findings are in line with our quantitative results, which show that participants are less accurate in target acquisition task when they are stressed.

4.5.3 Self-Reflection. We also found that several participants reflected on their performance in the speech and arithmetic tasks. This self-reflection was mostly performed during the final part of the study when participants were left alone in the desensitisation room for a post-stress recovery period. While some participants (N = 7) expressed their concerns regarding their performance during the arithmetic task: "I was thinking that I could have done better in the arithmetic task", others (N = 7) reflected on their performance during the speech task: "I was thinking how bad I did in the presentation" (P16). These comments indicate that even after the stress-inducing tasks during post-stress recovery period, participants were still preoccupied with their performance.

#### 24:12 • Z. Sarsenbayeva et al.

	Baseline			Situational Impairments								
	Warm	Silent	No stress	Cold	Music Fast	Music Slow	Noise Indoor	Noise Outdoor	Speech Meaningful	Speech Meaningless	Stress	Post-Stress Recovery
Time to tap a circle, ms	593.00	591.84	538.68	603.00 + 1.7%	577.76 - 2.4%	574.15 - 3.0%	583.56 - 1.4%	573.56 - 3.1%	592.81 + 0.2%	609.38 + 3.0%	534.77 - 0.7%	520.18 - 3.4%
Offset size, px	41.34	39.38	46.47	<b>42.66</b> + 3.1%	42.95 +9.1%	42.11 + 6.9%	41.06 + 4.3%	40.91 + 3.9%	42.90 + 8.9%	39.47 + 0.3%	<b>49.14</b> + 5.7%	50.20 + 8.0%
Time to memorise an icon, ms	815.00	748.76	800.51	854.00 + 4.8%	737.71 - 1.5%	745.15 - 0.5%	712.11 - 4.9%	743.56 - 0.7%	753.20 + 0.6%	738.90 - 1.3%	714.52 - 4.6%	702.31 - 12.3%
Time to find an icon, ms	1632.24	1587.74	1602.45	<b>1942.46</b> + 19.0%	1564.70 - 1.4%	1753.19 + 10.4%	1543.12 - 2.8%	1633.15 + 2.9%	1637.54 + 3.1%	1520.99 - 4.2%	1496.50 - 6.6%	1506.38 - 6.0%
Character entry rate, ms/char	n/a	454.39	560.88	n/a	488.11 + 7.4%	485.14 + 6.8%	480.96 + 5.8%	523.89 + 15.3%	539.82 + 18.8%	<b>508.99</b> + 12.0%	547.16 - 2.4%	495.83 - 11.6%
Total error rate, %	n/a	7.54	7.46	n/a	7.45 - 1.2%	7.38 - 2.1%	8.54 + 13.3%	8.22 + 9.0%	7.09 - 6.0%	7.07 - 6.2%	10.50 + 40.8%	9.40 + 26.0%

Table 1. Comparison of the Effects of Situational Impairments against Respective Baseline.

Mean values per condition, relative change in % as compared to baseline.

Colour code explanation: Blue – findings of study on ambient temperature [42].

Yellow – findings of study on ambient noise [44]. Gray – findings of this study.

#### 4.6 Comparison between Situational Impairments

In this section, we compare stress-, cold- and noise-induced situational impairments. This comparison allows us to increase our understanding on the impact of different situational impairments, thus contributing towards this growing research agenda.

We present the mean values for the following variables in Table 1: time taken to tap a circle, size of the offset, time taken to memorise an icon, time taken to find an icon, character entry rate, and total error rate in typing. The last two variables are not available for the cold-induced situational impairments, as the authors did not quantify performance during the text entry task under cold ambience [42]. Furthermore, we calculate the magnitude of the impact of each situational impairment, when compared to their respective baseline (as a percentage). Given these calculations and the fact that the experimental tasks were identical between the three studies, we are able to directly compare magnitudes of the effects of each factor causing situational impairments (stress, cold ambience [42], ambient noise [44]).

From the table, we observe that when it comes to hitting buttons and icons on the screen (target acquisition) stress has a more detrimental effect than other situational impairments, even than cold. In fact, our results show that during post-stress recovery, participants are less accurate at hitting targets than when they are exposed to cold ambience. Further, stress affected participants differently when compared to noise: character entry rate dropped by -11.6% and total error rate increased by +26.0%.

Interestingly, however, in Table 1 we see although stress is detrimental to target acquisition, its effect is not as acute as the one of music with fast tempo (+9.1%), and meaningful speech (+8.9%). Nevertheless, we can conclude that the effect of stress is the greatest on memorising time (-12.3%) compared to cold or ambient noise.

#### 5 DISCUSSION

Our findings suggest that even though we do not find a significant effect of stress on throughput in the target acquisition task, it does lead participants to favour speed to the detriment of accuracy. We also show that under stress our participants are quicker to memorise an icon during the visual search task. However, our results do not reveal a significant effect of stress on the text entry task. Below we discuss our findings with regard to existing work. We then discuss the benefits and potential future contributions of this research to the HCI community.

#### 5.1 Effects of Stress on Interaction

Our results show that the target acquisition time and memorising time are significantly shorter when participants tap circles and complete visual search tasks after the post-stress recovery period. Our findings are in line with prior literature, which shows that people tend to rush and require significantly less time to complete tasks when stressed [33]. Moreover, our participants confirm this behaviour during the interview sessions. They mention that stress caused them anxiety and discomfort, and, hence they rushed through the tasks. Furthermore, the participants report that stress affected their concentration and focus. As a result, they did not pay attention to the icon shown during the visual search task. Instead, they proceeded to the next screen for the sake of completing the task.

Additionally, our results show that stress causes our participants to be less accurate when performing target acquisition tasks. This result also reflects prior literature, which shows that people tend to have higher error rates in completing tasks when stressed [33]. Interestingly, this is also the case in the post-stress recovery measurements. During our interviews the participants claimed that they spent this post-stress recovery period self-reflecting on the tasks, which kept them stressed throughout the recovery period. Consequently, their performance during mobile interaction was negatively affected. Our findings are in line with literature [20], as the authors of the TSST report a high cortisol level in participants during the post-stress recovery period.

Nevertheless, our findings do not show a statistically significant effect of stress on effective throughput in the target acquisition task. This result is in line with prior findings by MacKenzie and Isokoski, where they show that throughput remains constant when target acquisition time and accuracy drop during tapping tasks [28]. This can be explained by the fact that throughput accounts for both speed and accuracy and as these two variables change in opposite directions, throughput does not significantly change [28]. However, we note that while throughput remained relatively constant in this scenario, this is likely not to be the case when a stressed user interacts with a real-world application that has a cost associated with errors.

Furthermore, our results show that text difficulty had a larger effect on performance during the text entry task than participant stress levels. Regarding the text entry task several participants (N = 9) report that they made comparatively more errors after they experienced stress. While our quantitative results show a tendency towards this being the case, this effect was not statistically significant.

#### 5.2 Contrasting Situational Impairments

Previous work has highlighted the importance of accumulating knowledge within different areas of HCI for the formation of paramount research themes [24, 25]. Here, we contribute towards this notion by contrasting the magnitudes of the effects of different situational impairments on mobile interaction.

Previous work, utilising the same tasks reported in this study, has shown that text entry rate is significantly affected by ambient noise, particularly when considering meaningful speech (speech in a language the participant understands) [44]. However, in our study, participants' performance during text entry (in terms of character entry rate) was not significantly affected by stress, suggesting that stress has a smaller effect on participants' ability to perform text entry compared to meaningful speech. This is not surprising given the nature of the text entry task. Meaningful speech is particularly disruptive when a person is thinking about what they should say or write [30].

#### 24:14 • Z. Sarsenbayeva et al.

Furthermore, in Table 1 we can see that the effect of cold ambience on the target access time was opposite to the one of stress. To be precise, target acquisition time was longer in cold environment [42]; however, it was shorter during the post-stress recovery period (as compared to the relative baselines). This can be explained by the nature of the environment, as cold ambience decreases manual dexterity [60], and hence, increases the completion time for the tasks involving fine-motor manipulations [66]. Whereas, stress is known to increase anxiety [23], thus leading people to rush to complete the tasks [33]. This effect was also highlighted in our qualitative data; and is applicable not only for tasks requiring fine-motor skills, but also for visual search tasks. For example, our participants spent less time on memorising and searching for icons, as compared to the participants from the study presented in [42], where they took longer time both to memorise and find icons. Interestingly, cold-, noise-, and stress-induced situational impairments have a roughly comparable adverse effect on the tapping accuracy. This can be explained by the nature of the task, as both external and internal factors affect accuracy in only one direction – negative. The only difference is the magnitude of the effect, shown in detail in Table 1.

#### 5.3 Implications for Research and Design

Previous research has emphasised the importance of understanding the effects of situational impairments on mobile interaction, as it further enables the construction of sensing mechanisms to detect situational impairments and, thus, adapt the interface accordingly [43]. Our study shows that stress can impair mobile interaction, and, hence, this information can be used to adapt the interface to accommodate such impairments. For instance, to mitigate the effects of stress-induced situational impairments on target acquisition tasks, it is possible to increase target sizes or use techniques proposed in the literature to improve input accuracy, such as GraspZoom [34] and Fat Thumb [6]. GraspZoom allows zooming on a particular part of the screen with a long press using only the thumb [34]. Fat Thumb includes only two interaction gesture modes: panning and zooming, with gesture being defined by the contact size of the thumb [6]. We acknowledge that the effect size of stress on interaction performance might not seem to be strong enough to impair the user experience in real world scenario. However, we expect that when completing more complex tasks on a smartphone, the effect size would be more pronounced.

However, it is necessary to first detect stress before adaptation actually takes place. Wearable sensors have successfully been leveraged for this purpose by measuring physiological measurements such as skin conductivity [50] and heart rate variability [56]. In our study we use HRV measured through a wearable device alongside self-reports, to confirm that participants are indeed being affected by the TSST protocol. Beyond sensor data and self-reports, previous work has shown that daily stress can also be inferred from smartphone usage, personality traits, and weather data [5]. Moreover, the Intel Mobile Heart Health [36] prototype uses data from sources such as mobile phone-based ecological momentary assessment and a small electrocardiograph sensor with an accelerometer to detect changes in heart rate variability, activity, and mood. If individualised threshold values are reached, the mobile phone delivers cognitive behavioural and mindfulness techniques designed to reduce stress [36]. Within the literature on situational impairments, there are also several examples of sensor information being used to detect certain contextual factors. For example, Goel *et al.* [11] used a smartphone's battery temperature to detect cold-induced situational impairments [45].

Ultimately, this detection should ideally be performed unobtrusively and without creating additional stress [14]. Unobtrusive and continuous stress detection would benefit mobile device users. For example, an individual being aware of their exposure to stress, could change their behaviour to eliminate unnecessary stressors [14]. A smartphone's operating system that understands the user's application usage behaviour [17, 58], and having detected the user being stressed, could prevent unnecessary notifications or updates being presented [14], and adapt the interface to minimise errors.

#### 5.4 Limitations

We acknowledge a number of limitations in our study. First, the study settings were strictly controlled. It is possible that in a more naturalistic environment, participants can experience a stronger level of stress. However, it was necessary to follow an established protocol to strictly control stress induction and avoid causing harm to our participants. The types of smartphone tasks presented in this study were limited to target acquisition, visual search, and text entry, whereas, in a naturalistic environment, users may perform more complex tasks, requiring more cognitive demand [2]. Nevertheless, in this study we reported a statistically significant effect of stress on basic smartphone tasks (*e.g.*, target acquisition and visual search), which suggests that this effect would be more pronounced when completing more complex tasks during mobile interaction. Furthermore, we did not find any statistically significant effect of stress on text entry tasks. However, we acknowledge that stress might have an impact on text entry performance with more complex sentences. Future research is needed to investigate this assumption. Finally, we restricted our participants to only one interaction mode with the smartphone – using the index finger. We argue that controlling the interaction mode was necessary to draw a fair comparison between the effects of stress, ambient noise [44], and cold ambience [42].

#### 6 CONCLUSION

In this work, we investigate the effects of stress on performance on three common mobile interaction tasks: target acquisition, visual search, and text entry. We use the Trier Social Stress Test to induce stress on participants. Our findings show that the target acquisition time and the time to memorise an icon become significantly shorter during stress and post-stress recovery periods. We also show that stress deteriorates participants' accuracy during target acquisition tasks. These findings can be used to inform how interfaces should adapt to accommodate such impairments.

We then compare the effects of cold-, noise-, and stress-induced situational impairments on performance during mobile interaction. Our results show that the effect was more pronounced on target acquisition tasks in terms of task completion time and accuracy. Our findings on the effects of stress on mobile interaction extend our understanding of situational impairments. This knowledge is paramount to inform the development of adaptive interfaces that accommodate such impairments.

#### ACKNOWLEDGMENTS

This work is partially funded by a Samsung Global Research Outreach grant and JST ERATO Grant Number JPMJER1501, Japan. Eduardo Velloso is the recipient of an Australian Research Council Discovery Early Career Award (Project Number: DE180100315) funded by the Australian Government.

#### REFERENCES

- [1] 2007. Work-related stress. http://www.eurofound.europa.eu/ewco/reports/TN0502TR01/TN0502TR01/Ddf
- [2] Brian P Bailey, Joseph A Konstan, and John V Carlis. 2001. The Effects of Interruptions on Task Performance, Annoyance, and Anxiety in the User Interface.. In Interact, Vol. 1. 593–601.
- [3] Gerald Bauer and Paul Lukowicz. 2012. Can smartphones detect stress-related changes in the behaviour of individuals?. In Pervasive Computing and Communications Workshops (PERCOM Workshops), 2012 IEEE International Conference on. IEEE, 423–426.
- [4] Gary G Berntson and John T Cacioppo. 2004. Heart rate variability: Stress and psychiatric conditions. Dynamic electrocardiography (2004), 57–64.
- [5] Andrey Bogomolov, Bruno Lepri, Michela Ferron, Fabio Pianesi, and Alex Sandy Pentland. 2014. Daily stress recognition from mobile phone data, weather conditions and individual traits. In Proceedings of the 22nd ACM international conference on Multimedia. ACM, 477–486.
- [6] Sebastian Boring, David Ledo, Xiang'Anthony' Chen, Nicolai Marquardt, Anthony Tang, and Saul Greenberg. 2012. The fat thumb: using the thumb's contact size for single-handed mobile interaction. In Proceedings of the 14th international conference on Human-computer interaction with mobile devices and services. ACM, 39–48.

#### 24:16 • Z. Sarsenbayeva et al.

- [7] Wolfram Boucsein. 2012. Electrodermal activity. Springer Science & Business Media.
- [8] Jason J Braithwaite, Derrick G Watson, Robert Jones, and Mickey Rowe. 2013. A guide for analysing electrodermal activity (EDA) & skin conductance responses (SCRs) for psychological experiments. *Psychophysiology* 49, 1 (2013), 1017–1034.
- [9] Sheldon Cohen, Tom Kamarck, and Robin Mermelstein. 1983. A global measure of perceived stress. Journal of health and social behavior (1983), 385–396.
- [10] Marc Exposito, Rosalind W Picard, and Javier Hernandez. 2018. Affective keys: towards unobtrusive stress sensing of smartphone users. In Proceedings of the 20th International Conference on Human-Computer Interaction with Mobile Devices and Services Adjunct. ACM, 139–145.
- [11] Mayank Goel, Leah Findlater, and Jacob Wobbrock. 2012. WalkType: Using Accelerometer Data to Accomodate Situational Impairments in Mobile Touch Screen Text Entry. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '12). ACM, New York, NY, USA, 2687–2696. https://doi.org/10.1145/2207676.2208662
- [12] Jorge Goncalves, Zhanna Sarsenbayeva, Niels van Berkel, Chu Luo, Simo Hosio, Sirkka Risanen, Hannu Rintamäki, and Vassilis Kostakos. 2017. Tapping Task Performance on Smartphones in Cold Temperature. *Interacting with Computers* 29, 3 (2017), 355–367. https://doi.org/10.1093/iwc/iww029
- [13] Niels Henze, Enrico Rukzio, and Susanne Boll. 2011. 100,000,000 Taps: Analysis and Improvement of Touch Performance in the Large. In Proceedings of the 13th International Conference on Human Computer Interaction with Mobile Devices and Services (MobileHCI '11). ACM, New York, NY, USA, 133–142. https://doi.org/10.1145/2037373.2037395
- [14] Javier Hernandez, Pablo Paredes, Asta Roseway, and Mary Czerwinski. 2014. Under pressure: sensing stress of computer users. In Proceedings of the SIGCHI conference on Human factors in computing systems. ACM, 51–60.
- [15] Nis Hjortskov, Dag Rissén, Anne Katrine Blangsted, Nils Fallentin, Ulf Lundberg, and Karen Søgaard. 2004. The effect of mental stress on heart rate variability and blood pressure during computer work. *European journal of applied physiology* 92, 1-2 (2004), 84–89.
- [16] Marko Jelici, Elke Geraerts, Harald Merckelbach, and Ramona Guerrieri. 2004. Acute stress enhances memory for emotional words, but impairs memory for neutral words. International Journal of Neuroscience 114, 10 (2004), 1343–1351.
- [17] Simon L. Jones, Denzil Ferreira, Simo Hosio, Jorge Goncalves, and Vassilis Kostakos. 2015. Revisitation Analysis of Smartphone App Use. In Proceedings of the 2015 ACM International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp '15). ACM, New York, NY, USA, 1197–1208. https://doi.org/10.1145/2750858.2807542
- [18] Indika Karunaratne, Ajantha Sanjeewa Atukorale, and Hemamali Perera. 2011. Surveillance of human-computer interactions: A way forward to detection of users' Psychological Distress. In *Humanities, Science and Engineering (CHUSER), 2011 IEEE Colloquium on*. IEEE, 491–496.
- [19] J Peter Kincaid, Robert P Fishburne Jr, Richard L Rogers, and Brad S Chissom. 1975. Derivation of new readability formulas (automated readability index, fog count and flesch reading ease formula) for navy enlisted personnel. Technical Report. Naval Technical Training Command Millington TN Research Branch.
- [20] Clemens Kirschbaum, Karl-Martin Pirke, and Dirk H Hellhammer. 1993. The 'Trier Social Stress Test'-a tool for investigating psychobiological stress responses in a laboratory setting. *Neuropsychobiology* 28, 1-2 (1993), 76–81.
- [21] Clemens Kirschbaum, Oliver T Wolf, Mark May, W Wippich, and Dirk H Hellhammer. 1996. Stress-and treatment-induced elevations of cortisol levels associated with impaired declarative memory in healthy adults. *Life sciences* 58, 17 (1996), 1475–1483.
- [22] Sabrina Kuhlmann, Marcel Piel, and Oliver T Wolf. 2005. Impaired memory retrieval after psychosocial stress in healthy young men. Journal of Neuroscience 25, 11 (2005), 2977–2982.
- [23] Richard S Lazarus, James Deese, and Sonia F Osler. 1952. The effects of psychological stress upon performance. *Psychological bulletin* 49, 4 (1952), 293.
- [24] Yong Liu, Jorge Goncalves, Denzil Ferreira, Simo Hosio, and Vassilis Kostakos. 2014. Identity Crisis of Ubicomp?: Mapping 15 Years of the Field's Development and Paradigm Change. In Proceedings of the 2014 ACM International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp '14). ACM, New York, NY, USA, 75–86. https://doi.org/10.1145/2632048.2632086
- [25] Yong Liu, Jorge Goncalves, Denzil Ferreira, Bei Xiao, Simo Hosio, and Vassilis Kostakos. 2014. CHI 1994-2013: Mapping Two Decades of Intellectual Progress Through Co-word Analysis. In Proceedings of the 32nd Annual ACM Conference on Human Factors in Computing Systems (CHI '14). ACM, New York, NY, USA, 3553–3562.
- [26] SJ Lupien, S Gaudreau, BM Tchiteya, F Maheu, S Sharma, NPV Nair, RL Hauger, BS McEwen, and MJ Meaney. 1997. Stress-induced declarative memory impairment in healthy elderly subjects: relationship to cortisol reactivity. *The Journal of Clinical Endocrinology & Metabolism* 82, 7 (1997), 2070–2075.
- [27] Sonia J Lupien, Christian J Gillin, and Richard L Hauger. 1999. Working memory is more sensitive than declarative memory to the acute effects of corticosteroids: A dose-response study in humans. *Behavioral neuroscience* 113, 3 (1999), 420.
- [28] I Scott MacKenzie and Poika Isokoski. 2008. Fitts' throughput and the speed-accuracy tradeoff. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. ACM, 1633–1636.
- [29] Dorothy Irene Marquart. 1948. The pattern of punishment and its relation to abnormal fixation in adult human subjects. The Journal of general psychology 39, 1 (1948), 107–144.

- [30] Randi C Martin, Michael S Wogalter, and Janice G Forlano. 1988. Reading comprehension in the presence of unattended speech and music. Journal of memory and language 27, 4 (1988), 382–398.
- [31] Daniel J McDuff, Javier Hernandez, Sarah Gontarek, and Rosalind W Picard. 2016. Cogcam: Contact-free measurement of cognitive stress during computer tasks with a digital camera. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. ACM, 4000–4004.
- [32] Fred McKinney. 1933. Certain emotional factors in learning and efficiency. The Journal of General Psychology 9, 1 (1933), 101-116.
- [33] Fred McKinney, George B Strother, Ruth R Hines, and Ruth A Allee. 1951. Experimental frustration in a group test situation. The Journal of Abnormal and Social Psychology 46, 3 (1951), 316.
- [34] Takashi Miyaki and Jun Rekimoto. 2009. GraspZoom: zooming and scrolling control model for single-handed mobile interaction. In Proceedings of the 11th International Conference on Human-Computer Interaction with Mobile Devices and Services. ACM, 11.
- [35] Sachi Mizobuchi, Mark Chignell, and David Newton. 2005. Mobile text entry: relationship between walking speed and text input task difficulty. In Proceedings of the 7th international conference on Human computer interaction with mobile devices & services. ACM, 122–128.
- [36] Margaret Morris and Farzin Guilak. 2009. Mobile heart health: project highlight. IEEE Pervasive Computing 2 (2009), 57–61.
- [37] John W Newcomer, Gregg Selke, Angela K Melson, Tamara Hershey, Suzanne Craft, Katherine Richards, and Amy L Alderson. 1999. Decreased memory performance in healthy humans induced by stress-level cortisol treatment. Archives of general psychiatry 56, 6 (1999), 527–533.
- [38] Alexander Ng, Stephen A. Brewster, and John H. Williamson. 2014. Investigating the Effects of Encumbrance on One- and Two- Handed Interactions with Mobile Devices. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (CHI '14). ACM, New York, NY, USA, 1981–1990. https://doi.org/10.1145/2556288.2557312
- [39] Jessica D Payne, Lynn Nadel, John JB Allen, Kevin GF Thomas, and W Jake Jacobs. 2002. The effects of experimentally induced stress on false recognition. *Memory* 10, 1 (2002), 1–6.
- [40] Manuel Rodrigues, Sérgio Gonçalves, Davide Carneiro, Paulo Novais, and Florentino Fdez-Riverola. 2013. Keystrokes and clicks: Measuring stress on e-learning students. In Management Intelligent Systems. Springer, 119–126.
- [41] Akane Sano and Rosalind W Picard. 2013. Stress recognition using wearable sensors and mobile phones. In Affective Computing and Intelligent Interaction (ACII), 2013 Humaine Association Conference on. IEEE, 671–676.
- [42] Zhanna Sarsenbayeva, Jorge Goncalves, Juan García, Simon Klakegg, Sirkka Rissanen, Hannu Rintamäki, Jari Hannu, and Vassilis Kostakos. 2016. Situational Impairments to Mobile Interaction in Cold Environments. In Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp '16). ACM, New York, NY, USA, 85–96. https://doi.org/10.1145/2971648.2971734
- [43] Zhanna Sarsenbayeva, Niels van Berkel, Chu Luo, Vassilis Kostakos, and Jorge Goncalves. 2017. Challenges of situational impairments during interaction with mobile devices. In Proceedings of the 29th Australian Conference on Computer-Human Interaction. ACM, 477–481.
- [44] Zhanna Sarsenbayeva, Niels van Berkel, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. 2018. Effect of Distinct Ambient Noise Types on Mobile Interaction. Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies 2, 3 (2018).
- [45] Zhanna Sarsenbayeva, Niels van Berkel, Aku Visuri, Sirkka Rissanen, Hannu Rintamäki, Vassilis Kostakos, and Jorge Goncalves. 2017. Sensing Cold-Induced Situational Impairments in Mobile Interaction Using Battery Temperature. Proc. ACM Interact. Mob. Wearable Ubiquitous Technol. 1, 3, Article 98 (Sept. 2017), 9 pages. https://doi.org/10.1145/3130963
- [46] Sidas Saulynas, Lawrence Burgee, and Ravi Kuber. 2017. All situational impairments are not created equal: A classification system for situational impairment events and the unique nature of severely constraining situational impairments. *iConference 2017* (2017).
- [47] Michael A Sayette. 1999. Does drinking reduce stress? Alcohol research and health 23, 4 (1999), 250–255.
  [48] Bastian Schildbach and Enrico Rukzio. 2010. Investigating selection and reading performance on a mobile phone while walking. In Proceedings of the 12th international conference on Human computer interaction with mobile devices and services. ACM, 93–102.
- [49] Andrew Sears, Min Lin, Julie Jacko, and Yan Xiao. 2003. When computers fade: Pervasive computing and situationally-induced impairments and disabilities. In *HCI International*, Vol. 2. 1298–1302.
- [50] Cornelia Setz, Bert Arnrich, Johannes Schumm, Roberto La Marca, Gerhard Tröster, and Ulrike Ehlert. 2010. Discriminating stress from cognitive load using a wearable EDA device. IEEE Transactions on information technology in biomedicine 14, 2 (2010), 410-417.
- [51] Riccardo Sioni and Luca Chittaro. 2015. Stress detection using physiological sensors. Computer 48, 10 (2015), 26-33.
- [52] Aaron Smith. 1967. The serial sevens subtraction test. Archives of Neurology 17, 1 (1967), 78-80.
- [53] R William Soukoreff and I Scott MacKenzie. 2003. Metrics for text entry research: an evaluation of MSD and KSPC, and a new unified error metric. In Proceedings of the SIGCHI conference on Human factors in computing systems. ACM, 113–120.
- [54] Charles D Spielberger. 2010. State-Trait anxiety inventory. Wiley Online Library.
- [55] Mark A Staal. 2004. Stress, cognition, and human performance: A literature review and conceptual framework. (2004).
- [56] David Sun, Pablo Paredes, and John Canny. 2014. MouStress: detecting stress from mouse motion. In *Proceedings of the SIGCHI conference* on Human factors in computing systems. ACM, 61–70.
- [57] Garreth W. Tigwell, David R. Flatla, and Rachel Menzies. 2018. It's Not Just the Light: Understanding the Factors Causing Situational Visual Impairments During Mobile Interaction. In Proceedings of the 10th Nordic Conference on Human-Computer Interaction (NordiCHI '18). ACM, New York, NY, USA. https://doi.org/10.1145/3240167.3240207

#### 24:18 • Z. Sarsenbayeva et al.

- [58] Niels van Berkel, Chu Luo, Theodoros Anagnostopoulos, Denzil Ferreira, Jorge Goncalves, Simo Hosio, and Vassilis Kostakos. 2016. A Systematic Assessment of Smartphone Usage Gaps. In Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems (CHI '16). ACM, New York, NY, USA, 4711–4721. https://doi.org/10.1145/2858036.2858348
- [59] Ruud Van den Bos, Marlies Harteveld, and Hein Stoop. 2009. Stress and decision-making in humans: performance is related to cortisol reactivity, albeit differently in men and women. *Psychoneuroendocrinology* 34, 10 (2009), 1449–1458.
- [60] JA Vaughan, EA Higgins, GE Funkhouser, Elinore M Galerston, et al. 1968. Effects of body thermal state on manual performance. Aerospace medicine 39, 12 (1968), 1310–1315.
- [61] Elinor Verville. 1946. The effect of emotional and motivational sets on the perception of incomplete pictures. *The Pedagogical Seminary and Journal of Genetic Psychology* 69, 2 (1946), 133–145.
- [62] J Wahlström, M Hagberg, P Johnson, J Svensson, and D Rempel. 2002. Influence of time pressure and verbal provocation on physiological and psychological reactions during work with a computer mouse. *European journal of applied physiology* 87, 3 (2002), 257–263.
- [63] Jacob O Wobbrock. 2006. The future of mobile device research in HCI. In CHI 2006 workshop proceedings: what is the next generation of human-computer interaction. 131–134.
- [64] Owen M Wolkowitz, Victor I Reus, Herbert Weingartner, Karen Thompson, Alan Breier, et al. 1990. Cognitive effects of corticosteroids. The American journal of psychiatry 147, 10 (1990), 1297.
- [65] Robert M Yerkes and John D Dodson. 1908. The relation of strength of stimulus to rapidity of habit-formation. *Journal of comparative neurology and psychology* 18, 5 (1908), 459–482.
- [66] Mark A Yeshnik. 1988. The variability of German winter temperature in relation to human performance and its implications for tactical military operations. Technical Report. Army Military Personnel Center Alexandria VA.
- [67] AH Young, BJ Sahakian, TW Robbins, and PJ Cowen. 1999. The effects of chronic administration of hydrocortisone on cognitive function in normal male volunteers. *Psychopharmacology* 145, 3 (1999), 260–266.

Received November 2018; revised January 2019; accepted January 2019

## Chapter 6

# Quantifying the Effects of Dim Ambient Light on Mobile Interaction

This chapter presents our work on quantifying the effects of dim ambient light on mobile interaction. Prior research has previously established the negative effects of bright sunlight on mobile interaction, as with an increased ambient illuminance, smartphone screens become more challenging to use [77] due to different factors (*e.g.*, reduced screen contrast [220] or screen glare [139]). However, the effect of dim ambient light on mobile interaction has not been thoroughly investigated. The existing knowledge on this area is limited to the study conducted by Lee *et al.* [130], where they establish a negative effect of low ambient light on three smartphone tasks (target acquisition, visual search, and text entry) under three ambient light conditions:

- Baseline condition set to 330 lux, which is a standard illuminance level in the office environment [190];
- Surrounding dim illuminance set to 20 lux;
- Dim illuminance caused by participants wearing sunglasses under the baseline condition. The sunglasses were of category 2 non-polarised lenses, commonly used among general population.

The study was conducted in laboratory settings and we strictly controlled for the illuminance level to exclude the presence of natural sunlight or other potential confounding factors (*e.g.*, screen glare). These conditions were selected as they are commonly present during mobile interaction (*e.g.*, in the office room with no windows or in a dark room).

The findings of our study show that dim ambient light negatively affects the target acquisition task. In particular, participants were significantly slower and less accurate when accessing targets under both surrounding and local dim light conditions as compared to the baseline condition. Our results also show that the participants took longer time to memorise icons in visual search task when wearing sunglasses.

This work contributes towards understanding of dim light-induced SIIDs and their effect on smartphone interaction. Technicalities of our approach are detailed in the attached publication in Section 6.1.

## 6.1 Publication

©Copyright 2019, IFIP International Federation for Information Processing. This is the authors' version of the work. It is posted here for your personal use. Not for redistribution. The definitive Version of Record was published in:

Sarsenbayeva, Z., van Berkel, N., Jiang, W., Hettiachchi, D., Kostakos, V., and Goncalves, J. (2019, September). Effect of Ambient Light on Mobile Interaction. In *IFIP Conference on Human-Computer Interaction*(pp. 465-475). Springer, Cham. DOI: https://doi.org/10.1007/978-3-030-29387-1\_26

Ethics ID: 1852612.1, The University of Melbourne Human Ethics Advisory Group.

#### Effect of Ambient Light on Mobile Interaction

Zhanna Sarsenbayeva<sup>1</sup>, Niels van Berkel<sup>2</sup>,

Weiwei Jiang<sup>1</sup>, Danula Hettiachchi<sup>1</sup>, Vassilis Kostakos<sup>1</sup>, and Jorge Goncalves<sup>1</sup>

<sup>1</sup> The University of Melbourne, Australia, firstname.lastname@unimelb.edu.au <sup>2</sup> University College London, UK, n.vanberkel@ucl.ac.uk

**Abstract.** In this work we investigate the effect of ambient light on performance during mobile interaction. We evaluate three conditions of ambient light – normal light, dimmed light, normal light while wearing sunglasses. Our results show that wearing sunglasses and dimmed light negatively affect reaction time, while dimmed light negatively affects accuracy performance in target acquisition tasks. We also show that wearing sunglasses increases memorising time in visual search tasks. Our study contributes to the growing body of research on the effects of different situational impairments on mobile interaction.

Keywords: Mobile interaction. Situational visual impairments. Ambient light.

#### 1 Introduction

Smartphones have become an integral part of human daily life, and a focus of research conducted in our community [8,9]. People find themselves using their smartphones under various challenging contexts [23]. Factors such as cold ambience [2,18], encumbrance [10], walking [1], and ambient noise [16] have all been shown to hinder smartphone interaction [15] and cause situational impairments [19,24]. While the effects of a number of situational impairments on mobile interaction have been studied and are established within the HCI research community, many situational impairments remain underexplored. In their overview of situational impairments, Sarsenbayeva *et al.* [15] identify a research gap concerning the effects of ambient light on mobile interaction, despite the fact that it is common to use one's smartphone in varying light conditions (*e.g.*, watching a movie with the lights off or interacting with the device while wearing sunglasses).

Therefore, in this paper we investigate the effect of ambient light conditions on mobile interaction. We quantify mobile interaction performance in terms of three everyday smartphone activities – target acquisition, visual search, and text entry – under three distinct ambient lighting conditions: 1) normal light condition (operationalised as recommended indoor light levels for easy to normal office work [12]), 2) dimmed light condition, and 3) normal light condition while wearing sunglasses. We limit our investigation to the effect of reduced illuminance conditions. We do not study the effect of bright light on smartphone interaction for a number of reasons. First, the existing literature has already established the adverse effect of bright light on performance in visual tasks on mobile device screens [7]. Second, we want to exclude the effect of confounding parameters, such as glare, that is caused by bright light and leads to a decrease in visual task performance on mobile phone screens [7].

Our study shows that dimmed light, as well as wearing sunglasses, negatively affects mobile interaction performance in terms of target acquisition time. We also show that

#### 2 Z. Sarsenbayeva et al.

tapping accuracy decreases under the dimmed light condition, while wearing sunglasses increases target memorising time. Our work contributes to the growing body of research in the HCI community on situational impairments.

#### 2 Related Work

#### 2.1 Situational Visual Impairments during Mobile Interaction

Tigwell *et al.* [22] identified ambient light as one of the leading causes of visual situational impairments. Ambient light has been shown to affect people's perception as well as the clarity of a smartphone's display. For example, Gong et al. [3] show that as ambient light intensity increases, mobile device screens become more challenging to use. This might be because increasing ambient brightness while decreasing monitor brightness reduces colour differentiation abilities, as found by Reinecke et al. [13]. Furthermore, it has been shown that for illuminance levels higher than 1000 lx, participants' visual task performance declines at a faster rate compared to illuminance levels lower than 1000 lx, due to screen glare [7]. Dimmed ambient light has also been shown to visually impair users; however, only limited research has investigated its effect. For example, Lee et al. [6] investigated the effect of ambient illuminance on performance while reading e-papers. They found that search speed and illuminance level were directly proportional: with low search speed being associated with low levels of illuminance. Liu et al. [7] studied the effect of ambient light on handheld display image quality. The authors found that in darker environments, participants performed better in visual tasks as compared to bright environments. These findings are in line with findings from Kim et al. [5] which demonstrate that perceived image quality on screens decreases in bright environments. However, both of the aforementioned studies featured a limited number of participants (3 participants in [7], 10 participants in [5]). Furthermore, both of the studies focused on the perception of image quality on mobile device screens.

#### 3 Method

In this study, we investigate the effect of an environmental factor – ambient light – on mobile interaction. In particular, we focus on mobile interaction under dimmed light conditions. We quantify interaction performance across three typical smartphone tasks: target acquisition, visual search, and text entry. We used the tasks developed and presented in a study by Sarsenbayeva *et al.* [16] in order to directly compare the effect of ambient light-induced situational impairments to the established effects of cold- [2, 18], noise- [16], and stress-induced [14] situational impairments.

#### 3.1 Smartphone Tasks

In this study, we used a Samsung Galaxy S7 smartphone running Android 7.0 with  $1080 \times 1920$ px screen size (similar to the one used in the studies by Sarsenbayeva *et al.* [16, 18]). To minimise sequence effects, participants completed the three tasks in a counterbalanced order. Furthermore, we minimised any potential learning effects by asking our participants to undergo extensive training in all three tasks prior to the start of the actual experiment. The participants completed the tasks in a standing position, interacting with the phone with the index finger of their dominant hand while holding the phone in their non-dominant hand.



Fig. 1: Interface of the application with Target Acquisition Task (A), Visual Search Task (B-C), and Text Entry Task with user's input for easy and difficult texts (D-E).

**Target Acquisition** In this task, participants tap circular targets (Radius = 135px) with an indicated centre (Figure 1-A). The targets appear on a random position on the screen, one circle at a time. We asked our participants to tap the centre of the circles as precisely and quickly as possible. We measure participants' performance in terms of their reaction speed (time required to tap targets) and accuracy (offset size).

**Visual Search** In this task, participants are asked to find a target icon among 24 other icons, arranged according to a  $4 \times 6$  grid [4]. The participants are first shown the icon, and given as much time as required to memorise it (Figure 1-B). Then, participants must find this icon in the subsequent screen (Figure 1-C). Target icons are selected randomly from the pool, and the icons are placed at random positions on the screen to minimise any potential learning effects. We quantify participants' performance in terms of cognition (time to memorise an icon), reaction (time to find an icon), and accuracy (error rate).

**Text Entry** In this task, participants are instructed to type a snippet of text shown in a text box. The texts are of two difficulties: 1) easy – consisting of only one sentence with commonly used words (Figure 1-D), and 2) difficult – consisting of several sentences with outdated words (Figure 1-E). For each round, participants are presented a randomly selected easy sentence (10 in total) and a randomly selected difficult sentence (10 in total). We measure how quick (character entry rate) and accurate (error rate) participants were in entering the text.

#### 3.2 Participants

We recruited 28 participants through our university's mailing lists. Participants are between 18 and 33 (M=23,SD=3.70) years old. In total, we recruited 19 female and 9 male participants. Our participants have a diverse range of educational background (*e.g.*, Accounting, Actuarial Sciences, Biomedicine, Business, Chemistry, Food Science, Urban Planning). All participants have normal or corrected-to-normal vision (contact lenses) and are right-handed. All of our participants were used to wearing sunglasses.

#### 4 Z. Sarsenbayeva et al.

#### 3.3 Procedure

Our experiment contains three conditions: 1) normal light condition (recommended indoor light levels for easy to normal office work [12]), 2) dimmed light condition, and 3) normal light condition while wearing sunglasses. We followed the guidelines for illuminance standards in a working environment, and hence set the room's illuminance to 335 lx for the normal light condition [12]. In the dimmed light condition, the illuminance of the room was set to 20 lx, a light level which we consider a dark environment to perform most activities. Finally, for the third condition, participants were required to wear nonpolarised sunglasses with category 2 lenses under the same illuminance as the normal light condition. Our choice of sunglasses is justified by its popularity of use among the general population, as a category 2 lens provides a medium level of sun glare reduction and UV protection with a visible light transmission of  $18 \sim 45\%$  [21]. We ensured that the brightness level of the smartphone was kept constant at a medium level throughout the entire experiment, and disabled brightness auto-adjustment to ensure consistency in the study setup. Furthermore, we counterbalanced the presentation order of the conditions across the participants. At the end of the experiment we conducted semi-structured interviews with each participant to understand their perceived performance during the completion of the tasks. The Human Ethics committee of our university approved this experiment.

#### 4 Results

To investigate the effect of ambient light on performance during smartphone interaction, we conducted a one-way repeated measures ANOVA on the aforementioned performance-measurement variables. We describe the results of our findings per each smartphone task. We removed extreme outliers from our data (3 individual data points in total from the whole dataset).

#### 4.1 Target Acquisition

First, we investigated the effect of ambient light on target acquisition time (milliseconds). The result of a one-way repeated measures ANOVA revealed a statistically significant effect of ambient light on target acquisition time (F(2,7607)=8.20,p<0.01). Post-hoc comparison using the Tukey HSD test (with Bonferroni corrections) showed that there is a significant difference (p=0.02) between target acquisition time under the dimmed light condition (M=495,SD=110) and the normal light condition (M=485,SD=103). Moreover, our results show that the participants took a significantly longer time tapping a target (p<0.01) while wearing sunglasses (M=498,SD=115) when compared to the normal light condition. However, there was no significant difference between the dimmed light condition and wearing sunglasses (p>0.05). Mean values for target acquisition time are presented in Figure 2 (a).

We then examined the effect of ambient light on touch accuracy. A one-way repeated measures ANOVA showed a statistically significant effect of ambient light on the participants' offset size (F(2,7607)=7.32,p<0.01). Post-hoc comparison using the Tukey HSD test (with Bonferroni corrections) indicated that the offset size was significantly larger under the dimmed light condition (M=49.50,SD=26.70,p=0.02) as compared to the normal light condition (M=47.70,SD=25.80). We also found a statistically

significant difference in offset size between the dimmed light and sunglasses conditions (M=46.7, SD=26.40, p<0.01). However, there was no significant difference between the offset size under normal light and sunglasses conditions. Mean values for offset size are presented in Figure 2 (b).



Fig. 2: Mean Target Acquisition Time and Offset Size (95% CI)

We also studied the effect of ambient light on effective throughput, calculated as proposed by Soukoreff & MacKenzie [20]. A one-way repeated measures ANOVA did not reveal a significant effect of ambient light on the effective throughput during target acquisition tasks.

#### 4.2 Visual Search

We examined the effect of ambient light on the time taken to memorise (milliseconds) and subsequently find an icon (milliseconds). The result of a one-way repeated measures ANOVA revealed a statistically significant effect of ambient light on the time taken to memorise an icon (F(2,2045)=4.42,p=0.01). Post-hoc comparisons using the Tukey HSD test (with Bonferroni corrections) indicated that participants took significantly longer time to memorise icons in the sunglasses condition (M=744,SD=271,p=0.01) than the dimmed light condition (M=703,SD=206). The mean values to memorise an icon are presented in Figure 3. However, we did not find a statistically significant difference between the normal light condition and the dimmed light condition (p>0.05) for the time taken to memorise an icon. We found similar results when comparing the wearing sunglasses condition to the normal light condition (p>0.05).

#### 4.3 Text Entry

In the text entry task we measured participants' performance in terms of time per character entry in milliseconds and total error rate [20]. We calculated character entry rate as time taken to input a character, while the total error rate was calculated as the ratio between the number of total errors and total entered characters. A one-way repeated measures ANOVA did not reveal a statistically significant effect of ambient light on either character entry rate or error rate (p > 0.05). We built two gerenalised linear mixed-effect models to describe the effect of ambient light on character entry rate and error rate. None of the predictors had a significant effect on text entry rate.

Z. Sarsenbayeva et al.

6



Fig. 3: Mean Time to Memorise an Icon (95% CI)

#### 4.4 Qualitative Results

During the interview, our participants commented on their subjective perception of performance during the smartphone tasks. A number of participants (8 out of 28) mentioned that wearing sunglasses and dimmed light conditions affected their target acquisition time: "A little longer when I'm wearing the sunglasses." (P02), "I was quicker when there was more light." (P12), "Longer time was required under dimmed light" (P06). These findings are in line with our quantitative findings which show a negative effect of dimmed light and wearing sunglasses on target acquisition time.

Regarding the self-perceived accuracy during the target acquisition task, half of our participants (14 out of 28) indicated that they were more accurate under the normal light condition as compared to the dimmed light condition. "In the dimmed light it was harder to accurately tap the center of circles" (P04). These findings correspond to our quantitative data that shows that under the dimmed light participants were less accurate. Surprisingly, two of our participants claimed they were more accurate under the dimmed light condition, as the contrast of the screen was brighter and they could see the circle clearer: "In the dim light I felt I was more accurate because the circles were more visible" (P19).

When we asked the participants about their perceived performance on the time taken to memorising the icon during the visual search task, a large majority of our participants (17 out of 24) stated that the light did not affect their performance. However, these perceptive statements contradict our quantitative results, showing that participants took a significantly longer time when wearing sunglasses as compared to the dimmed light condition. Interestingly enough, one of our participants mentioned that it took them less time to memorise icons under the dimmed light condition: "When performing in dimmed light it takes less time to memorise icons" (P15).

Nevertheless, participants reported a negative effect of ambient light on their perceived performance in time taken to find an icon, even though our quantitative analysis does not support this observation. In total, 10 participants mentioned that they believe it took them a longer time to find an icon under the dimmed light: *"It affects me so much. I took a bit longer to find the icons in a dim light condition"* (P18); *"When the light is on, I can find the icon easier compared to when the light is dimmed"* (P13).

Regarding the text entry task, most of the participants (N=19) claimed that the light did not have any effect on their performance. A total of 4 participants believed

that they were slower to type under the dimmed light condition and when wearing sunglasses. "It took me more time to type text under dim light" (P20). Furthermore, 4 participants claimed to make more errors when the lights were dimmed. "It was more difficult for me to type the text accurately with dim lighting. I was more confident in typing under normal lighting" (P12). Nonetheless, our quantitative results did not show any significant support for these statements.

#### 5 Discussion

#### 5.1 Impact of Ambient Light on Mobile Interaction

Our findings show that participants took a significantly longer time to tap a target while wearing sunglasses and under the dimmed light as compared to normal light condition. However, only a minority of our participants (8 out of 28) reported the negative impact of ambient light on target acquisition time. This is an indication that dimmed light caused situational visual impairments in our participant without them noticing it. Previous work has shown that various environmental and internal factors have a different effect on target acquisition time. For example, previous research has shown that participants took a significantly longer time to tap circles under cold ambience due to stiff muscles [17,18]. However, under music (fast and slow tempo) and urban noise (indoor and outdoor) conditions [16], and when exposed to stress [14], target acquisition time was significantly shorter due to the rhythm of the music, and the anxiety caused by urban noise and stress.

Furthermore, participants were significantly less accurate in target acquisition tasks under the dimmed light condition. This was confirmed in our qualitative data as the participants mentioned that they felt the negative effect of dimmed light on their interaction. In particular, our participants acknowledged that their perceived accuracy when tapping circles in dimmed environment is worse, compared to normal ambient illumination. This may be due to the fact that as the illuminance decreases, retinal dopaminergic activation from photoreceptors drops, and, hence causes a situational visual impairment [11]. In addition, our analysis did not reveal a significant effect of wearing sunglasses on participants' tap accuracy as compared to normal light condition. This might be due to the fact that we used commonly available non-polarised sunglasses that are unlikely to cause strong visual impairments.

However, we observed a negative effect of wearing sunglasses on memorising time in visual search tasks. Moreover, we anticipate that the effect of wearing sunglasses under bright sun light might be exacerbated as the effect of glare contributes to the magnitude of the visual situational impairment. Although ambient light did not have a negative impact on our participants' visual search time, the majority of our participants claimed that it took longer time to find an icon when the light was dimmed. However, this may be the case given the simple nature of the task, as prior research has shown that low illuminance levels are associated with slower search speed [6].

Finally, our analysis did not reveal a negative effect of dimmed light or wearing sunglasses on performance during text entry tasks. This may be due to the fact that we used a limited number of text entry tasks that are not sufficient to observe the effect of ambient light on text entry performance. However, previous work has shown a significant effect on participants exposed to meaningful speech on a similar typing task (i.e. listening to someone speak in a language they understand while typing on their smartphone) [16]. This confirms that different situational factors have a different effect on typical smartphone tasks.

#### 8 Z. Sarsenbayeva et al.

#### 5.2 Accommodating Ambient Light-Induced Situational Impairments

In summary, we demonstrate the negative impact of dimmed ambient light on fundamental smartphone interaction tasks. We argue that accounting for situational visual impairments in mobile interaction is important, as the effect might accumulate as task complexity rises. Moreover, Tigwell et al. [22] argue that the value of reducing the effects of situational visual impairments grows as the importance of the task increases. Previous work has proposed different methods to accommodate for situational visual impairments during mobile interaction. For example, in the study by Tigwell et al. [23] participants suggested increasing the contrast of the screen to reduce the effect of situational visual impairments. Moreover, Reinecke et al. [13] suggest increasing button sizes and adjusting the background colour to reduce the adverse effect of situational visual impairments on mobile interaction. As smartphones already come equipped with an ambient light sensor, these methods can be applied once a detrimental ambient light condition is detected and the user is performing a particular task (e.q., targetacquisition task under the dimmed light). Given that different people have differences in their perception of contrast colours, the ambient light sensor together with adaptive techniques (e.g., screen contrast, background colour) could be used to build personalised interfaces to improve the smartphone interaction experience, beyond simply adjusting the screen brightness as is the case with current devices.

#### 5.3 Limitations

We acknowledge several limitations in this study. First, the study settings were strictly controlled. In particular, we examined only two levels of ambient illumination – normal and the dimmed light, and do not investigate the effect of bright ambient light (outdoor illuminance) on smartphone interaction performance. The reason for this exclusion is to eliminate the effect of additional external factors, such as glare and ambient noise, on smartphone interaction performance. Finally, our experiment is limited to three types of smartphone tasks. We argue that these tasks are representative of the vast majority of activities that typical users undertake while using their smartphone.

#### 6 Conclusion

In this study, we investigate the effect of three ambient light conditions on smartphone interaction performance in target acquisition, visual search, and text entry tasks. We found that dimmed ambient light significantly impairs target acquisition. Participants took a significantly longer time to hit targets while wearing sunglasses or are under dimmed light, as compared to the normal light condition. Furthermore, participants were less accurate when tapping targets under the dimmed light condition. We also show that participants took longer to memorise icons while wearing sunglasses when completing visual search tasks. Our findings enhance the understanding of situational visual impairments impact on mobile interaction and contribute to the growing body of research in the HCI community on situational impairments.

#### 7 Acknowledgements

This work is partially funded by a Samsung GRO grant, and the ARC Discovery Project DP190102627. We also thank Silan Li for her help during data collection.

#### References

- Goel, M., Findlater, L., Wobbrock, J.: Walktype: Using accelerometer data to accomodate situational impairments in mobile touch screen text entry. In: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. pp. 2687–2696. CHI '12, ACM, New York, NY, USA (2012). https://doi.org/10.1145/2207676.2208662
- Goncalves, J., Sarsenbayeva, Z., van Berkel, N., Luo, C., Hosio, S., Risanen, S., Rintamäki, H., Kostakos, V.: Tapping task performance on smartphones in cold temperature. Interacting with Computers 29(3), 355–367 (2017). https://doi.org/10.1093/iwc/iww029
- Gong, R., Xu, H., Wang, B., Luo, M.R.: Image quality evaluation for smart-phone displays at lighting levels of indoor and outdoor conditions. Optical Engineering 51(8), 084001 (2012)
- Henze, N., Rukzio, E., Boll, S.: 100,000,000 taps: Analysis and improvement of touch performance in the large. In: Proceedings of the 13th International Conference on Human Computer Interaction with Mobile Devices and Services. pp. 133–142. MobileHCI '11, ACM, New York, NY, USA (2011). https://doi.org/10.1145/2037373.2037395
- Kim, Y.J., Luo, M.R., Choe, W., Kim, H.S., Park, S.O., Baek, Y., Rhodes, P., Lee, S., Kim, C.Y.: Factors affecting the psychophysical image quality evaluation of mobile phone displays: the case of transmissive liquid-crystal displays. JOSA A 25(9), 2215–2222 (2008)
- Lee, D.S., Shieh, K.K., Jeng, S.C., Shen, I.H.: Effect of character size and lighting on legibility of electronic papers. Displays 29(1), 10–17 (2008)
- Liu, P., Zafar, F., Badano, A.: The effect of ambient illumination on handheld display image quality. Journal of digital imaging 27(1), 12–18 (2014)
- Liu, Y., Goncalves, J., Ferreira, D., Hosio, S., Kostakos, V.: Identity crisis of ubicomp?: Mapping 15 years of the field's development and paradigm change. In: Proceedings of the 2014 ACM International Joint Conference on Pervasive and Ubiquitous Computing. pp. 75–86. UbiComp '14, ACM, New York, NY, USA (2014). https://doi.org/10.1145/2632048.2632086
- Liu, Y., Goncalves, J., Ferreira, D., Xiao, B., Hosio, S., Kostakos, V.: Chi 1994-2013: Mapping two decades of intellectual progress through co-word analysis. In: Proceedings of the 32nd Annual ACM Conference on Human Factors in Computing Systems. pp. 3553–3562. CHI '14, ACM, New York, NY, USA (2014)
- Ng, A., Brewster, S.A., Williamson, J.H.: Investigating the effects of encumbrance on one- and two- handed interactions with mobile devices. In: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems. pp. 1981–1990. CHI '14, ACM, New York, NY, USA (2014). https://doi.org/10.1145/2556288.2557312
- Norton, T.T., Siegwart Jr, J.T.: Light levels, refractive development, and myopia–a speculative review. Experimental eye research 114, 48–57 (2013)
- Observatory, T.N.O.A.: Recommended light levels (2015), https://www.noao.edu/ education/QLTkit/ACTIVITY\_Documents/Safety/LightLevels\_outdoor+indoor.pdf
- Reinecke, K., Flatla, D.R., Brooks, C.: Enabling designers to foresee which colors users cannot see. In: Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems. pp. 2693–2704. ACM (2016)
- Sarsenbayeva, Z., van Berkel, N., Hettiachchi, D., Jiang, W., Dingler, T., Velloso, E., Kostakos, V., Goncalves, J.: Measuring the effects of stress on mobile interaction. Proc. ACM Interact. Mob. Wearable Ubiquitous Technol. 3(1), 24:1–24:18 (Mar 2019). https://doi.org/10.1145/3314411
- Sarsenbayeva, Z., van Berkel, N., Luo, C., Kostakos, V., Goncalves, J.: Challenges of situational impairments during interaction with mobile devices. In: Proceedings of the 29th Australian Conference on Computer-Human Interaction. pp. 477–481. ACM (2017)
- Sarsenbayeva, Z., van Berkel, N., Velloso, E., Kostakos, V., Goncalves, J.: Effect of distinct ambient noise types on mobile interaction. Proc. ACM Interact. Mob. Wearable Ubiquitous Technol. 2(2), 82:1–82:23 (Jul 2018). https://doi.org/10.1145/3214285

- 10 Z. Sarsenbayeva et al.
- Sarsenbayeva, Z., van Berkel, N., Visuri, A., Rissanen, S., Rintamäki, H., Kostakos, V., Goncalves, J.: Sensing cold-induced situational impairments in mobile interaction using battery temperature. Proc. ACM Interact. Mob. Wearable Ubiquitous Technol. 1(3), 98:1–98:9 (Sep 2017). https://doi.org/10.1145/3130963
- Sarsenbayeva, Z., Goncalves, J., García, J., Klakegg, S., Rissanen, S., Rintamäki, H., Hannu, J., Kostakos, V.: Situational impairments to mobile interaction in cold environments. In: Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing. pp. 85–96. UbiComp '16, ACM, New York, NY, USA (2016). https://doi.org/10.1145/2971648.2971734
- Sears, A., Lin, M., Jacko, J., Xiao, Y.: When computers fade: Pervasive computing and situationally-induced impairments and disabilities. In: HCI International. vol. 2, pp. 1298–1302 (2003)
- Soukoreff, R.W., MacKenzie, I.S.: Metrics for text entry research: an evaluation of msd and kspc, and a new unified error metric. In: Proceedings of the SIGCHI conference on Human factors in computing systems. pp. 113–120. ACM (2003)
- Standard, A.Z.: Sunglasses and fashion spectacles (2003), https://www.saiglobal.com/ pdftemp/previews/osh/as/as1000/1000/1067.pdf
- 22. Tigwell, G.W., Flatla, D.R., Menzies, R.: It's not just the light: understanding the factors causing situational visual impairments during mobile interaction. In: Proceedings of the 10th Nordic Conference on Human-Computer Interaction. pp. 338–351. ACM (2018)
- Tigwell, G.W., Menzies, R., Flatla, D.R.: Designing for situational visual impairments: Supporting early-career designers of mobile content. In: Proceedings of the 2018 on Designing Interactive Systems Conference 2018. pp. 387–399. ACM (2018)
- Wobbrock, J.O.: The future of mobile device research in hci. In: CHI 2006 workshop proceedings: what is the next generation of human-computer interaction. pp. 131–134 (2006)

## Chapter 7

# **Contrasting the Effects of Different SIIDs on Mobile Interaction**

This chapter provides comparison of the effects of the SIIDs: ambient noise, stress, and dim ambient light on mobile interaction presented in Chapters 4, 5, and 6 respectively. Furthermore, this chapter highlights the importance of contrasting the effects of these SIIDs on mobile interaction.

Table 7.1 presents mean values for the variables that are used to quantify the effects of the SIIDs on mobile interaction performance: time taken to tap targets and the size of the touch offset in the target acquisition task, time taken to memorise and time taken to find a target icon in the visual search task, and, finally, time per character entry and total error rate in the text entry task. Furthermore, the table presents the magnitude of the impact of each of the SIIDs in percentage when compared to their respective baselines. This direct comparison of magnitudes is possible due to the fact that the protocol followed in the studies presented in Chapters 4, 5, and 6 was identical. Moreover, each baseline condition presented in Table 7.1 was controlled by excluding additional confounding factors that could possibly lead to decrease in performance during mobile interaction. In particular, we ensured that the baseline conditions for each of the studies presented in Chapters 4, 5, and 6 were silent, warm, stress-free, and illuminated with normal ambient light.

It is important to note that Table 7.1 was partially presented in the publication presented in Chapter 5; however, it did not include the data for the effects of dim ambient light. Hence, Table 7.1 in this chapter presents complete data on the effects of SIIDs investigated within the scope of this thesis. However, this chapter will mostly focus on and discuss the effects of SIIDs that were statistically significant on mobile interaction performance as was established in the studies presented in Chapters 4, 5, and 6.

### 7.1 Target Acquisition Task

From Table 7.1 we can observe that particular SIIDs have a more pronounced effect on mobile interaction performance during the target acquisition task. For example, when exposed to the cold environment and dim light, participants took longer time to tap on the targets as compared to the respective baselines, *e.g.*, warm environment and normal light. The effect of cold ambience can be explained by the fact that cold temperature decreases the dexterity of extremities, fingers in this case; hence, increases task completion time [24, 106]. According to Table 7.1, dim ambient light also slowed down the target access time in the target acquisition task. This effect can be explained by the reduced colour differentiation abilities due to dim light [220] as well as reduced photoreceptors' dopaminergic activation [189], resulting in prolonged target access time and reduced accuracy. Table 7.1 also demonstrates that under meaningful and meaningless speech conditions, target acquisition time was longer as compared to the baseline condition (silence). However, this effect was not statistically significant.

In contrast to cold ambience and dim light, under music conditions (both fast and slow), the target acquisition time was significantly lower than compared to the respective baseline (*i.e.*, silence). These findings are in line with literature, as it has been previously shown that music tempo influences people's task completion time [44, 162, 166, 167]. In particular, research has shown that under fast tempo music, people tend to complete their tasks, such as eating and drinking, faster [162, 166, 167]. Nevertheless, according to literature, slow tempo music is associated with slower task completion times [44]; however, our study demonstrates that when listening to music with both slow and fast tempo, participants were quicker to access targets as compared to the silent condition. We argue this was due to the fact that in our study we used the same musical composition – Bach's "Brandenburg Concerto No. 2" sampled at a lower rate for slow tempo music condition (60 bpm).

Furthermore, target acquisition time was faster when participants were experiencing and recovering from the induced stress as compared to no stress baseline. These findings were also in line with the existing literature [163], as it has previously been shown that due to stress, people experience anxiety [128] and rush through their tasks [163]. Table 7.1 also shows that the effect of stress was the most profound one on target access time compared to other SIIDs presented in the table. This implies that when designing adaptive interfaces to accommodate the effects of SIIDs on mobile interaction performance, from the ones presented in Table 7.1, stress should be given a higher priority as its' effect on target acquisition task is more severe. Interestingly, regardless of the SIID type, we can observe in Table 7.1 that when the target acquisition time decreases, the size of the offset grows; hence, the accuracy of the touch drops. This can be explained by the fact that when the participants favour speed of target acquisition task, they neglect the accuracy [150, 249]. In case of cold ambience, while the target acquisition time increases, the accuracy drops due to the loss of dexterity in fingers [24, 106].

### 7.2 Visual Search Task

The visual search task had two components that were quantified within the scope of this thesis: target icon memorisation and target icon search. The effects of SIIDs on this mobile task was not as diverse as on the target acquisition task. From Table 7.1 it can be observed that SIIDs had a significant effect on target memorisation time, but not the target visual search time.

For example, in the study presented in [231], cold ambience had a significant negative effect on target memorisation time as compared to the warm environment. This might be due to the fact that cold ambience can negatively affect cognitive abilities of people [61]; however, the exposure to cold needs to be prolonged (at least 45 min [61]). This can partially explain why the authors did not observe negative effect of cold ambience on visual search time [231].

Moreover, in the study presented in Chapter 4 and as demonstrated in Table 7.1, we observed that under urban indoor noise, icon memorisation time was significantly lower as compared to the memorisation time in the baseline condition. This might be due to the fact that urban noise was not pleasant for the participants to experience; hence, they tried to "escape" this condition and rushed through the task (as mentioned in the qualitative data). Similarly, in the study presented in Chapter 5, memorisation time was significantly shorter when participants were experiencing and recovering the induced stress. We argue that the reason behind such behaviour is again anxiety [128] that has shown to prompt people to rush through the tasks [163].

Finally, according to Table 7.1, it can be observed that wearing sunglasses had a negative effect on icon memorisation time, but not on icon search time. Similarly, there was no significant effect of ambient noise and stress on visual search time. This might be because the visual search task was not complex enough to observe stronger effects of these SIIDs and further research is needed to fully understand its impact on mobile interaction.

### 7.3 Text Entry Task

From Table 7.1 it can be observed that the effect of outdoor urban noise had a negative influence on participants' performance during the text entry task as demonstrated in the study presented in Chapter 4. In particular, the participants took significantly longer time per character entry when exposed to urban outdoor noise condition as compared to the silence. This might be because urban outdoor noise is distracting in nature [19], and, hence, added to the cognitive load of the participants who needed to concentrate to complete the text entry task [18]. As a result, participants took longer time to complete the text entry task when being exposed to urban outdoor noise condition.

Furthermore, Chapter 4 demonstrates the negative effect of meaningful speech on participants' performance during the text entry task: time per character entry was significantly longer under the meaningful speech condition as compared to the silent condition. Such behaviour can be explained by the distracting nature of meaningful speech that also increases people's cognitive load [261]. As text entry task also requires concentration and occupies participants' cognitive load, presence of meaningful speech adds to the cognitive load and deteriorates performance during the text entry task. Interestingly, this effect was not observed in the meaningless speech condition on the text entry task possibly due to the fact that it was easier for participants to ignore the meaningless speech, hence decrease its' effect on cognitive load, as they did not understand it. Finally, the effect of both outdoor urban noise and meaningful speech was not observed on the error rate during text entry. This might be due to the limited number of text entry messages that did not let us observe statistical significance of these SIIDs on the text entry task.

Finally, the effects of dim ambient light and the effects of stress on the text entry task were not statistically significant in the studies presented in Chapters 5 and 6 neither in terms of time per character entry nor error rate. This might be due to several reasons. First, the number of text entry messages within the task was limited and, perhaps, was not sufficient enough to observe the effect of these SIIDs (stress and dim ambient light) on text entry performance. Second, the SIIDs were strictly controlled according to the protocol. It is possible that under naturalistic conditions, the effects of both stress and dim ambient light might be stronger than the ones presented in our studies. It is possible that under stronger levels of stress and dim ambient light while completing more complex text entry tasks, the effects of these SIIDs might be more profound and remain to be investigated in future work. Overall, from Table 7.1 we can observe that meaningful speech has a greater detrimental effect on the text entry task performance; hence, meaningful speech should be given higher priority to accommodate mobile interaction during text entry in circumstances with multiple SIIDs being detected.

In addition, the effects of cold ambience on the text entry task is not presented in Table 7.1, as the authors of [231] did not investigate and did not report this data in their study. Hence, it is not applicable to review the effects of cold ambience on text entry performance within the scope of this thesis. Future work could further investigate the effects of different ambient temperatures on text entry performance.

## 7.4 Implications for Design and Summary

Our findings demonstrate that different SIIDs have different effects on mobile interaction tasks. This knowledge could be considered when designing detection mechanisms or adaptive user interfaces for smartphones and other ubiquitous technology that require mobile interaction in presence of SIIDs. For example, to reduce the effects of cold, stress, and ambient noise on target acquisition task performance, mobile interfaces could increase the size of the targets or actuate input techniques directed to improve accuracy, presented in literature [26,37,116].

Moreover, as smartphones and other ubiquitous devices come equipped with a number of sensors, both contextual, sensor, and input data can be used to detect the presence of SIIDs. For example, built-in microphone of the device could be used to detect noise-induced SIID, while built-in ambient light sensor could be used to detect ambient light-induced SIIDs.

Furthermore, our findings could be used to suggest or avoid using certain input techniques when particular SIIDs are detected. For instance, the interface could avoid using voice assistance in the presence of ambient noise; however, it could be a preferred input technique when cold-induced SIIDs are sensed. In addition, the interface could potentially warn the user about possible frostbite risks if long exposures to cold has been recorded by the device [231]. Similarly, to decrease the effect of meaningful speech on the text entry task, the interface could launch adaptive keyboard to improve user experience and minimise errors [70].

To summarise, these comparisons enhance our understanding of the effects of different SIIDs on mobile interaction, and therefore, accumulate knowledge on SIIDs research and contribute to the research agenda. This knowledge can also assist researchers and designers in developing mechanisms for accommodating SIIDs when a combination of them is present and when particular tasks are being completed.

Table 7.1:	Comparison	of the Effects	of SIIDs aga	inst Respective	e Baseline
				City ation al Im	

Baseline				Situational Impairments											
Mean (SD)	Warm	Silent	No stress	Normal Light	Cold	Music Fast	Music Slow	Urban Indoor	Urban Outdoor	Speech Meaningful	Speech Meaningless	Stress	Post-Stress Recovery	Dim Light	Wearing Sunglasses
Time to tap	593.00	591.84	538.68	485.00	603.00 *	577.76 *	574.15 *	583.56 *	573.56 *	592.81	609.38	534.77 *	520.18 *	495.00 *	498.00 *
a circle, ms	(137.89)	(125.07)	(127.58)	(103.00)	+ 1.7%	- 2.4%	- 3.0%	- 1.4%	- 3.1%	+ 0.2%	+ 3.0%	- 0.7%	- 3.4%	+2.27%	+2.89%
Offset	41.34	39.38	46.47	47.70	42.66 *	42.95	42.11 *	41.06	40.91	42.90	39.47	49.14 *	50.20 *	49.50 *	46.70
size, px	(40.89)	(35.26)	(25.57)	(25.80)	+ 3.1%	+9.1%	+ 6.9%	+ 4.3%	+ 3.9%	+ 8.9%	+ 0.3%	+ 5.7%	+ 8.0%	+3.77%	-2.09%
Time to memorise an icon, ms	815.00 (150.15)	748.76 (253.89)	800.51 (331.15)	719.00 (290.00)	854.00 * + 4.8%	737.71 - 1.5%	745.15 - 0.5%	712.11 * - 4.9%	743.56 - 0.7%	753.20 + 0.6%	738.90 - 1.3%	714.52 * - 4.6%	702.31 * - 12.3%	703.00 -2.25%	744.00 +3.47%
Time to find an icon, ms	1632.24 (1235.27)	1587.74 (871.95)	1602.45 (1315.82)	1499.00 (827.03)	1942.46 + 19.0%	1564.70 - 1.4%	1753.19 + 10.4%	1543.12 - 2.8%	1633.15 + 2.9%	1637.54 + 3.1%	1520.99 - 4.2%	1496.50 - 6.6%	1506.38 - 6.0%	1477.07 -1.46%	1513.00 +0.93%
Character entry rate, ms/char	n/a	454.39 (219.95)	560.88 (173.96)	532.20 (121.00)	n/a	488.11 + 7.4%	485.14 +6.8%	480.96 + 5.8%	523.89 * + 15.3%	539.82 * + 18.8%	508.99 + 12%	547.16 - 2.4%	495.83 - 11.6%	519.01 -2.48%	527.00 -0.94%
Total error rate, %	n/a	7.54 (7.72)	7.46 (6.63)	5.55 (4.81)	n/a	7.45 - 1.2%	7.38 - 2.1%	8.54 + 13.3%	8.22 + 9.0%	7.09 - 6.0%	7.07 - 6.2%	10.50 + 40.8%	9.40 + 26.0%	4.94 -10.99%	5.06 -8.83%

\* indicates statistically significant effect

## **Chapter 8**

## Sensing Cold-Induced SIIDs

Previous work has already quantified the effects of cold environments on smartphone interaction [76, 231]. Cold has been shown to have a negative effect on fine-motor performance during mobile interaction due to decreased dexterity and stiffness of the muscles [76]. In particular, participants took significantly longer time and were less accurate when completing target acquisition tasks under cold ambience when compared to a warm environment. The study demonstrates the importance of accounting for cold-induced SIIDs that might impair mobile interaction [231].

This chapter presents our work where we propose a sensing mechanism to detect cold-induced situational impairments using an off-the-shelf smartphone (RQ3). Current smartphones are equipped with different sensors that can be used to detect the presence of different situational impairments. For example, the ambient light sensor can be used to detect ambient light-induced SIIDs, while ambient noise-induced situational impairments can be detected with the help of the device's built-in microphone. Internal SIIDs, such as stress, can be detected via wearable devices. For example, physiological data collected from wearable sensors – HRV and EDA [161] – can indicate the presence of stress. Nevertheless, despite the established adverse effect of cold-induced situational impairments [231] on mobile interaction, the researchers did not propose a sensing mechanism to detect them.

This is particularly challenging, as there are several factors (*e.g.*, context, location, interaction duration) that need to be accounted for when designing a sensing mechanism to detect cold-induced SIIDs. For example, a general weather forecast cannot be used for this purpose as while indoor/outdoor detection mechanisms for smartphones exist [7], exposure to cold conditions may vary for a myriad of reasons (*e.g.*, the user is wearing gloves or is located in an somewhat environmentally shielded outdoor location).

Our work suggests using the smartphone's built-in battery temperature sensor to infer changes in users' finger temperature. This approach is fairly reliable as it demonstrates a similar trend in behaviour between the smartphone battery temperature, and users' finger temperature in cold and warm environments: decreases in the cold ambience, and increases in the warm environment. In particular, we found a high correlation between these two variables (r = 0.86). Our findings contribute towards building sensing mechanisms to detect cold-induced SIIDs using an off-the-shelf smartphone. The details of our approach can be found in the attached publication in Section 8.1.

### 8.1 Publication

©Copyright held by the authors. Publication rights licensed to ACM. This is the authors' version of the work. It is posted here for your personal use. Not for redistribution. The definitive Version of Record was published in:

Zhanna Sarsenbayeva, Niels van Berkel, Aku Visuri, Sirkka Rissanen, Hannu Rintamaki, Vassilis Kostakos, and Jorge Goncalves. 2017. Sensing Cold-Induced Situational Impairments in Mobile Interaction Using Battery Temperature. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 1, 3, Article 98 (September 2017), 9 pages. DOI: https://doi.org/10.1145/3130963

Ethics for this study was obtained from the University of Oulu (Finland) under local ethics advisory group.

# Sensing Cold-Induced Situational Impairments in Mobile Interaction Using Battery Temperature

ZHANNA SARSENBAYEVA, The University of Melbourne NIELS VAN BERKEL, The University of Melbourne AKU VISURI, University of Oulu SIRKKA RISSANEN, Finnish Institute of Occupational Health HANNU RINTAMAKI, Finnish Institute of Occupational Health VASSILIS KOSTAKOS, The University of Melbourne JORGE GONCALVES, The University of Melbourne

Previous work has highlighted the detrimental effect of cold ambience on fine-motor skills during interaction with mobile devices. In this work, we develop a method to infer changes in finger temperature of smartphone users without the need for specialised hardware. Specifically, we demonstrate that smartphone battery temperature is a reliable gauge for determining changes to finger temperature. In addition, we show that the behaviour of smartphone battery temperature in cold settings is consistent across different smartphone models and battery configurations. Our method can be used to determine cold-induced situational impairments, and trigger interface adaptations during mobile interaction.

CCS Concepts: • Human-centered computing  $\rightarrow$  Empirical studies in HCI; • Human-centered computing  $\rightarrow$  Ubiquitous and mobile computing; • Human-centered computing  $\rightarrow$  Smartphones

#### **KEYWORDS**

Smartphones, ambient temperature, battery temperature, situational impairments, finger temperature, cold chamber.

#### **ACM Reference format:**

Zhanna Sarsenbayeva, Niels van Berkel, Aku Visuri, Sirkka Rissanen, Hannu Rintamaki, Vassilis Kostakos, Jorge Goncalves. 2017. Sensing Cold-Induced Situational Impairments in Mobile Interaction Using Battery Temperature. *PACM Interact. Mob. Wearable Ubiquitous Technol.* 1, 3, Article 98 (September 2017), 9 pages. DOI: 10.1145/3130963

This work is partially funded by the Academy of Finland (Grants 276786-AWARE, 286386-CPDSS, 285459-iSCIENCE, 304925-CARE), the European Commission (Grant, 6AIKA-A71143-AKAI), and Marie Skłodowska-Curie Actions (645706-GRAGE).

Author's addresses: Z. Sarsenbayeva, The University of Melbourne, School of Computing and Information Systems, Parkville 3010, Melbourne, Australia; email: zsarsenbayev@student.unimelb.edu.au; N. van Berkel, The University of Melbourne, School of Computing and Information Systems, Parkville 3010, Melbourne, Australia; email: n.vanberkel@student.unimelb.edu.au; A. Visuri, University of Oulu, Center for Ubiquitous Computing, Pentti Kaiteran katu 1, PO Box 4500, FI-90014 Oulu, Finland; email: aku.visuri@oulu.fi; S. Rissanen, Finnish Institute of Occupational Health, PL 40, 00251 Helsinki, Finland; email: sirkka.rissanen@ttl.fi; H. Rintamaki, Finnish Institute of Occupational Health, PL 40, 00251 Helsinki, Finland; email: hannu.rintamaki@ttl.fi; V. Kostakos, The University of Melbourne, School of Computing and Information Systems, Parkville 3010, Melbourne, Australia; email: vassilis.kostakos@unimelb.edu.au; J. Goncalves, The University of Melbourne, School of Computing and Information Systems, Parkville 3010, Melbourne, Australia; email: jorge.goncalves@unimelb.edu.au

Permission to make digital or hard copies of part or all of this work for personal or classroom use is granted without fee provided that copies are not made or distributed for profit or commercial advantage and that copies bear this notice and the full citation on the first page. Copyrights for components of this work owned by others than ACM must be honored. To copy otherwise, distribute, republish, or post, requires prior specific permission and/or a fee. Request permissions from permissions@acm.org.

<sup>2474-9567/2017/</sup>September- 98 \$15.00

Copyright is held by the owner/author(s). Publication rights licensed to ACM. https://doi.org/10.1145/3130963

98:2 • Z. Sarsenbayeva et al.

#### **1 INTRODUCTION**

Cold ambience is known to affect fine-motor skills during interaction with mobile devices. This is due to the fine-motor dexterity loss caused by drops in finger temperature [6,13]. However, while previous work has shown that finger temperature is an important consideration when designing mobile phone applications, no reliable way of inferring finger temperature in situ has been proposed without the need for external measurement devices (*e.g.*, skin thermistors).

Future smartphones may come with embedded environmental temperature sensor to gauge ambient temperature, but this is still not the case today. In our work, we make creative use of existing smartphone sensors to gauge finger temperature. Specifically, we investigate if finger temperature can be inferred from the smartphone's battery temperature. The literature contains a growing list of examples where smartphone sensors originally intended for a particular function can actually be appropriate for a wholly different use case. For example, researchers have shown how to transmit data using the magnetometer [9], or determine social context through Bluetooth traces [2].

Here we demonstrate that it is possible to leverage data on smartphone battery temperature to infer fluctuations of a user's finger temperature during interaction. We show a significant correlation between battery and finger temperature, and as such one can be used to infer the other. Our method can infer changes in users' finger temperature using off-the-shelf smartphones without the need for additional sensors or hardware.

#### 2 RELATED WORK

#### 2.1 Environmental Sensing Using Smartphones

Previous work has described the use of smartphone sensors to better understand surrounding weather conditions. For instance, Mass & Madaus [10] discuss the use of smartphone pressure sensors for surface pressure observations. They describe that a fine-grained network of smartphone users would allow for high-resolution weather prediction as pressure readings are "not influenced by characteristics of the underlying surface, as are temperature and wind" [10], and therefore readings are not influenced by different contexts (e.g., shade, urbanisation).

Overeem *et al.* [12] also suggest that smartphones can be used as an effective instrument to study environmental and climate changes. Sensor data such as air humidity, air pressure, and air temperature can be used for data assimilation in weather prediction models, water management, and urban planning [12]. More relevant to our work, they report the analysis of a six-month dataset of 2.1 million battery temperature readings from eight major cities collected using the OpenSignal<sup>1</sup> Android application. They found that daily battery temperature was strongly correlated with the observed daily air temperature (r = 0.82) [12]. In addition, literature shows a substantial impact of ambient temperature on human's body temperature, particularly for extremities (*e.g.* fingers, toes) [11]. These findings provide us with strong motivation towards investigating the possibility of inferring changes in user finger temperature based on changes in smartphone battery temperature.

#### 2.2 Extending Smartphone's Capabilities

The literature also contains examples on the use of smartphones in combination with specialised devices for environmental sensing. A combination of general smartphone sensors and purpose-made (add-on) sensors allows for the collection of sensor data not feasible with a standalone smartphone. By distributing 8000 small add-ons for regular smartphones among the Dutch population, Snik *et al.* [14] deployed a 'citizen science' experiment to carry out aerosol measurements of high temporal resolution. "*The optical design of iSPEX uses the smartphone camera as the detector, and the iSPEX add-on produces a spectrum of the light that entered the slit* [...]" [14].

<sup>&</sup>lt;sup>1</sup> http://opensignal.com

PACM Interact. Mob. Wearable Ubiquitous Technol., Vol. 1, No. 3, Article 98. Publication date: September 2017.

In a different project, researchers developed an attachment for smartphones to crowdsource the collection of pollution data [8]. Combining this data with the GPS sensor of the mobile phone allowed for a more detailed understanding of urban air pollution.

Aram *et al.* [1] monitored changes in temperature and humidity using a Bluetooth-based acquisition system. The system consisted of device with a built-in temperature and humidity sensor, and a microcontroller wirelessly transmitting the climatic parameters to a receiver via Bluetooth. The device was left in a climatic chamber to monitor the temperature drop from 25 °C down to -20 °C followed by a temperature rise back to 25 °C, as well as a constant humidity level. The results show that temperature values obtained by the system were consistent with the temperature values inside the climatic chamber, but not for humidity values. They suggest that their approach might be useful to observe climate conditions for small environments, such as laboratories, home rooms, or medical spaces, and to trigger alarms when these conditions change [1].

#### 2.3 Impact of Situational Impairments on Smartphone Use

Previous work suggests that smartphone interfaces should be adapted when used under various situational impairments, such as dynamic state of the phone [5], ambient light and noise [16], movement [4], and cold ambience [6,13]. A number of different solutions have been developed to overcome these situational impairments when interacting with a mobile device. For example, Goel *et al.* [4] suggest adapting the keyboard interface when walking is detected to improve typing experience. This adaptive interface reduced errors and increased typing speed. Moreover, Wobbrock [16] argues that a better understanding of situational impairments can in addition contribute to improved accessibility and adaptive user interfaces. Furthermore, he considers the possibility that solutions designed for people with physical impairments can be applied to those with situational impairments during mobile interaction (*e.g.*, finger arthritis and people with cold fingers [16]).

On the topic of situational impairments caused by cold temperatures, previous work has shown that user performance when completing tapping tasks on a smartphone is affected by finger temperature. As a result of this finding, the researchers suggest adding finger temperature as a parameter in Fitts' Law modelling [6]. Furthermore, Sarsenbayeva *et al.* [13] show that precision and quickness are adversely affected by cold temperatures. While the researchers provide design suggestions for mobile device interface adaptation in cold environments, they do not describe how user exposure to cold ambience can be detected in a naturalistic setting (*i.e.*, without temperature sensors attached to the user's fingers). Hence, we extend previous work by proposing a method to determine changes in finger temperature by considering the smartphones' battery temperature.

#### 3 STUDY

We conducted two experiments to 1) assess the effect of a cold environment on the battery temperature of several smartphone models, and 2) investigate the relationship between smartphone battery temperature and human finger temperature.

#### 3.1 Experiment 1: Device Comparison

We conducted a comparative study of changes in battery temperature over time using multiple handsets. Our objective was to investigate whether different phones and batteries behave similarly in cold settings. We considered four different smartphones that vary in manufacturer, size, battery capacity, and other factors (see Table 1). Given the wide range of characteristics in our selected phones, we argue that they provide a good representation of existing models in the market.

#### 98:4 • Z. Sarsenbayeva et al.

	Motorola Moto G3	Samsung Galaxy S4 Mini	Huawei Nexus 6P	LG Nexus 5X
Dimensions	142.1 x 72.4 x	124.6 x 61.3 x	159.3 x 77.8 x	147.0 x 72.6 x 7.9
(H x D x W)	11.6 mm	8.49 mm	7.3 mm	mm
Weight	155 g	107 g	178 g	136 g
Screen size	5.0"	4.3"	5.7"	5.2"
Body material	Polycarbonate	Polycarbonate	Metal	Polycarbonate
Battery type	Li-Ion	Li-Ion	Li-Po	Li-Po
Battery capacity	2470 mAh	1900 mAh	3450 mAh	2700 mAh
Removable battery	No	Yes	No	No

Table 1. Specifications of smartphones used in our study.

The experiment took place in a medical testing facility, in two separate rooms with independent climate controls. As previous work identified the effect of cold ambience on performance during mobile interaction [13], we replicated the protocol to determine if under the same conditions we could use battery temperature to infer finger temperature. Namely, we set the cold room temperature to -10 °C, with wind velocity below 0.1 m/s and 70-75% humidity, while the warm room was set to 20 °C. All four smartphones were rotated between these two spaces to measure fluctuations in battery temperature: 14 minutes in cold temperature, 14 minutes in room temperature, 14 minutes in the cold temperature, and again 14 minutes in room temperature. We developed a custom Android application to record smartphone battery temperature for each device every 30 seconds during the experiment. Due to its power saving features, Android only allows polling of battery related data when an event is registered by the device's battery. Therefore, we ensured the triggering of an event on smartphones' battery every 30 seconds by plugging and unplugging the devices into an electrical outlet to collect battery temperature measurements.

*3.1.1 Results.* Fig. 1 shows the battery temperature fluctuation for each smartphone used in Experiment 1. Vertical lines represent the start of the experiment in each of the rooms. We calculated the Pearson product-moment correlation coefficient between each device to assess the relationship between the measured battery temperatures. A statistically significant positive correlation was observed between the battery temperatures of all four devices (Table 2).



Fig. 1. Smartphone battery temperature change.

	Motorola	Samsung	Huawei	LG	
	Moto G3	Galaxy S4 Mini	Nexus 6P	Nexus 5X	
Motorola Moto G3	-	0.96	0.98	0.92	
Samsung Galaxy S	<b>54</b> 0.96	-	0.99	0.88	
Mini					
Huawei Nexus 6P	0.98	0.99	-	0.91	
LG Nexus 5X	0.92	0.88	0.91	-	

Table 2. Pearson correlation coefficients between battery temperatures of the used smartphones.

#### 3.2 Experiment 2: Finger & Battery Temperature

In this experiment, we investigate if smartphone battery temperature and human finger temperature co-vary when exposed to changing ambient temperature. The experimental design was adopted from previous work [13]. We recruited 24 participants for our experiment through social media and mailing lists. We controlled for gender by having an equal distribution of males and females. All participants had lived in cold climates (*e.g.*, Scandinavia) for more than six months and owned a smartphone for at least one year. We also controlled participants' clothing by instructing them to wear a single layer of trousers and top garments on the day of the study and providing them with additional winter attire.

After participants were briefed on the purpose of the study, we asked them to sign a consent form if they agreed to the study's specifications. Participants were then instrumented with thermal sensors attached below the nails of their index finger and thumb on the back of their dominant hand. Thermal data was logged every 1 second using a mobile battery-powered Grant Squirrel meter/logger series 1000 (Fig. 2), allowing us to measure finger temperature, as described in [6,13]. We chose to only instrument these two fingers as they are the ones most likely to be used during one-handed or two-handed interaction with a smartphone. Participants did not wear gloves during the experiment and were asked to not place their hands inside of their pockets. The experiment took place in the same medical facility and rooms as Experiment 1, using the same room temperatures. Participants were asked to wear additional winter attire provided by us when exposed to the cold chamber (Fig. 2). Following the design of Sarsenbayeva et al. [13], participants alternated between the two rooms (cold-warm-cold-warm), and we recorded both battery and finger temperature every 30 seconds. During the study, participants were asked to use a provided smartphone to complete target acquisition tasks for 4 minutes in 3 different instances (at 1:00, 6:00 and 11:00, with a 1-minute break in between) for each condition (cold/index, cold/thumb, warm/index, warm/thumb). The hand posture was randomised and counterbalanced. We avoided using colder temperatures in our study (i.e., below -10 °C) to ensure the wellbeing of the participants. The experimental design was approved by the ethics committee of our university.

3.2.1 Results. A Pearson product-moment correlation coefficient was calculated to assess the relationship between participants' finger temperatures (index and thumb) and the phone's battery temperature. A positive correlation was observed for both fingers (index: r = 0.86, p < 0.01; thumb: r = 0.85, p < 0.01). Fig. 3 shows the changes in finger temperature and the smartphone's battery temperature throughout the experiment for each participant. Furthermore, our results show that participants' finger temperature *decreased* on average 0.73 °C per minute in the cold room, and *increased* on average 0.79 °C per minute in the warm room.

#### 98:6 • Z. Sarsenbayeva et al.



Fig. 2. Left: Grant Squirrel meter/logger (series 1000) and a participant's hand with thermal sensors attached. Right: Participant in the cold chamber and in the warm room.



---- Battery temperature (°C) - - Index finger temperature (°C) - Thumb temperature (°C)

Fig. 3. Participants' finger temperature and smartphone battery temperature.

#### 4 DISCUSSION

#### 4.1 Inferring Cold-Induced Situational Impairments

Previous work has shown that ambient temperature is highly correlated with smartphone battery temperature [1,12], and that ambient temperature has a significant impact on body temperature, particularly the extremities (*e.g.* fingers) [11]. In our work, we investigate the missing link regarding the relationship between finger temperature and smartphone battery temperature (Fig. 4). Our results demonstrate that smartphone battery

temperature is highly correlated with users' finger temperature (index: r = 0.86, thumb: r = 0.85) and can therefore be used to infer potential cold-induced situational impairments during mobile interaction for both interaction modes (one-handed and two-handed).

Our work demonstrates that changes in smartphone battery temperature along with other factors (*e.g.*, user input pattern) can be leveraged to signal the need to adapt the smartphone for cold environment. In addition, we show that this can be achieved without adding any additional sensors to the smartphone. Our approach of detecting a situational impairment corresponds to the method used by Goel *et al.* [4], where an accelerometer (a general-purpose sensor on most smartphones) was used to detect and reduce situational impairments induced by walking.

While previous work was able to achieve high accuracy when measuring device surface and screen temperature through additional sensors [3], an approach with add-on sensors aimed at measuring finger temperature is currently not feasible in real world situations. This is a challenge for researchers or interface designers who wish to adapt the interaction and interface when cold-induced situational impairments are detected. Our method shows that off-the-shelf smartphones can be used to detect users' finger temperature, thus providing a practical way to detect and accommodate for cold-induced situational impairments.



Fig. 4. Smartphone battery temperature relation to ambient temperature and finger temperature.

#### 4.2 Method Applicability

Previous work has argued that cold environments cause significant deterioration of fine-motor performance when interacting with the mobile device [6,13]. To improve fine-motor performance during mobile interaction, it is important for the device to be aware of the changes in a user's finger temperature and adapt its interface accordingly. Here we present an unobtrusive method to detect these cold-induced situational impairments during mobile interaction.

In practice, our method can be used as follows. When a user initiates interaction with the device, the device registers its current battery temperature. Subsequent decreases in battery temperature *while* the user is still interacting with the phone can then be used to infer drops in finger temperature. Once a certain threshold of finger temperature decrease has occurred, the smartphone interface should then adapt accordingly or provide warnings to the user. It is important to note that for our method to work, finger temperature inference can only occur *during* interaction. When not using their phone, users might put their hands in their pockets or warm up their hands through other means.

Finally, previous work has shown that a decrease of just a few degrees Celsius can affect mobile interaction performance [6]. Here, we show that participants' finger temperature dropped on average 0.73 °C per minute under our cold exposure conditions. However, at temperatures lower than -10 °C, the cooling rate of the fingers would be accelerated. This means that decreases in mobile interaction performance would occur in increasingly shorter spans of time.

#### 98:8 • Z. Sarsenbayeva et al.

#### 4.3 Limitations

This study has several limitations. While the choice for a constant room temperature is well motivated (temperature hardly changes during a typical session of smartphone usage [15]), we were limited in the actual temperature level imposed on our participants due to safety concerns. While outdoor temperatures in some parts of the world reach much lower temperatures than -10 °C, we decided not to study more extreme scenarios. In addition, we have not been able to test a wider range of smartphone models due to both time and financial constraints. However, the range of smartphones used in our study (Table 1) does show a considerable level of diversity. This, combined with the high correlation in our results, provides us with sufficient confidence that our results hold true for a much larger selection of smartphone devices.

Further, participants were instructed to not wear gloves or warm their hand by any other means, unlike in naturalistic settings where they could wear touchscreen gloves to interact with their devices or warm their hands inside of their pockets. This was by design as we wanted to: 1) avoid touch inaccuracies during mobile interaction, and 2) observe steady finger temperature drop.

Finally, we did not run compute-intensive applications on the devices during our experiment as we chose to focus on more typical interactions that are more likely to occur in outdoor cold environments.

#### 5 CONCLUSION

In this study, we demonstrate that changes in smartphone battery temperature can be used to infer changes in users' finger temperature. By doing so we also filled an important gap in the literature. While previous work has identified a relationship between ambient temperature and finger temperature, and between ambient temperature and battery temperature, our work is the first to establish a relationship between finger temperature and battery temperature. This is an important finding as it shows that cold-induced situational impairments can be predicted using off-the-shelf smartphones. This information can then be used to adapt mobile interfaces to overcome cold-induced situational impairments or simply to provide warnings to the user on over-exposure to cold environments.

#### REFERENCES

- S. Aram, A. Troiano and E. Pasero. 2012. Environment sensing using smartphone. IEEE Sensors Applications Symposium, 1-4. http://dx.doi.org/10.1109/SAS.2012.6166275.
- [2] Trinh M. T. Do and Daniel Gatica-Perez. 2011. GroupUs: Smartphone Proximity Data and Human Interaction Type Mining. In Proceedings of the 2011 15th Annual International Symposium on Wearable Computers, IEEE Computer Society, 21-28. http://dx.doi.org/10.1109/ISWC.2011.28.
- [3] Begum Egilmez, Gokhan Memik, Seda Ogrenci-Memik and Oguz Ergin. 2015. User-specific Skin Temperature-aware DVFS for Smartphones. In Proceedings of the 2015 Design, Automation & Test in Europe Conference & Exhibition, EDA Consortium, 1217-1220.
- [4] Mayank Goel, Leah Findlater and Jacob Wobbrock. 2012. WalkType: using accelerometer data to accomodate situational impairments in mobile touch screen text entry. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, 2687-2696. http://dx.doi.org/10.1145/2207676.2208662.
- [5] Mayank Goel, Jacob Wobbrock and Shwetak Patel. 2012. GripSense: Using Built-in Sensors to Detect Hand Posture and Pressure on Commodity Mobile Phones. In Proceedings of the 25th Annual ACM Symposium on User Interface Software and Technology, ACM, 545-554. http://dx.doi.org/10.1145/2380116.2380184.
- [6] Jorge Goncalves, Zhanna Sarsenbayeva, Niels van Berkel, Chu Luo, Simo Hosio, Sirkka Rissanen, Hannu Rintamäki and Vassilis Kostakos. 2016. Tapping Task Performance on Smartphones in Cold Temperature. *Interacting with Computers*, 29, 3: 355-367.
- [7] G. Havenith, R. Heus and H. A. Daanen. 1995. The hand in the cold, performance and risk. Arctic Med Res 54 Suppl 2: 37-47.
- [8] Richard Honicky, Eric A. Brewer, Eric Paulos and Richard White. 2008. N-smarts: Networked Suite of Mobile Atmospheric Realtime Sensors. In Proceedings of the Second ACM SIGCOMM Workshop on Networked Systems for Developing Regions, ACM, 25-30. http://dx.doi.org/10.1145/1397705.1397713.
- [9] Weiwei Jiang, Denzil Ferreira, Jani Ylioja, Jorge Goncalves and Vassilis Kostakos. 2014. Pulse: Low bitrate wireless magnetic communication for smartphones. In *International Joint Conference on Pervasive and Ubiquitous Computing*, ACM, 261-265. http://dx.doi.org/10.1145/2632048.2632094.
- [10] Clifford F. Mass and Luke E. Madaus. 2014. Surface Pressure Observations from Smartphones: A Potential Revolution for High-Resolution Weather Prediction? *Bulletin of the American Meteorological Society* 95, 9: 1343-1349. http://dx.doi.org/10.1175/BAMS-D-13-00188.1.
- [11] Leslie D. Montgomery and Bill A. Williams. 1976. Effect of Ambient Temperature on the Thermal Profile of the Human Forearm, Hand, and Fingers. Annals of Biomedical Engineering 4, 3: 209-219. http://dx.doi.org/10.1007/BF02584515.
#### Sensing Cold-Induced Situational Impairments in Mobile Interaction • 98:9

- [12] Aart Overeem, J. Robinson, Hidde Leijnse, Gert-Jan Steeneveld, B. P Horn and Remko Uijlenhoet. 2013. Crowdsourcing urban air temperatures from smartphone battery temperatures. *Geophysical Research Letters* 40, 15: 4081-4085.
- [13] Zhanna Sarsenbayeva, Jorge Goncalves, Juan García, Simon Klakegg, Sirkka Rissanen, Hannu Rintamäki, Jari Hannu and Vassilis Kostakos. 2016. Situational Impairments to Mobile Interaction in Cold Environments. In *International Joint Conference on Pervasive and Ubiquitous Computing*, ACM 85-96. http://dx.doi.org/10.1145/2971648.2971734.
  [14] Frans Snik, Jeroen H. Rietjens, Arnoud Apituley, Hester Volten, Bas Mijling, Antonio Di Noia, Stephanie Heikamp, Ritse C.
- [14] Frans Snik, Jeroen H. Rietjens, Arnoud Apituley, Hester Volten, Bas Mijling, Antonio Di Noia, Stephanie Heikamp, Ritse C. Heinsbroek, Otto P. Hasekamp, J. M. Smit, Jan Vonk, Daphne M. Stam, Gerard van Harten, Jozua de Boer and Christoph U. Keller. 2014. Mapping atmospheric aerosols with a citizen science network of smartphone spectropolarimeters. *Geophysical Research Letters* 41, 20: 7351-7358. http://dx.doi.org/10.1002/2014GL061462.
- [15] Niels van Berkel, Chu Luo, Theodoros Anagnostopoulos, Denzil Ferreira, Jorge Goncalves, Simo Hosio and Vassilis Kostakos. 2016. A Systematic Assessment of Smartphone Usage Gaps. In *Conference on Human Factors in Computing Systems*, ACM, 4711-4721. http://dx.doi.org/10.1145/2858036.2858348.
- [16] Jacob O. Wobbrock. 2006. The future of mobile device research in HCI. In CHI 2006 workshop proceedings: what is the next generation of human-computer interaction, 131-134.

Received May 2017; revised June 2017; accepted July 2017

PACM Interact. Mob. Wearable Ubiquitous Technol., Vol. 1, No. 3, Article 98. Publication date: September 2017.

## **Chapter 9**

### Discussion

This chapter contemplates on the research contributions presented in Chapters 4, 5, 6, and 8 of this thesis in relation to existing literature and answers research questions defined at the beginning of the thesis. We conclude this chapter with the limitations faced by the studies included in this thesis and offer a pathway for future work to be conducted within this research area.

# 9.1 RQ1. What are the effects of different SIIDs on mobile interaction?

This thesis demonstrates that different SIIDs have different effects on mobile interaction. We also show that the effects of SIIDs varied depending on the smartphone tasks.

#### 9.1.1 Ambient Noise

In Chapter 4 we demonstrate that distinct types of ambient noise affect mobile interaction performance in different ways. For example, music with both fast and slow tempo decreased participants' target access time; hence, accelerated their performance in the target acquisition task. Regardless this improvement in target access time, participants were significantly less accurate when accessing targets as the touch offset size increased under both music conditions when compared to the baseline condition. These findings are in line with literature that has previously shown that although music accelerates task performance speed, this improvement comes at a cost of reduced accuracy [52]. However, the effect of music was limited to the target acquisition task only, and it did not have a statistically significant effect on neither the visual search nor the text entry tasks. In addition, the qualitative data from this study showed that music was in general positively perceived by the participants and added a "fun factor" to the

task. Participants felt that under the music condition they were quicker tapping on the targets as they wanted to be *"in line with the music rhythm"* (12 out of 24 participants).

We then show that urban noise had a detrimental effect on mobile interaction performance. Similar to music condition, under both urban indoor and outdoor noise types, participants were quicker to tap targets; however, unlike in the music conditions, this improvement in speed did not affect their accuracy. Furthermore, when exposed to urban indoor noise condition, participants also took significantly less time to memorise a target in the visual search task. However, such an effect of urban outdoor noise condition was not observed. In addition, neither of urban indoor and outdoor noise conditions did not have an effect on visual search time. Nevertheless, we observed a negative effect of urban outdoor noise on the text entry task, as the time per character entry was significantly longer as compared to the silent condition. These findings can be explained by the distracting nature of urban noise [19]; in order to escape these unpleasant conditions participants were quicker to complete these tasks. Moreover, as the outdoor urban noise increases cognitive load [18], it affected participants' performance during the text entry task; hence, they required more time to concentrate to type the text. This finding was also confirmed with the qualitative data. The majority of the participants (13 out of 24) mentioned that they were "distracted" and "annoyed" by the urban noise conditions and could not concentrate on the smartphone tasks. In fact, the participants mentioned that their perceived performance in both the visual search and text entry tasks was negatively affected by the urban noise conditions; hence, supporting our quantitative findings.

Finally, we demonstrate that the effects of speech were limited to the text entry task; but neither the target acquisition nor the visual search tasks. In particular, we show that when listening to the meaningful speech (i.e., English), participants took significantly longer time per character entry compared to the silent condition. This is due to the distracting nature of meaningful speech [19] and its negative effect on cognitive performance [17, 261]. As the text entry task involves cognitive abilities of people [229], it was affected by the additional load caused by the meaningful speech. Interestingly, meaningless speech did not have a significant effect on the text entry performance, which shows that participants could ignore the presence of the meaningless speech due to their inability to understand the language presented in this condition (i.e., Kazakh). Quantitative data conforms with the qualitative findings from the study. Our participants (13 out 24) mentioned that meaningful speech was distracting them, hence it was more difficult to focus on the text entry task. In addition, our participants mentioned that the meaningful speech also affected their performance on the visual search task. However, our quantitative data did not support this statement.

#### 9.1.2 Stress

Chapter 5 presents our findings on the effects of stress on mobile interaction performance. Our results show that stress negatively affected the target acquisition task; however, its' negative effect on the visual search and the text entry task was not observed. We demonstrate that participants took significantly less time to tap targets, and their accuracy was significantly lower when they were experiencing stress. These findings are in line with literature that demonstrates that under stress people tend to rush through their tasks and in general have higher error rates due to anxiety [163].

Qualitative data from this study shows that stress affected not only participants mobile interaction performance, but also physical and emotional well-being. The majority of the participants stated that they were more jittery when experiencing stress. This might have been a reason for reduced accuracy in the target acquisition task. Moreover, the majority of our participants (19 out of 24) said they wanted to complete the smartphone tasks as soon as possible as they did not feel comfortable and wanted to "escape" the situation. This also contributed to their reduced accuracy in the target acquisition task, even though we did not observe a significant negative effect of stress on the visual search and text entry tasks.

#### 9.1.3 Dim Ambient Light

Chapter 6 presents our results on the effects of dim ambient light on mobile interaction performance. Our study shows that dim light has a negative effect on the target acquisition task as it increases the target access time and increases the touch offset size as compared to the normal light condition. This effect of dim ambient light might be due to the decreased colour-differentiation abilities [220] and increased difficulty to see the content of the screen in dark environments [77]. Chapter 6 further demonstrates that wearing sunglasses had a negative effect on the visual search task, as the participants were significantly slower memorising a target icon when compared to a normal light condition. Previous work has shown that low illuminance decreases search speed in reading task [139], hence as sunglasses dim vision perception, our results agree with previous work [139].

Qualitative data of this study shows that participants (8 out of 28) felt the negative effect of dim ambient light and wearing sunglasses on their perceived smartphone interaction performance in the target acquisition task as they mentioned they took longer time and were less accurate to tap the targets under dim light and when wearing sunglasses. This statement is in line with our quantitative data. Interestingly enough, the majority of our participants (17 out of 28) did not feel the negative effect of dim ambient light on their time to memorise a target icon in the visual search task, even though our quantitative data shows the opposite of this statement, as we found that while wearing sunglasses participants

took significantly longer time to memorise an icon. Furthermore, the participants mentioned that it was difficult to find target icons (10 our out 28) and type the text (8 out of 28) under the dim light and when wearing sunglasses. Nevertheless, these statements were not supported via our quantitative data and we did not establish the effects of dim ambient light on mobile interaction performance in the visual search and the text entry tasks.

#### 9.1.4 Summary

The above-mentioned studies have quantified the effects of ambient noise, stress, and dim ambient light on mobile interaction performance. Our findings show that there are effects in both directions, *e.g.*, either improving or decreasing performance on mobile interaction and each SIID has an individual effect on each of the smartphone tasks. These effects are summarised in Table 7.1. The effects of SIIDs are different due to several factors including but not limited to the physiological and mental response of the human body to SIIDs. For example, in cold ambience, body temperature drops, including the temperature of the extremities, as it is a natural physiological response [170]. With the drop in temperature, the muscle stiffness grows and the dexterity drops [91]. Hence, cold-induced SIIDs will affect mobile tasks requiring finger dexterity more compared to other SIIDs that do not affect motor performance. Furthermore, we can observe from Table 7.1 that under stress, participants were significantly quicker to complete target access and memorisation in visual search task. This is due to the fact that stress affected our participants mentally and caused anxiety [128], which in its turn decreased task completion time as has been shown in prior research [163]. Having shown the impact of these three SIIDs on different types of mobile tasks, next we discuss how these effects compare to each other.

## 9.2 RQ2. How do these effects compare to each other?

A significant amount of research has been conducted to investigate the effects of different SIIDs on mobile interaction; however, the majority of the studies were conducted in an ad-hoc fashion and research on SIIDs lacks systematic investigation. This results in the inability to draw fair comparisons between the effects of different SIIDs, and, hence, establish a consensus within the SIIDs research area. Therefore, in order to draw a fair comparison between the effects of different SIIDs on mobile interaction several factors need to be taken into consideration.

First, it is necessary to use identical measure variables to observe the effects of SIIDs on mobile interaction. This can be achieved by using the same tasks

across multiple studies and measuring the same variables within the task. For example, this thesis presents how three smartphone tasks: target acquisition, visual search, and text entry were used across the different studies. Despite the difference between the SIIDs, these smartphone tasks measured the effects of the SIIDs on mobile interaction performance in terms of target access time and touch offset size (target acquisition task), memorisation time and visual search time (visual search task), time per character entry and total error rate (text entry task). This consistency among measure variables was the first step to provide a fair comparison between the effects of different SIIDs on mobile interaction performance. In addition, it is necessary to provide the participants with extensive training to avoid any possibility of sequence effects.

Then, there has to be a consistency within the protocol of the studies. Namely, each of the studies require having a baseline condition which allows observing participants' behaviour when SIIDs are not present. This further provides with an opportunity to compare the effects of different SIIDs respective to their baseline. Most importantly, the effects of SIIDs should not be compared directly to each other, but rather indirectly as a percentile difference to the respective baseline. Performing comparison between the SIIDs according to their percentile growth/drop respective to their baseline, reduces the presence of possible bias that can be caused by the variation in the participants sample.

This comparison empowers our understanding in terms of the magnitude of the effects for each SIID, as we can anticipate which SIID has a more prominent effect on mobile interaction performance. This knowledge can then facilitate building appropriate sensing, modelling, and adapting mechanisms to accommodate the effects of SIIDs, perhaps giving a preference to the most prominent one for a given task. Designing mobile technology that addresses user needs appropriate to their contextual factors can improve people's user experience and ensure that they receive procured information in a timely manner [247].

## 9.3 RQ3. How can mobile devices sense the onset of cold-induced SIIDs?

HCI and UbiComp research has demonstrated that smartphone sensors can be utilised to detect context [2,56,240], surrounding environment [198], and activity recognition [8,252,265]. We followed a similar approach and used a smartphone sensor to sense cold-induced SIIDs. In particular, Chapter 8 presents our work on how we utilised smartphone's built-in battery sensor to detect if a user is experiencing cold-induced SIIDs.

Our findings demonstrate that smartphone battery temperature highly correlates with participants' finger temperature (r = 0.86 and r = 0.85 for index finger and thumb respectively). Both smartphone battery temperature and participants' finger temperature dropped in a cold environment and increased in a warm setting. Hence, smartphone battery temperature can be an estimate to user's finger temperature and a drop in temperature can indicate that the user is exposed to cold setting and potentially experiencing cold-induced situational impairments.

The detection mechanism might work in the following way. The device's battery temperature should be registered by the OS at the moment of the first interaction with the device and needs to be recorded at particular time intervals. In case when the consistent temperature drop is observed and reached a specific threshold, the device can adapt the interface to accommodate cold-induced SIIDs and warn the user about possible overexposure to cold temperatures. Moreover, mechanisms to model or predict how a user interacts with their mobile phone and for how long [108, 125, 144, 258] can further assist with effective adaptations.

We also argue that the detection mechanism can be improved if smartphone's battery temperature is used in combination with user input (*e.g.*, performance during target acquisition task). Nevertheless, it is important to account for external factors that might be present during detection time. For example, a user might be wearing capacitive gloves while interacting with the smartphone, hence diminishing the accuracy of detection. However, we argue that a smartphone should be able to distinguish between the interaction of the user when wearing capacitive gloves and when not.

#### 9.4 Future Directions for SIID Research

As Wobbrock *et al.* argue in their work [273], the concept of disability caused by the context is applicable to everyone. Therefore, conducting research on SIIDs can benefit users of all abilities. For example, if technology is designed to be used in a one-handed interaction mode [37,72] to accommodate SIIDs caused by encumbrance [183,186], it can also be beneficial for a user who has only one arm.

Within the scope of this thesis, SIIDs such as ambient noise, stress, and dim ambient light have been investigated to quantify their effects on mobile interaction. These SIIDs have been investigated independently and under controlled laboratory settings. These can be limitations of the research conducted in this thesis; however, these experimental designs were necessary for several reasons. First of all, as the effects of the above-mentioned SIIDs have not been established on mobile interaction, it was necessary to exclude any confounding factors that might potentially influence and add to the effects of each of the SIIDs on mobile interaction. Therefore, having strictly controlled laboratory settings helped us to avoid any potential additional factors which would question reliability of our results.

Now that we have established the effects of these SIIDs on mobile interaction, it is time to take the research on SIIDs to the next step. For example, future research could examine the effects of combined situational impairments on mobile interaction, particularly because it is common for users to interact with mobile devices while being exposed to multiple SIIDs, *e.g.*, under bright light while walking on the street in winter in Northern Europe. Furthermore, future research on SIIDs should also be conducted in the wild to maximise the realism of the surrounding context. This could potentially bring new insights to the research agenda as new behaviour of people could be observed under realistic conditions that cannot be observed in laboratory settings.

Furthermore, there are other SIIDs that have been acknowledged and recently summarised [269]. Many of them include but are not limited to behavioural (*e.g.*, operating machinery), environmental (*e.g.*, difficult terrain, confinement, extraneous force), attentional (*e.g.*, diverted gaze, multitasking, distraction), affective (*e.g.*, fatigue, fear, intoxication), social (*e.g.*, laws, crowds, social norms), and technological (*e.g.*, lack of power and/or connectivity) SIIDs and remain underexplored as their effects on mobile interaction is unknown [269]. It is crucial to build a scientific understanding of the underexplored SIIDs, in order to be able to progress further in terms of building sensing, modelling, and adapting mechanisms for these SIIDs [269].

If the effects of underexplored SIIDs (*e.g.*, fear or difficult terrain) on mobile interaction performance are shown to be similar to the ones that have already been studied (*e.g.*, stress), then design guidelines for user interfaces to accommodate explored SIIDs could also be applied to address underexplored SIIDs. Furthermore, as researchers in accessible computing have already emphasised the link between building design solutions for people with disabilities and people in disabling situations [269], the broader knowledge on SIIDs and the ways to accommodate them enables potential solutions for broader range of accessible technology to assist people with permanent disabilities. Furthermore, it might be more beneficial to run lab studies to understand the effects of these aforementioned underexplored SIIDs before conducting in-the-wild studies to avoid exhaustion of resources (*e.g.*, human, financial, technological).

Wobbrock *et al.* defined SIIDs according to a 2D space defined by location and duration [269, 273]. Location SIIDs arise "from within" the user, "from without" the user, and might have a "mixed" nature. "From within" SIIDs are present in almost every context as they arise from user's internal states, *e.g.*, mood, emotions, being asleep [273]. Meanwhile, "from without" SIIDs are caused by the surrounding context and can be eliminated or reduced, if the context is removed [273]. Finally, as the names speaks for itself, "mixed" SIIDs are caused due to both the context and internal state of the user, *e.g.*, a visuallyimpaired user interacting with the device under bright light. In terms of duration, Wobbrock *et al.* distinguish ephemeral (short-term and quickly changing, *e.g.*, sleeping, drunkenness) and enduring (long-term or permanent, *e.g.*, blindness, arthritis) SIIDs [273].

Hence, one other potential direction for SIIDs research is to investigate SIIDs according to the above-mentioned 2D space [273]. For example, very few works have looked into the "from within" SIIDs, such as user affect and emotional

state, including our work presented in Chapter 5. Several works have looked into emotions and mood during mobile interaction, *e.g.*, [159, 164, 230, 232, 262]; however, research has not investigated user mood or emotions from the perspective of situational impairments. For example, Mehrotra *et al.* [164] in their work investigate the relationship between the user emotions and application use behaviour. Similarly, work presented in [232] investigates causal relationship between the applications usage behaviour and user emotions. Nevertheless, the aforementioned papers did not measure the performance during mobile interaction and did not investigate the effects of user's emotional state on mobile interaction. Thus, the effects of "from within" SIIDs on mobile interaction remain as a subject for future investigation.

Another aspect to consider when conducting research on SIIDs is the personal and individualistic characteristics of each user. However, designing and building adaptive interfaces to consider personalised behaviour and design might be costly and resource-consuming [273]. Nevertheless, Gajos *et al.* demonstrate that these challenges are solvable by using optimisation algorithms and formulating cost functions [66]. Hence, adaptive interfaces would record user individualistic preferences and abilities, and consider them in familiar and unfamiliar situations [273]. Moreover, researchers should develop and provide a better understanding and judgement of SIIDs and situations in order to decide if the adaptation should take place at all or simply advise the user not to engage in a certain task when they are situationally-impaired. This is crucially important for situations that might come at a high cost, especially when visual and auditory attention needs to be directed to a task of a greater significance (*e.g.*, interacting with a mobile device while crossing a busy intersection).

Finally, scientists conducting research on SIIDs should expand the research scope outside the context of mobile devices. For example, people might be situationally-impaired when using other technology (*e.g.*, musical instruments [9], smartwatches [263], wearable sensors [264], smartspeakers [93], tabletops [241, 254], tablets [101], public displays [73,74,160], miniaturised NIRS scanners [105, 123] and many others). Therefore, given the prevalence of different technology on the market and its availability to the general population, it is important to improve the accessibility of these devices by studying user experience under different SIIDs.

## Chapter 10

### Conclusion

This thesis systematically quantifies the effects of situationally-induced impairments and disabilities, namely ambient noise, stress, and dim ambient light on mobile interaction through common mobile tasks: target acquisition, visual search, and text entry. This thesis also offers a mechanism for sensing coldinduced situational impairments. The results of the studies presented in this thesis demonstrate that different types of ambient noise affect human performance on different smartphone tasks. For example, music shortens the target acquisition time, while decreases the target acquisition accuracy. Similar to music, urban noise decreases the target access time; however does not influence the accuracy of target acquisition. Nevertheless, urban noise decreases the target memorisation time - an aspect of the visual search task, as well as increases time per character entry in the text entry task. Furthermore, meaningful speech similar to urban noise negatively affects text entry as it prolongs the time per character entry. The effects of internal factor - stress - was also measured on mobile interaction performance. Our findings show that stress shortens the target access time, however increases the touch offset size, hence deteriorates the accuracy of touch. Finally, we also studied the effects of dim ambient light and report that under dim ambient light target access time increases while the accuracy of touch drops as well as the target memorisation time increases.

Furthermore, this thesis presents a methodology that enables a fair comparison of the magnitude of the effects of different SIIDs on mobile interaction. First, we recommend using the similar variables that quantify mobile interaction performance under different SIIDs to measure their effect. Second, it is important for each of the experiment to contain a baseline condition to record mobile interaction performance without the influence of SIIDs. Finally, when comparing the effects of different SIIDs to each other, it is necessary to compare a percentage growth/drop in performance respective to the baseline condition, rather than manipulating direct comparisons between the effects.

We then recommend using smartphone's battery temperature for sensing coldinduced SIIDs. We demonstrate that smartphone battery temperature correlates highly with human finger temperature in cold and warm environments. We advocate that our method is applicable during continuous interaction with the device and enables off-the-shelf smartphones to sense cold-induced SIIDs. Hence, being aware of this information, off-the-shelf smartphones can then adapt the interface accordingly to accommodate the effects of cold-induced SIIDs.

In addition, we argue that providing a systematic understanding of the effects of underexplored SIIDs is the pillar of SIIDs research agenda. Better understanding of the effects of different SIIDs enables further construction of sensing mechanisms, creation of techniques to model SIIDs and user behaviour under SIIDs, as well as interface adaptation to accommodate for SIIDs. We foresee that future directions for SIIDs research should focus on studying the effects of combined SIIDs and SIIDs that come "from within" the user (*e.g.*, emotions). Furthermore, the scope of SIIDs research should be extended outside mobile devices by considering their effects on other technology (*e.g.*, smartwatches, smartspeakers). Finally, a user's personal and individualistic characteristics should be considered when conducting future research on SIIDs.

## Bibliography

- [1] Ali Abdolrahmani, Ravi Kuber, and Amy Hurst. An empirical investigation of the situationally-induced impairments experienced by blind mobile device users. In *Proceedings of the 13th Web for All Conference*, W4A '16, pages 21:1–21:8, New York, NY, USA, 2016. ACM.
- [2] Gregory D. Abowd, Anind K. Dey, Peter J. Brown, Nigel Davies, Mark Smith, and Pete Steggles. Towards a better understanding of context and context-awareness. In Hans-W. Gellersen, editor, *Handheld and Ubiquitous Computing*, pages 304–307, Berlin, Heidelberg, 1999. Springer Berlin Heidelberg.
- [3] Dustin Adams, Lourdes Morales, and Sri Kurniawan. A qualitative study to support a blind photography mobile application. In *Proceedings of the* 6th International Conference on PErvasive Technologies Related to Assistive Environments, PETRA '13, pages 25:1–25:8, New York, NY, USA, 2013. ACM.
- [4] Tousif Ahmed, Roberto Hoyle, Kay Connelly, David Crandall, and Apu Kapadia. Privacy concerns and behaviors of people with visual impairments. In *Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems*, CHI '15, pages 3523–3532, New York, NY, USA, 2015. ACM.
- [5] Khaldoun Al Faraj, Mustapha Mojahid, and Nadine Vigouroux. Bigkey: A virtual keyboard for mobile devices. In Julie A. Jacko, editor, *Human-Computer Interaction*. Ambient, Ubiquitous and Intelligent Interaction, pages 3–10, Berlin, Heidelberg, 2009. Springer Berlin Heidelberg.
- [6] Robert Amar, Steven Dow, Richard Gordon, Muhammad Raafay Hamid, and Chad Sellers. Mobile advice: An accessible device for visually impaired capability enhancement. In CHI '03 Extended Abstracts on Human Factors in Computing Systems, CHI EA '03, pages 918–919, New York, NY, USA, 2003. ACM.

- [7] Theodoros Anagnostopoulos, Juan Camilo Garcia, Jorge Goncalves, Denzil Ferreira, Simo Hosio, and Vassilis Kostakos. Environmental exposure assessment using indoor/outdoor detection on smartphones. *Personal and Ubiquitous Computing*, 21(4):761–773, Aug 2017.
- [8] Alvina Anjum and Muhammad U. Ilyas. Activity recognition using smartphone sensors. In 2013 IEEE 10th Consumer Communications and Networking Conference (CCNC), pages 914–919, Jan 2013.
- [9] Juan Pablo Martinez Avila, Chris Greenhalgh, Adrian Hazzard, Steve Benford, and Alan Chamberlain. Encumbered interaction: A study of musicians preparing to perform. In *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, CHI '19, New York, NY, USA, 2019. Association for Computing Machinery.
- [10] Marco Avvenuti and Alessio Vecchio. Mobile visual access to legacy voicebased applications. In *Proceedings of the 6th International Conference on Mobile Technology, Application Systems,* Mobility '09, New York, NY, USA, 2009. Association for Computing Machinery.
- [11] Shiri Azenkot, Cynthia L. Bennett, and Richard E. Ladner. Digitaps: Eyesfree number entry on touchscreens with minimal audio feedback. In *Proceedings of the 26th Annual ACM Symposium on User Interface Software and Technology*, UIST '13, pages 85–90, New York, NY, USA, 2013. ACM.
- [12] Shiri Azenkot, Richard E. Ladner, and Jacob O. Wobbrock. Smartphone haptic feedback for nonvisual wayfinding. In *The Proceedings of the 13th International ACM SIGACCESS Conference on Computers and Accessibility*, ASSETS '11, pages 281–282, New York, NY, USA, 2011. ACM.
- [13] Shiri Azenkot, Sanjana Prasain, Alan Borning, Emily Fortuna, Richard E. Ladner, and Jacob O. Wobbrock. Enhancing independence and safety for blind and deaf-blind public transit riders. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '11, pages 3247– 3256, New York, NY, USA, 2011. ACM.
- [14] Shiri Azenkot, Kyle Rector, Richard Ladner, and Jacob O. Wobbrock. Passchords: Secure multi-touch authentication for blind people. In *Proceedings* of the 14th International ACM SIGACCESS Conference on Computers and Accessibility, ASSETS '12, pages 159–166, New York, NY, USA, 2012. ACM.
- [15] Shiri Azenkot, Jacob O. Wobbrock, Sanjana Prasain, and Richard E. Ladner.

Input finger detection for nonvisual touch screen text entry in perkinput. In *Proceedings of Graphics Interface 2012*, GI '12, pages 121–129, Toronto, Ont., Canada, Canada, 2012. Canadian Information Processing Society.

- [16] Shiri Azenkot and Shumin Zhai. Touch behavior with different postures on soft smartphone keyboards. In *Proceedings of the 14th international conference on Human-computer interaction with mobile devices and services*, pages 251–260. ACM, 2012.
- [17] Simon P. Banbury and Dianne C. Berry. Habituation and dishabituation to speech and office noise. *Journal of Experimental Psychology: Applied*, 3(3):181, 1997.
- [18] Simon P. Banbury and Dianne C. Berry. Disruption of office-related tasks by speech and office noise. *British journal of psychology*, 89(3):499–517, 1998.
- [19] Simon P. Banbury, William J. Macken, Sébastien Tremblay, and Dylan M. Jones. Auditory distraction and short-term memory: Phenomena and practical implications. *Human Factors*, 43(1):12–29, 2001. PMID: 11474757.
- [20] Leon Barnard, Ji Soo Yi, Julie A. Jacko, and Andrew Sears. Capturing the effects of context on human performance in mobile computing systems. *Personal Ubiquitous Comput.*, 11(2):81–96, January 2007.
- [21] Patrick Baudisch and Gerry Chu. Back-of-device interaction allows creating very small touch devices. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '09, page 1923–1932, New York, NY, USA, 2009. Association for Computing Machinery.
- [22] Lynda Berends and Jennifer Johnston. Using multiple coders to enhance qualitative analysis: The case of interviews with consumers of drug treatment. *Addiction Research & Theory*, 13(4):373–381, 2005.
- [23] Joanna Bergstrom-Lehtovirta, Antti Oulasvirta, and Stephen Brewster. The effects of walking speed on target acquisition on a touchscreen interface. In Proceedings of the 13th International Conference on Human Computer Interaction with Mobile Devices and Services, MobileHCI '11, pages 143–146, New York, NY, USA, 2011. ACM.
- [24] Anna-Christina Blomkvist and Gunvor Gard. Computer usage with cold hands; an experiment with pointing devices. *International Journal of Occupational Safety and Ergonomics*, 6(4):429–450, 2000. PMID: 11135678.

- [25] Matthew N. Bonner, Jeremy T. Brudvik, Gregory D. Abowd, and W. Keith Edwards. No-look notes: Accessible eyes-free multi-touch text entry. In Patrik Floréen, Antonio Krüger, and Mirjana Spasojevic, editors, *Pervasive Computing*, pages 409–426, Berlin, Heidelberg, 2010. Springer Berlin Heidelberg.
- [26] Sebastian Boring, David Ledo, Xiang "Anthony" Chen, Nicolai Marquardt, Anthony Tang, and Saul Greenberg. The fat thumb: Using the thumb's contact size for single-handed mobile interaction. In *Proceedings of the* 14th International Conference on Human-Computer Interaction with Mobile Devices and Services, MobileHCI '12, page 39–48, New York, NY, USA, 2012. Association for Computing Machinery.
- [27] Erin Brady, Meredith Ringel Morris, Yu Zhong, Samuel White, and Jeffrey P. Bigham. Visual challenges in the everyday lives of blind people. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '13, pages 2117–2126, New York, NY, USA, 2013. ACM.
- [28] Andrew Bragdon, Eugene Nelson, Yang Li, and Ken Hinckley. Experimental analysis of touch-screen gesture designs in mobile environments. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '11, pages 403–412, New York, NY, USA, 2011. ACM.
- [29] Virginia Braun, Victoria Clarke, Nikki Hayfield, and Gareth Terry. *Thematic Analysis*, pages 843–860. Springer Singapore, Singapore, 2019.
- [30] Sharon L. Brennan-Olsen, S. Cook, M.T. Leech, Steven J. Bowe, Paul Kowal, N. Naidoo, I.N. Ackerman, R.S. Page, S.M. Hosking, J.A. Pasco, and M. Mohebbi. Prevalence of arthritis according to age, sex and socioeconomic status in six low and middle income countries. *BMC musculoskeletal disorders*, 18(1):271, 2017.
- [31] Stephen Brewster. Overcoming the lack of screen space on mobile computers. *Personal and Ubiquitous Computing*, 6(3):188–205, 2002.
- [32] Stephen Brewster, Stephen Brewster, Faraz Chohan, and Lorna Brown. Tactile feedback for mobile interactions. In *Proceedings of the SIGCHI Conference* on Human Factors in Computing Systems, CHI '07, pages 159–162, New York, NY, USA, 2007. ACM.
- [33] Stephen Brewster, Joanna Lumsden, Marek Bell, Malcolm Hall, and Stuart Tasker. Multimodal 'eyes-free' interaction techniques for wearable devices.

In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, CHI '03, pages 473–480, New York, NY, USA, 2003. ACM.

- [34] Carl Brown. Assistive technology computers and persons with disabilities. *Communications of the ACM*, 35(5):36–46, 1992.
- [35] Daniel Buschek and Florian Alt. Probui: Generalising touch target representations to enable declarative gesture definition for probabilistic guis. In *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*, CHI '17, page 4640–4653, New York, NY, USA, 2017. Association for Computing Machinery.
- [36] Daniel Buschek, Alexander De Luca, and Florian Alt. Evaluating the influence of targets and hand postures on touch-based behavioural biometrics. In *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*, CHI '16, page 1349–1361, New York, NY, USA, 2016. Association for Computing Machinery.
- [37] Daniel Buschek, Maximilian Hackenschmied, and Florian Alt. Dynamic ui adaptations for one-handed use of large mobile touchscreen devices. In Regina Bernhaupt, Girish Dalvi, Anirudha Joshi, Devanuj K. Balkrishan, Jacki O'Neill, and Marco Winckler, editors, *Human-Computer Interaction – INTERACT 2017*, pages 184–201, Cham, 2017. Springer International Publishing.
- [38] Daniel Buschek, Fabian Hartmann, Emanuel von Zezschwitz, Alexander De Luca, and Florian Alt. Snapapp: Reducing authentication overhead with a time-constrained fast unlock option. In *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*, CHI '16, page 3736–3747, New York, NY, USA, 2016. Association for Computing Machinery.
- [39] Daniel Buschek, Oliver Schoenleben, and Antti Oulasvirta. Improving accuracy in back-of-device multitouch typing: A clustering-based approach to keyboard updating. In *Proceedings of the 19th International Conference on Intelligent User Interfaces*, IUI '14, page 57–66, New York, NY, USA, 2014. Association for Computing Machinery.
- [40] William Buxton, Ralph Hill, and Peter Rowley. Issues and techniques in touch-sensitive tablet input. In *Proceedings of the 12th Annual Conference on Computer Graphics and Interactive Techniques*, SIGGRAPH '85, pages 215–224, New York, NY, USA, 1985. ACM.

- [41] Maria Claudia Buzzi, Marina Buzzi, Barbara Leporini, and Amaury Trujillo. Designing a text entry multimodal keypad for blind users of touchscreen mobile phones. In *Proceedings of the 16th International ACM SIGACCESS Conference on Computers Accessibility*, ASSETS '14, page 131–136, New York, NY, USA, 2014. Association for Computing Machinery.
- [42] Kelly Caine. Local standards for sample size at chi. In *Proceedings of the* 2016 CHI Conference on Human Factors in Computing Systems, CHI '16, page 981–992, New York, NY, USA, 2016. Association for Computing Machinery.
- [43] Clare Caldwell and Sally A. Hibbert. The influence of music tempo and musical preference on restaurant patrons' behavior. *Psychology & Marketing*, 19(11):895–917, 2002.
- [44] Clare Caldwell and Sally A. Hibbert. The influence of music tempo and musical preference on restaurant patrons' behavior. *Psychology & Marketing*, 19(11):895–917, 2002.
- [45] B. Carignan, J.F. Daneault, C.E. Codere, A.F. Sadikot, and C. Duval. Identifying different pathological tremor characteristics with a smart phone. In *Movement disorders*, volume 27, pages S407–S408. WILEY-BLACKWELL 111 RIVER ST, HOBOKEN 07030-5774, NJ USA, 2012.
- [46] Gianna Cassidy and Raymond A.R. MacDonald. The effect of background music and background noise on the task performance of introverts and extraverts. *Psychology of Music*, 35(3):517–537, 2007.
- [47] Yao-Jen Chang, Shih-Kai Tsai, and Tsen-Yung Wang. A context aware handheld wayfinding system for individuals with cognitive impairments. In *Proceedings of the 10th International ACM SIGACCESS Conference on Computers and Accessibility*, Assets '08, page 27–34, New York, NY, USA, 2008. Association for Computing Machinery.
- [48] Yao-Jen Chang and Tsen-Yung Wang. Comparing picture and video prompting in autonomous indoor wayfinding for individuals with cognitive impairments. *Personal Ubiquitous Comput.*, 14(8):737–747, December 2010.
- [49] Xiang "Anthony" Chen, Tovi Grossman, and George Fitzmaurice. Swipeboard: A text entry technique for ultra-small interfaces that supports novice to expert transitions. In *Proceedings of the 27th Annual ACM Symposium on User Interface Software and Technology*, UIST '14, page 615–620, New York, NY, USA, 2014. Association for Computing Machinery.

- [50] Wendy Chisholm, Gregg Vanderheiden, and Ian Jacobs. Web content accessibility guidelines 1.0. *interactions*, 8(4):35–54, July 2001.
- [51] Yong Jun Choi. Investigation of the effects of context on users' behaviors and performance in mobile computing systems. In *Proceedings of the 12th WSEAS International Conference on Systems*, ICS'08, pages 675–680, Stevens Point, Wisconsin, USA, 2008. World Scientific and Engineering Academy and Society (WSEAS).
- [52] Brian H. Dalton and David G. Behm. Effects of noise and music on human and task performance: A systematic review. *Occupational ergonomics*, 7(3):143–152, 2007.
- [53] Jean-Francois Daneault, Benoit Carignan, Carl Éric Codère, Abbas Sadikot, and Christian Duval. Using a smart phone as a standalone platform for detection and monitoring of pathological tremors. *Frontiers in Human Neuroscience*, 6:357, 2013.
- [54] Melissa Dawe. Understanding mobile phone requirements for young adults with cognitive disabilities. In *Proceedings of the 9th International ACM SIGACCESS Conference on Computers and Accessibility*, Assets '07, pages 179–186, New York, NY, USA, 2007. ACM.
- [55] Melissa Dawe. Understanding mobile phone requirements for young adults with cognitive disabilities. In *Proceedings of the 9th International* ACM SIGACCESS Conference on Computers and Accessibility, Assets '07, page 179–186, New York, NY, USA, 2007. Association for Computing Machinery.
- [56] Anind K. Dey. Understanding and using context. *Personal and Ubiquitous Computing*, 5(1):4–7, Feb 2001.
- [57] Empatica. Real-time physiological signals: E4 eda/gsr sensor.
- [58] Leah Findlater, Jacob O. Wobbrock, and Daniel Wigdor. Typing on flat glass: Examining ten-finger expert typing patterns on touch surfaces. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '11, pages 2453–2462, New York, NY, USA, 2011. ACM.
- [59] Paul M. Fitts. The information capacity of the human motor system in controlling the amplitude of movement. *Journal of experimental psychology*, 47(6):381, 1954.
- [60] David R. Flatla and Carl Gutwin. Individual models of color differentiation

to improve interpretability of information visualization. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '10, page 2563–2572, New York, NY, USA, 2010. Association for Computing Machinery.

- [61] Andreas D. Flouris, David A. Westwood, and Stephen S. Cheung. Thermal balance effects on vigilance during 2-hour exposures to- 20 c. *Aviation*, *space, and environmental medicine*, 78(7):673–679, 2007.
- [62] Raymond Fok, Harmanpreet Kaur, Skanda Palani, Martez E. Mott, and Walter S. Lasecki. Towards more robust speech interactions for deaf and hard of hearing users. In *Proceedings of the 20th International ACM SIGAC-CESS Conference on Computers and Accessibility*, ASSETS '18, page 57–67, New York, NY, USA, 2018. Association for Computing Machinery.
- [63] Center for Persons with Disabilities. Screen reader user survey 7 results, Dec 2017.
- [64] Brian Frey, Caleb Southern, and Mario Romero. Brailletouch: Mobile texting for the visually impaired. In Constantine Stephanidis, editor, Universal Access in Human-Computer Interaction. Context Diversity, pages 19–25, Berlin, Heidelberg, 2011. Springer Berlin Heidelberg.
- [65] Jon Froehlich, Jacob O. Wobbrock, and Shaun K. Kane. Barrier pointing: Using physical edges to assist target acquisition on mobile device touch screens. In *Proceedings of the 9th International ACM SIGACCESS Conference* on Computers and Accessibility, Assets '07, pages 19–26, New York, NY, USA, 2007. ACM.
- [66] Krzysztof Z. Gajos, Daniel S. Weld, and Jacob O. Wobbrock. Automatically generating personalized user interfaces with supple. *Artificial Intelligence*, 174(12):910 – 950, 2010.
- [67] Iván García-Magariño, Carlos Medrano, Inmaculada Plaza, and Bárbara Oliván. A smartphone-based system for detecting hand tremors in unconstrained environments. *Personal and Ubiquitous Computing*, 20(6):959–971, 2016.
- [68] Abraham Glasser, Kesavan Kushalnagar, and Raja Kushalnagar. Deaf, hard of hearing, and hearing perspectives on using automatic speech recognition in conversation. In Proceedings of the 19th International ACM SIGACCESS Conference on Computers and Accessibility, ASSETS '17, page 427–432, New

York, NY, USA, 2017. Association for Computing Machinery.

- [69] Kentaro Go and Yuki Endo. Catkey: Customizable and adaptable touchscreen keyboard with bubble cursor-like visual feedback. In Cécilia Baranauskas, Philippe Palanque, Julio Abascal, and Simone Diniz Junqueira Barbosa, editors, *Human-Computer Interaction – INTERACT 2007*, pages 493–496, Berlin, Heidelberg, 2007. Springer Berlin Heidelberg.
- [70] Mayank Goel, Leah Findlater, and Jacob O. Wobbrock. Walktype: Using accelerometer data to accomodate situational impairments in mobile touch screen text entry. In *Proc. of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '12, pages 2687–2696, New York, NY, USA, 2012. ACM.
- [71] Mayank Goel, Alex Jansen, Travis Mandel, Shwetak N. Patel, and Jacob O. Wobbrock. Contexttype: using hand posture information to improve mobile touch screen text entry. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, pages 2795–2798. ACM, 2013.
- [72] Mayank Goel, Jacob Wobbrock, and Shwetak Patel. Gripsense: using built-in sensors to detect hand posture and pressure on commodity mobile phones. In *Proceedings of the 25th annual ACM symposium on User interface software and technology*, pages 545–554. ACM, 2012.
- [73] Jorge Goncalves, Denzil Ferreira, Simo Hosio, Yong Liu, Jakob Rogstadius, Hannu Kukka, and Vassilis Kostakos. Crowdsourcing on the spot: Altruistic use of public displays, feasibility, performance, and behaviours. In *Proceedings of the 2013 ACM International Joint Conference on Pervasive and Ubiquitous Computing*, UbiComp '13, page 753–762, New York, NY, USA, 2013. Association for Computing Machinery.
- [74] Jorge Goncalves, Simo Hosio, Jakob Rogstadius, Evangelos Karapanos, and Vassilis Kostakos. Motivating participation and improving quality of contribution in ubiquitous crowdsourcing. *Computer Networks*, 90:34 – 48, 2015. Crowdsourcing.
- [75] Jorge Goncalves, Vassilis Kostakos, Simo Hosio, Evangelos Karapanos, and Olga Lyra. Inclucity: Using contextual cues to raise awareness on environmental accessibility. In *Proceedings of the 15th International ACM SIGACCESS Conference on Computers and Accessibility*, ASSETS '13, New York, NY, USA, 2013. Association for Computing Machinery.

- [76] Jorge Goncalves, Zhanna Sarsenbayeva, Niels van Berkel, Chu Luo, Simo Hosio, Sirkka Risanen, Hannu Rintamäki, and Vassilis Kostakos. Tapping task performance on smartphones in cold temperature. *Interacting with Computers*, 29(3):355–367, 2017.
- [77] Rui Gong, Haisong Xu, Binyu Wang, and M. Ronnier Luo. Image quality evaluation for smart-phone displays at lighting levels of indoor and outdoor conditions. *Optical Engineering*, 51(8):1 – 7, 2012.
- [78] Joshua Goodman, Gina Venolia, Keith Steury, and Chauncey Parker. Language modeling for soft keyboards. In *Proceedings of the 7th International Conference on Intelligent User Interfaces*, IUI '02, pages 194–195, New York, NY, USA, 2002. ACM.
- [79] Wayne A. Gordon. The interface between cognitive impairments and access to information technology. SIGACCESS Access. Comput., (83):3–6, September 2005.
- [80] João Guerreiro, Daniel Gonçalves, Diogo Marques, Tiago Guerreiro, Hugo Nicolau, and Kyle Montague. The today and tomorrow of braille learning. In Proceedings of the 15th International ACM SIGACCESS Conference on Computers and Accessibility, ASSETS '13, New York, NY, USA, 2013. Association for Computing Machinery.
- [81] Tiago Guerreiro and Joaquim Jorge. Assistive technologies for spinal cord injured individuals: Electromyographic mobile accessibility. In *Proceedings of GW 2007, 7th International Workshop on Gesture in Human-Computer Interaction and Simulation, 2007.*
- [82] Tiago Guerreiro, Hugo Nicolau, Joaquim Jorge, and Daniel Gonçalves. Towards accessible touch interfaces. In *Proceedings of the 12th International* ACM SIGACCESS Conference on Computers and Accessibility, ASSETS '10, pages 19–26, New York, NY, USA, 2010. ACM.
- [83] Tiago João Vieira Guerreiro, Hugo Nicolau, Joaquim Jorge, and Daniel Gonçalves. Assessing mobile touch interfaces for tetraplegics. In *Proceedings* of the 12th International Conference on Human Computer Interaction with Mobile Devices and Services, MobileHCI '10, pages 31–34, New York, NY, USA, 2010. ACM.
- [84] Asela Gunawardana, Tim Paek, and Christopher Meek. Usability guided key-target resizing for soft keyboards. In *Proceedings of the 15th International*

*Conference on Intelligent User Interfaces,* IUI '10, pages 111–118, New York, NY, USA, 2010. ACM.

- [85] Sidhant Gupta, Tim Campbell, Jeffrey R. Hightower, and Shwetak N. Patel. Squeezeblock: Using virtual springs in mobile devices for eyes-free interaction. In *Proceedings of the 23nd Annual ACM Symposium on User Interface Software and Technology*, UIST '10, page 101–104, New York, NY, USA, 2010. Association for Computing Machinery.
- [86] Lilit Hakobyan, Jo Lumsden, Dympna O'Sullivan, and Hannah Bartlett. Mobile assistive technologies for the visually impaired. *Survey of Ophthalmology*, 58(6):513 – 528, 2013.
- [87] Md. Munirul Haque, Shams Zawoad, and Ragib Hasan. Secure techniques and methods for authenticating visually impaired mobile phone users. In 2013 IEEE International Conference on Technologies for Homeland Security (HST), pages 735–740, Nov 2013.
- [88] Susumu Harada, Daisuke Sato, Dustin W. Adams, Sri Kurniawan, Hironobu Takagi, and Chieko Asakawa. Accessible photo album: Enhancing the photo sharing experience for people with visual impairment. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '13, pages 2127–2136, New York, NY, USA, 2013. ACM.
- [89] Simon Harper, Yeliz Yesilada, and Tianyi Chen. Mobile device impairment ... similar problems, similar solutions? *Behaviour & Information Technology*, 30(5):673–690, 2011.
- [90] Morgan Harvey and Matthew Pointon. Perceptions of the effect of fragmented attention on mobile web search tasks. In *Proceedings of the 2017 Conference on Conference Human Information Interaction and Retrieval*, CHIIR '17, pages 293–296, New York, NY, USA, 2017. ACM.
- [91] George Havenith, Hein A. M. Daanen, and Ronald Heus. The hand in the cold, performance and risk. *Arctic medical research*, 54(Suppl 2):37–47, 1995.
- [92] Niels Henze, Enrico Rukzio, and Susanne Boll. 100,000,000 taps: Analysis and improvement of touch performance in the large. In *Proceedings of the 13th International Conference on Human Computer Interaction with Mobile Devices and Services*, MobileHCI '11, pages 133–142, New York, NY, USA, 2011. ACM.

- [93] Danula Hettiachchi, Zhanna Sarsenbayeva, Fraser Allison, Niels van Berkel, Tilman Dingler, Gabriele Marini, Vassilis Kostakos, and Jorge Goncalves.
  "Hi! I am the crowd tasker" Crowdsourcing through digital voice assistants. In *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*, CHI '20, pages 193:1–193:14, New York, NY, USA, 2020. Association for Computing Machinery.
- [94] David R. Hill and Christiane Grieb. Substitution for a restricted visual channel in multimodal computer-human dialogue. *IEEE Transactions on Systems, Man, and Cybernetics,* 18(2):285–304, March 1988.
- [95] Johan Himberg, Jonna Häkkilä, Petri Kangas, and Jani Mäntyjärvi. On-line personalization of a touch screen based keyboard. In *Proceedings of the 8th International Conference on Intelligent User Interfaces*, IUI '03, pages 77–84, New York, NY, USA, 2003. ACM.
- [96] Julia Himmelsbach, Markus Garschall, Sebastian Egger, Susanne Steffek, and Manfred Tscheligi. Enabling accessibility through multimodality? Interaction modality choices of older adults. In *Proceedings of the 14th International Conference on Mobile and Ubiquitous Multimedia*, MUM '15, page 195–199, New York, NY, USA, 2015. Association for Computing Machinery.
- [97] Eve Hoggan and Stephen Brewster. Designing audio and tactile crossmodal icons for mobile devices. In *Proceedings of the 9th International Conference on Multimodal Interfaces*, ICMI '07, pages 162–169, New York, NY, USA, 2007. ACM.
- [98] Eve Hoggan, Stephen A. Brewster, and Jody Johnston. Investigating the effectiveness of tactile feedback for mobile touchscreens. In *Proceedings* of the SIGCHI Conference on Human Factors in Computing Systems, CHI '08, pages 1573–1582, New York, NY, USA, 2008. ACM.
- [99] Eve Hoggan, Andrew Crossan, Stephen A. Brewster, and Topi Kaaresoja. Audio or tactile feedback: Which modality when? In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '09, pages 2253–2256, New York, NY, USA, 2009. ACM.
- [100] Richang Hong, Meng Wang, Xiao-Tong Yuan, Mengdi Xu, Jianguo Jiang, Shuicheng Yan, and Tat-Seng Chua. Video accessibility enhancement for hearing-impaired users. ACM Trans. Multimedia Comput. Commun. Appl., 7S(1), November 2011.

- [101] Simo Hosio, Jorge Goncalves, Vili Lehdonvirta, Denzil Ferreira, and Vassilis Kostakos. Situated crowdsourcing using a market model. In *Proceedings of the 27th Annual ACM Symposium on User Interface Software and Technology*, UIST '14, page 55–64, New York, NY, USA, 2014. Association for Computing Machinery.
- [102] Mohit Jain and Ravin Balakrishnan. User learning and performance with bezel menus. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '12, pages 2221–2230, New York, NY, USA, 2012. ACM.
- [103] Chandrika Jayant, Hanjie Ji, Samuel White, and Jeffrey P. Bigham. Supporting blind photography. In *The Proceedings of the 13th International ACM SIGACCESS Conference on Computers and Accessibility*, ASSETS '11, pages 203–210, New York, NY, USA, 2011. ACM.
- [104] Marko Jelici, Elke Geraerts, Harald Merckelbach, and Ramona Guerrieri. Acute stress enhances memory for emotional words, but impairs memory for neutral words. *International Journal of Neuroscience*, 114(10):1343–1351, 2004.
- [105] Weiwei Jiang, Gabriele Marini, Niels van Berkel, Zhanna Sarsenbayeva, Zheyu Tan, Chu Luo, Xin He, Tilman Dingler, Jorge Goncalves, Yoshihiro Kawahara, and Vassilis Kostakos. Probing sucrose contents in everyday drinks using miniaturized near-infrared spectroscopy scanners. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.*, 3(4), December 2019.
- [106] Zhao Xia Jin, Tom Plocher, and Liana Kiff. Touch screen user interfaces for older adults: Button size and spacing. In Constantine Stephanidis, editor, *Universal Access in Human Computer Interaction. Coping with Diversity*, pages 933–941, Berlin, Heidelberg, 2007. Springer Berlin Heidelberg.
- [107] Eleanor Jones, Jason Alexander, Andreas Andreou, Pourang Irani, and Sriram Subramanian. Gestext: Accelerometer-based gestural text-entry systems. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '10, pages 2173–2182, New York, NY, USA, 2010. ACM.
- [108] Simon L. Jones, Denzil Ferreira, Simo Hosio, Jorge Goncalves, and Vassilis Kostakos. Revisitation analysis of smartphone app use. In *Proceedings of the* 2015 ACM International Joint Conference on Pervasive and Ubiquitous Computing, UbiComp '15, page 1197–1208, New York, NY, USA, 2015. Association

for Computing Machinery.

- [109] Raed A. Joundi, John-Stuart Brittain, Ned Jenkinson, Alexander L. Green, and Tipu Aziz. Rapid tremor frequency assessment with the iphone accelerometer. *Parkinsonism & related disorders*, 17(4):288–290, 2011.
- [110] Nikolaos Kaklanis, Konstantinos Votis, and Dimitrios Tzovaras. A mobile interactive maps application for a visually impaired audience. In *Proceedings of the 10th International Cross-Disciplinary Conference on Web Accessibility*, W4A '13, New York, NY, USA, 2013. Association for Computing Machinery.
- [111] Shaun K. Kane, Jeffrey P. Bigham, and Jacob O. Wobbrock. Slide rule: Making mobile touch screens accessible to blind people using multi-touch interaction techniques. In *Proceedings of the 10th International ACM SIGAC-CESS Conference on Computers and Accessibility*, Assets '08, pages 73–80, New York, NY, USA, 2008. ACM.
- [112] Shaun K. Kane, Chandrika Jayant, Jacob O. Wobbrock, and Richard E. Ladner. Freedom to roam: A study of mobile device adoption and accessibility for people with visual and motor disabilities. In *Proceedings of the 11th International ACM SIGACCESS Conference on Computers and Accessibility*, Assets '09, pages 115–122, New York, NY, USA, 2009. ACM.
- [113] Shaun K. Kane, Jacob O. Wobbrock, Mark Harniss, and Kurt L. Johnson. Truekeys: Identifying and correcting typing errors for people with motor impairments. In *Proceedings of the 13th International Conference on Intelligent* User Interfaces, IUI '08, pages 349–352, New York, NY, USA, 2008. ACM.
- [114] Shaun K. Kane, Jacob O. Wobbrock, and Richard E. Ladner. Usable gestures for blind people: Understanding preference and performance. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '11, pages 413–422, New York, NY, USA, 2011. ACM.
- [115] Shaun K. Kane, Jacob O. Wobbrock, and Ian E. Smith. Getting off the treadmill: Evaluating walking user interfaces for mobile devices in public spaces. In *Proceedings of the 10th International Conference on Human Computer Interaction with Mobile Devices and Services*, MobileHCI '08, pages 109–118, New York, NY, USA, 2008. ACM.
- [116] Amy K. Karlson and Benjamin B. Bederson. Thumbspace: Generalized onehanded input for touchscreen-based mobile devices. In Cécilia Baranauskas, Philippe Palanque, Julio Abascal, and Simone Diniz Junqueira Barbosa,

editors, *Human-Computer Interaction – INTERACT 2007*, pages 324–338, Berlin, Heidelberg, 2007. Springer Berlin Heidelberg.

- [117] Amy K. Karlson, Benjamin B. Bederson, and John SanGiovanni. Applens and launchtile: Two designs for one-handed thumb use on small devices. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '05, page 201–210, New York, NY, USA, 2005. Association for Computing Machinery.
- [118] Simeon Keates, John Clarkson, and Peter Robinson. Investigating the applicability of user models for motion-impaired users. In ACM SIGACCESS Conference on Assistive Technologies: Proceedings of the fourth international ACM conference on Assistive technologies, volume 13, pages 129–136, 2000.
- [119] Simeon Keates, Faustina Hwang, Patrick Langdon, P. John Clarkson, and Peter Robinson. Cursor measures for motion-impaired computer users. In Proceedings of the Fifth International ACM Conference on Assistive Technologies, Assets '02, pages 135–142, New York, NY, USA, 2002. ACM.
- [120] Wilhelm Kirch, editor. Pearson's Correlation Coefficient, pages 1090–1091. Springer Netherlands, Dordrecht, 2008.
- [121] Clemens Kirschbaum, Karl-Martin Pirke, and Dirk H. Hellhammer. The 'trier social stress test'–a tool for investigating psychobiological stress responses in a laboratory setting. *Neuropsychobiology*, 28(1-2):76–81, 1993.
- [122] Clemens Kirschbaum, Oliver T. Wolf, Mark May, W. Wippich, and Dirk H. Hellhammer. Stress- and treatment-induced elevations of cortisol levels associated with impaired declarative memory in healthy adults. *Life Sciences*, 58(17):1475 – 1483, 1996.
- [123] Simon Klakegg, Jorge Goncalves, Chu Luo, Aku Visuri, Alexey Popov, Niels van Berkel, Zhanna Sarsenbayeva, Vassilis Kostakos, Simo Hosio, Scott Savage, Alexander Bykov, Igor Meglinski, and Denzil Ferreira. Assisted medication management in elderly care using miniaturised near-infrared spectroscopy. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.*, 2(2), July 2018.
- [124] Sadanori Konishi and Genshiro Kitagawa. *Information criteria and statistical modeling*. Springer Science & Business Media, 2008.
- [125] Vassilis Kostakos, Denzil Ferreira, Jorge Goncalves, and Simo Hosio. Mod-

elling smartphone usage: A markov state transition model. In *Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing*, UbiComp '16, page 486–497, New York, NY, USA, 2016. Association for Computing Machinery.

- [126] Sabrina Kuhlmann, Marcel Piel, and Oliver T. Wolf. Impaired memory retrieval after psychosocial stress in healthy young men. *Journal of Neuro-science*, 25(11):2977–2982, 2005.
- [127] Jonathan Lazar, Jinjuan Heidi Feng, and Harry Hochheiser. *Research methods in human-computer interaction*. Wiley, 2015.
- [128] Richard S. Lazarus, James Deese, and Sonia F. Osler. The effects of psychological stress upon performance. *Psychological bulletin*, 49(4):293, 1952.
- [129] Der-Song Lee, Kong-King Shieh, Shie-Chang Jeng, and I-Hsuan Shen. Effect of character size and lighting on legibility of electronic papers. *Displays*, 29(1):10–17, 2008.
- [130] Der-Song Lee, Kong-King Shieh, Shie-Chang Jeng, and I-Hsuan Shen. Effect of character size and lighting on legibility of electronic papers. *Displays*, 29(1):10 – 17, 2008.
- [131] V. Kathlene Leonard, Julie A. Jacko, and Joseph J. Pizzimenti. An exploratory investigation of handheld computer interaction for older adults with visual impairments. In *Proceedings of the 7th International ACM SIGAC-CESS Conference on Computers and Accessibility*, Assets '05, pages 12–19, New York, NY, USA, 2005. ACM.
- [132] Frank Chun Yat Li, David Dearman, and Khai N. Truong. Leveraging proprioception to make mobile phones more accessible to users with visual impairments. In *Proceedings of the 12th International ACM SIGACCESS Conference on Computers and Accessibility*, ASSETS '10, pages 187–194, New York, NY, USA, 2010. ACM.
- [133] Kevin A. Li, Patrick Baudisch, and Ken Hinckley. Blindsight: Eyes-free access to mobile phones. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '08, pages 1389–1398, New York, NY, USA, 2008. ACM.
- [134] Min Lin, Rich Goldman, Kathleen J. Price, Andrew Sears, and Julie Jacko. How do people tap when walking? An empirical investigation of nomadic

data entry. *International Journal of Human-Computer Studies*, 65(9):759 – 769, 2007.

- [135] Christian Lindholm and Turkku Keinonen. Mobile Usability: How Nokia Changed the Face of the Cellular Phone. McGraw-Hill, Inc., New York, NY, USA, 2003.
- [136] Rich Ling. *The mobile connection: The cell phone's impact on society*. Elsevier, 2004.
- [137] Richard Ling and Birgitte Yttri. 10 hyper–coordination via mobile phones in norway. *Perpetual contact: Mobile communication, private talk, public performance*, 139, 2002.
- [138] Alan L. Liu, Harlan Hile, Henry Kautz, Gaetano Borriello, Pat A. Brown, Mark Harniss, and Kurt Johnson. Indoor wayfinding: Developing a functional interface for individuals with cognitive impairments. *Disability and Rehabilitation: Assistive Technology*, 3(1-2):69–81, 2008.
- [139] Peter Liu, Fahad Zafar, and Aldo Badano. The effect of ambient illumination on handheld display image quality. *Journal of Digital Imaging*, 27(1):12–18, Feb 2014.
- [140] Yong Liu, Jorge Goncalves, Denzil Ferreira, Simo Hosio, and Vassilis Kostakos. Identity crisis of ubicomp? mapping 15 years of the field's development and paradigm change. In *Proceedings of the 2014 ACM International Joint Conference on Pervasive and Ubiquitous Computing*, UbiComp '14, page 75–86, New York, NY, USA, 2014. Association for Computing Machinery.
- [141] Yong Liu, Jorge Goncalves, Denzil Ferreira, Bei Xiao, Simo Hosio, and Vassilis Kostakos. Chi 1994-2013: Mapping two decades of intellectual progress through co-word analysis. In *Proceedings of the SIGCHI Conference* on Human Factors in Computing Systems, CHI '14, page 3553–3562, New York, NY, USA, 2014. Association for Computing Machinery.
- [142] Elan D. Louis and Joaquim J. Ferreira. How common is the most common adult movement disorder? Update on the worldwide prevalence of essential tremor. *Movement Disorders*, 25(5):534–541, 2010.
- [143] Chu Luo, Angelos Fylakis, Juha Partala, Simon Klakegg, Jorge Goncalves, Kaitai Liang, Tapio Seppänen, and Vassilis Kostakos. A data hiding ap-

proach for sensitive smartphone data. In *Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing*, UbiComp '16, pages 557–568, New York, NY, USA, 2016. ACM.

- [144] Chu Luo, Aku Visuri, Simon Klakegg, Niels van Berkel, Zhanna Sarsenbayeva, Antti Möttönen, Jorge Goncalves, Theodoros Anagnostopoulos, Denzil Ferreira, Huber Flores, Eduardo Velloso, and Vassilis Kostakos. Energy-efficient prediction of smartphone unlocking. *Personal and Ubiquitous Computing*, 23(1):159–177, Feb 2019.
- [145] S. J. Lupien, S. Gaudreau, B. M. Tchiteya, F. Maheu, S. Sharma, N. P. V. Nair, R. L. Hauger, B. S. McEwen, and M. J. Meaney. Stress-Induced Declarative Memory Impairment in Healthy Elderly Subjects: Relationship to Cortisol Reactivity1. *The Journal of Clinical Endocrinology Metabolism*, 82(7):2070– 2075, 07 1997.
- [146] Sonia J. Lupien, Christian J. Gillin, and Richard L. Hauger. Working memory is more sensitive than declarative memory to the acute effects of corticosteroids: A dose–response study in humans. *Behavioral neuroscience*, 113(3):420, 1999.
- [147] Kent Lyons, Christopher Skeels, Thad Starner, Cornelis M. Snoeck, Benjamin A. Wong, and Daniel Ashbrook. Augmenting conversations using dual-purpose speech. In *Proceedings of the 17th Annual ACM Symposium on User Interface Software and Technology*, UIST '04, pages 237–246, New York, NY, USA, 2004. ACM.
- [148] Kent Lyons, Thad Starner, Daniel Plaisted, James Fusia, Amanda Lyons, Aaron Drew, and E. W. Looney. Twiddler typing: One-handed chording text entry for mobile phones. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '04, pages 671–678, New York, NY, USA, 2004. ACM.
- [149] Bonnie MacKay, David Dearman, Kori Inkpen, and Carolyn Watters. Walk 'n scroll: A comparison of software-based navigation techniques for different levels of mobility. In *Proceedings of the 7th International Conference on Human Computer Interaction with Mobile Devices & Amp; Services*, MobileHCI '05, pages 183–190, New York, NY, USA, 2005. ACM.
- [150] I. Scott MacKenzie and Poika Isokoski. Fitts' throughput and the speedaccuracy tradeoff. In *Proceedings of the SIGCHI Conference on Human Factors*

*in Computing Systems,* CHI '08, page 1633–1636, New York, NY, USA, 2008. Association for Computing Machinery.

- [151] Päivi Majaranta and Kari-Jouko Räihä. Twenty years of eye typing: Systems and design issues. In *Proceedings of the 2002 Symposium on Eye Tracking Research & Applications*, ETRA '02, pages 15–22, New York, NY, USA, 2002. ACM.
- [152] R. Manduchi and J. Coughlan. Portable and mobile systems in assistive technology. In Klaus Miesenberger, Joachim Klaus, Wolfgang Zagler, and Arthur Karshmer, editors, *Computers Helping People with Special Needs*, pages 1078–1080, Berlin, Heidelberg, 2008. Springer Berlin Heidelberg.
- [153] Alex Mariakakis, Sayna Parsi, Shwetak N. Patel, and Jacob O. Wobbrock. Drunk user interfaces: Determining blood alcohol level through everyday smartphone tasks. In *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, CHI '18, pages 234:1–234:13, New York, NY, USA, 2018. ACM.
- [154] Alexander Mariakakis, Mayank Goel, Md Tanvir Islam Aumi, Shwetak N. Patel, and Jacob O. Wobbrock. Switchback: Using focus and saccade tracking to guide users' attention for mobile task resumption. In *Proceedings* of the 33rd Annual ACM Conference on Human Factors in Computing Systems, CHI '15, pages 2953–2962, New York, NY, USA, 2015. ACM.
- [155] Randi C. Martin, Michael S. Wogalter, and Janice G. Forlano. Reading comprehension in the presence of unattended speech and music. *Journal of Memory and Language*, 27(4):382 – 398, 1988.
- [156] Sergio Mascetti, Cristian Bernareggi, and Matteo Belotti. Typeinbraille: A braille-based typing application for touchscreen devices. In *The Proceedings of the 13th International ACM SIGACCESS Conference on Computers and Accessibility*, ASSETS '11, pages 295–296, New York, NY, USA, 2011. ACM.
- [157] Clifford F. Mass and Luke E. Madaus. Surface pressure observations from smartphones: A potential revolution for high-resolution weather prediction? *Bulletin of the American Meteorological Society*, 95(9):1343–1349, 2014.
- [158] Dominic W. Massaro. A computer-animated tutor for spoken and written language learning. In *Proceedings of the 5th International Conference on Multimodal Interfaces*, ICMI '03, page 172–175, New York, NY, USA, 2003. Association for Computing Machinery.

- [159] Oludamilare Matthews, Zhanna Sarsenbayeva, Weiwei Jiang, Joshua Newn, Eduardo Velloso, Sarah Clinch, and Jorge Goncalves. Inferring the mood of a community from their walking speed: A preliminary study. In Proceedings of the 2018 ACM International Joint Conference and 2018 International Symposium on Pervasive and Ubiquitous Computing and Wearable Computers, UbiComp '18, page 1144–1149, New York, NY, USA, 2018. Association for Computing Machinery.
- [160] Sven Mayer, Lars Lischke, Jens Emil Grønbæk, Zhanna Sarsenbayeva, Jonas Vogelsang, Paweł W. Woundefinedniak, Niels Henze, and Giulio Jacucci. Pac-many: Movement behavior when playing collaborative and competitive games on large displays. In *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, CHI '18, New York, NY, USA, 2018. Association for Computing Machinery.
- [161] Daniel J. McDuff, Javier Hernandez, Sarah Gontarek, and Rosalind W. Picard. Cogcam: Contact-free measurement of cognitive stress during computer tasks with a digital camera. In *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*, CHI '16, page 4000–4004, New York, NY, USA, 2016. Association for Computing Machinery.
- [162] Heather McElrea and Lionel Standing. Fast music causes fast drinking. *Perceptual and Motor skills*, 1992.
- [163] Fred McKinney, George B Strother, Ruth R Hines, and Ruth A Allee. Experimental frustration in a group test situation. *The Journal of Abnormal and Social Psychology*, 46(3):316, 1951.
- [164] Abhinav Mehrotra, Fani Tsapeli, Robert Hendley, and Mirco Musolesi. Mytraces: Investigating correlation and causation between users' emotional states and mobile phone interaction. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.*, 1(3), September 2017.
- [165] Matthew B. Miles and Michael A. Huberman. *Qualitative data analysis: An expanded sourcebook.* sage, 1994.
- [166] Ronald E. Milliman. Using background music to affect the behavior of supermarket shoppers. *Journal of Marketing*, 46(3):86–91, 1982.
- [167] Ronald E. Milliman. The Influence of Background Music on the Behavior of Restaurant Patrons. *Journal of Consumer Research*, 13(2):286–289, 09 1986.

- [168] Sachi Mizobuchi, Mark Chignell, and David Newton. Mobile text entry: Relationship between walking speed and text input task difficulty. In Proceedings of the 7th International Conference on Human Computer Interaction with Mobile Devices & Amp; Services, MobileHCI '05, pages 122–128, New York, NY, USA, 2005. ACM.
- [169] Girish M. Mody and Peter M. Brooks. Improving musculoskeletal health: global issues. *Best Practice & Research Clinical Rheumatology*, 26(2):237–249, 2012.
- [170] Leslie D. Montgomery and Bill A. Williams. Effect of ambient temperature on the thermal profile of the human forearm, hand, and fingers. *Annals of biomedical engineering*, 4(3):209–219, 1976.
- [171] Martez E. Mott and Jacob O. Wobbrock. Cluster touch: Improving touch accuracy on smartphones for people with motor and situational impairments. In *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*, CHI '19, New York, NY, USA, 2019. Association for Computing Machinery.
- [172] Molly Follette Story M.S. Maximizing usability: The principles of universal design. Assistive Technology, 10(1):4–12, 1998. PMID: 10181150.
- [173] Terhi Mustonen, Maria Olkkonen, and Jukka Hakkinen. Examining mobile phone text legibility while walking. In CHI '04 Extended Abstracts on Human Factors in Computing Systems, CHI EA '04, pages 1243–1246, New York, NY, USA, 2004. ACM.
- [174] Brad A. Myers, Jacob O. Wobbrock, Sunny Yang, Brian Yeung, Jeffrey Nichols, and Robert Miller. Using handhelds to help people with motor impairments. In *Proceedings of the Fifth International ACM Conference on Assistive Technologies*, Assets '02, pages 89–96, New York, NY, USA, 2002. ACM.
- [175] Maia Naftali and Leah Findlater. Accessibility in context: Understanding the truly mobile experience of smartphone users with motor impairments. In Proceedings of the 16th International ACM SIGACCESS Conference on Computers & Accessibility, ASSETS '14, pages 209–216, New York, NY, USA, 2014. ACM.
- [176] Priya Narasimhan, Rajeev Gandhi, and Dan Rossi. Smartphone-based assistive technologies for the blind. In *Proceedings of the 2009 International*

*Conference on Compilers, Architecture, and Synthesis for Embedded Systems,* CASES '09, pages 223–232, New York, NY, USA, 2009. ACM.

- [177] Arshad Nasser, Kening Zhu, and Sarah Wiseman. Thermo-haptic earable display for the hearing and visually impaired. In *The 21st International ACM SIGACCESS Conference on Computers and Accessibility*, ASSETS '19, page 630–632, New York, NY, USA, 2019. Association for Computing Machinery.
- [178] John W. Newcomer, Gregg Selke, Angela K. Melson, Tamara Hershey, Suzanne Craft, Katherine Richards, and Amy L. Alderson. Decreased Memory Performance in Healthy Humans Induced by Stress-Level Cortisol Treatment. JAMA Psychiatry, 56(6):527–533, 06 1999.
- [179] Alan F. Newell. Extra-ordinary human-computer interaction. Extra-ordinary human-computer interaction-Interfaces for users with disabilities, pages 3–18, 1995.
- [180] Alan F. Newell, Alex Carmichael, Peter Gregor, and Norman Alm. The human-computer interaction handbook. chapter Information Technology for Cognitive Support, pages 464–481. L. Erlbaum Associates Inc., Hillsdale, NJ, USA, 2003.
- [181] Keith Newman et al. The open interface beyond keyboards and mice. *E.nz Magazine: The Magazine of Technical Enterprise*, 3(3):6, 2002.
- [182] Alexander Ng, Stephen A. Brewster, and John Williamson. The impact of encumbrance on mobile interactions. In Paula Kotzé, Gary Marsden, Gitte Lindgaard, Janet Wesson, and Marco Winckler, editors, *Human-Computer Interaction – INTERACT 2013*, pages 92–109, Berlin, Heidelberg, 2013. Springer Berlin Heidelberg.
- [183] Alexander Ng, Stephen A. Brewster, and John H. Williamson. Investigating the effects of encumbrance on one- and two- handed interactions with mobile devices. In *Proceedings of the 32Nd Annual ACM Conference on Human Factors in Computing Systems*, CHI '14, pages 1981–1990, New York, NY, USA, 2014. ACM.
- [184] Alexander Ng, Stephen A. Brewster, and John H. Williamson. Investigating the effects of encumbrance on one- and two- handed interactions with mobile devices. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '14, pages 1981–1990, New York, NY, USA, 2014. ACM.

- [185] Alexander Ng, John Williamson, and Stephen Brewster. The effects of encumbrance and mobility on touch-based gesture interactions for mobile phones. In *Proceedings of the 17th International Conference on Human-Computer Interaction with Mobile Devices and Services*, MobileHCI '15, pages 536–546, New York, NY, USA, 2015. ACM.
- [186] Alexander Ng, John H. Williamson, and Stephen A. Brewster. Comparing evaluation methods for encumbrance and walking on interaction with touchscreen mobile devices. In *Proceedings of the 16th International Conference on Human-computer Interaction with Mobile Devices & Services,* MobileHCI '14, pages 23–32, New York, NY, USA, 2014. ACM.
- [187] Hugo Nicolau, Tiago Guerreiro, Joaquim Jorge, and Daniel Gonçalves. Mobile touchscreen user interfaces: Bridging the gap between motor-impaired and able-bodied users. *Univers. Access Inf. Soc.*, 13(3):303–313, August 2014.
- [188] Hugo Nicolau and Joaquim Jorge. Touch typing using thumbs: Understanding the effect of mobility and hand posture. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '12, pages 2683–2686, New York, NY, USA, 2012. ACM.
- [189] Thomas T. Norton and John T. Siegwart. Light levels, refractive development, and myopia a speculative review. *Experimental Eye Research*, 114:48 57, 2013. Josh Wallman Special Tribute Edition.
- [190] The National Optical Astronomy Observatory. Recommended light levels, 2015.
- [191] Australian Bureau of Statistics. National health survey: First results, 2017-18. arthritis and osteoporosis. Australian Government, 2018.
- [192] João Oliveira, Tiago Guerreiro, Hugo Nicolau, Joaquim Jorge, and Daniel Gonçalves. Blind people and mobile touch-based text-entry: Acknowledging the need for different flavors. In *The Proceedings of the 13th International* ACM SIGACCESS Conference on Computers and Accessibility, ASSETS '11, pages 179–186, New York, NY, USA, 2011. ACM.
- [193] João Oliveira, Tiago Guerreiro, Hugo Nicolau, Joaquim Jorge, and Daniel Gonçalves. Brailletype: Unleashing braille over touch screen mobile phones. In Pedro Campos, Nicholas Graham, Joaquim Jorge, Nuno Nunes, Philippe Palanque, and Marco Winckler, editors, *Human-Computer Interaction – IN-TERACT 2011*, pages 100–107, Berlin, Heidelberg, 2011. Springer Berlin

Heidelberg.

- [194] Eamonn O'Neill, Manasawee Kaenampornpan, Vassilis Kostakos, Andrew Warr, and Dawn Woodgate. Can we do without guis? Gesture and speech interaction with a patient information system. *Personal Ubiquitous Comput.*, 10(5):269–283, July 2006.
- [195] World Health Organisation. Vision impairment and blindness, Oct 2019.
- [196] Antti Oulasvirta. The fragmentation of attention in mobile interaction, and what to do with it. *Interactions*, 12(6):16–18, November 2005.
- [197] Antti Oulasvirta, Sakari Tamminen, Virpi Roto, and Jaana Kuorelahti. Interaction in 4-second bursts: The fragmented nature of attentional resources in mobile hci. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '05, page 919–928, New York, NY, USA, 2005. Association for Computing Machinery.
- [198] Aart Overeem, R. Robinson, Hidde Leijnse, Gert-Jan Steeneveld, B. K.
   P. Horn, and Remko Uijlenhoet. Crowdsourcing urban air temperatures from smartphone battery temperatures. *Geophysical Research Letters*, 40(15):4081–4085, 2013.
- [199] Leysia Palen. Mobile telephony in a connected life. *Communications of the ACM*, 45(3):78–82, 2002.
- [200] Phil Parette and Marcia Scherer. Assistive technology use and stigma. *Education and Training in Developmental Disabilities*, 39(3):217–226, 2004.
- [201] Pekka Parhi, Amy K. Karlson, and Benjamin B. Bederson. Target size study for one-handed thumb use on small touchscreen devices. In *Proceedings* of the 8th Conference on Human-computer Interaction with Mobile Devices and Services, MobileHCI '06, pages 203–210, New York, NY, USA, 2006. ACM.
- [202] Andy Park. Brain surgery without surgery: Incisionless procedure helping fix uncontrollable tremors. *ABC News*, Jan 2019.
- [203] Yong S. Park, Sung H. Han, Jaehyun Park, and Youngseok Cho. Touch key design for target selection on a mobile phone. In *Proceedings of the 10th International Conference on Human Computer Interaction with Mobile Devices and Services*, MobileHCI '08, pages 423–426, New York, NY, USA, 2008. ACM.
- [204] Kurt Partridge, Saurav Chatterjee, Vibha Sazawal, Gaetano Borriello, and Roy Want. Tilttype: Accelerometer-supported text entry for very small devices. In *Proceedings of the 15th Annual ACM Symposium on User Interface Software and Technology*, UIST '02, pages 201–204, New York, NY, USA, 2002. ACM.
- [205] Jessica D. Payne, Lynn Nadel, John J.B. Allen, Kevin G.F. Thomas, and W. Jake Jacobs. The effects of experimentally induced stress on false recognition. *Memory*, 10(1):1–6, 2002. PMID: 11747571.
- [206] Keith B. Perry and Juan Pablo Hourcade. Evaluating one handed thumb tapping on mobile touchscreen devices. In *Proceedings of Graphics Interface* 2008, GI '08, pages 57–64, Toronto, Ont., Canada, Canada, 2008. Canadian Information Processing Society.
- [207] Ken Pfeuffer, Jason Alexander, Ming Ki Chong, and Hans Gellersen. Gazetouch: Combining gaze with multi-touch for interaction on the same surface. In *Proceedings of the 27th Annual ACM Symposium on User Interface Software and Technology*, UIST '14, page 509–518, New York, NY, USA, 2014. Association for Computing Machinery.
- [208] Martin Pielot and Susanne Boll. Tactile wayfinder: Comparison of tactile waypoint navigation with commercial pedestrian navigation systems. In Patrik Floréen, Antonio Krüger, and Mirjana Spasojevic, editors, *Pervasive Computing*, pages 76–93, Berlin, Heidelberg, 2010. Springer Berlin Heidelberg.
- [209] Martin Pielot, Niels Henze, and Susanne Boll. Supporting map-based wayfinding with tactile cues. In *Proceedings of the 11th International Conference on Human-Computer Interaction with Mobile Devices and Services*, Mobile-HCI '09, pages 23:1–23:10, New York, NY, USA, 2009. ACM.
- [210] Martin Pielot, Anastasia Kazakova, Tobias Hesselmann, Wilko Heuten, and Susanne Boll. Pocketmenu: Non-visual menus for touch screen devices. In Proceedings of the 14th International Conference on Human-computer Interaction with Mobile Devices and Services, MobileHCI '12, pages 327–330, New York, NY, USA, 2012. ACM.
- [211] Martin Pielot, Benjamin Poppinga, Wilko Heuten, and Susanne Boll. 6th senses for everyone!: The value of multimodal feedback in handheld navigation aids. In *Proceedings of the 13th International Conference on Multimodal*

Interfaces, ICMI '11, pages 65-72, New York, NY, USA, 2011. ACM.

- [212] Martin Pielot, Benjamin Poppinga, Wilko Heuten, and Susanne Boll. A tactile compass for eyes-free pedestrian navigation. In Pedro Campos, Nicholas Graham, Joaquim Jorge, Nuno Nunes, Philippe Palanque, and Marco Winckler, editors, *Human-Computer Interaction – INTERACT 2011*, pages 640–656, Berlin, Heidelberg, 2011. Springer Berlin Heidelberg.
- [213] Martin Pielot, Benjamin Poppinga, Wilko Heuten, and Susanne Boll. Pocketnavigator: Studying tactile navigation systems in-situ. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '12, pages 3131–3140, New York, NY, USA, 2012. ACM.
- [214] Antti Pirhonen, Stephen Brewster, and Christopher Holguin. Gestural and audio metaphors as a means of control for mobile devices. In *Proceedings* of the SIGCHI Conference on Human Factors in Computing Systems, CHI '02, pages 291–298, New York, NY, USA, 2002. ACM.
- [215] Katrin Plaumann, Milos Babic, Tobias Drey, Witali Hepting, Daniel Stooss, and Enrico Rukzio. Improving input accuracy on smartphones for persons who are affected by tremor using motion sensors. *Proceedings of the ACM on Interactive, Mobile, Wearable and Ubiquitous Technologies*, 1(4):156, 2018.
- [216] Benjamin Poppinga, Wilko Heuten, and Susanne Boll. Navmem explorer: An orientation aid for people with mild cognitive impairments. In *Proceedings of the 8th International Conference on Pervasive Computing Technologies for Healthcare*, PervasiveHealth '14, page 178–181, Brussels, BEL, 2014. ICST (Institute for Computer Sciences, Social-Informatics and Telecommunications Engineering).
- [217] Huimin Qian, Ravi Kuber, and Andrew Sears. Developing tactile icons to support mobile users with situationally-induced impairments and disabilities. In *Proceedings of the 15th International ACM SIGACCESS Conference* on Computers and Accessibility, ASSETS '13, pages 47:1–47:2, New York, NY, USA, 2013. ACM.
- [218] Philip Quinn, Seungyon Claire Lee, Melissa Barnhart, and Shumin Zhai. Active edge: Designing squeeze gestures for the google pixel 2. In *Proceed-ings of the 2019 CHI Conference on Human Factors in Computing Systems*, CHI '19, New York, NY, USA, 2019. Association for Computing Machinery.
- [219] Ahmad Rahmati, Clayton Shepard, and Lin Zhong. Noshake: Content sta-

bilization for shaking screens of mobile devices. In 2009 IEEE International Conference on Pervasive Computing and Communications, pages 1–6, March 2009.

- [220] Katharina Reinecke, David R. Flatla, and Christopher Brooks. Enabling designers to foresee which colors users cannot see. In *Proceedings of the* 2016 CHI Conference on Human Factors in Computing Systems, CHI '16, pages 2693–2704, New York, NY, USA, 2016. ACM.
- [221] Tiago Reis, Luís Carriço, and Carlos Duarte. Mobile interaction: Automatically adapting audio output to users and contexts on communication and media control scenarios. In Constantine Stephanidis, editor, Universal Access in Human-Computer Interaction. Intelligent and Ubiquitous Interaction Environments, pages 384–393, Berlin, Heidelberg, 2009. Springer Berlin Heidelberg.
- [222] Shyam Reyal, Shumin Zhai, and Per Ola Kristensson. Performance and user experience of touchscreen and gesture keyboards in a lab setting and in the wild. In *Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems*, CHI '15, pages 679–688, New York, NY, USA, 2015. ACM.
- [223] Vaspol Ruamviboonsuk, Shiri Azenkot, and Richard E. Ladner. Tapulator: A non-visual calculator using natural prefix-free codes. In *Proceedings of the* 14th International ACM SIGACCESS Conference on Computers and Accessibility, ASSETS '12, pages 221–222, New York, NY, USA, 2012. ACM.
- [224] Dmitry Rudchenko, Tim Paek, and Eric Badger. Text text revolution: A game that improves text entry on mobile touchscreen keyboards. In Kent Lyons, Jeffrey Hightower, and Elaine M. Huang, editors, *Pervasive Computing*, pages 206–213, Berlin, Heidelberg, 2011. Springer Berlin Heidelberg.
- [225] T.A. Saifee, P. Kassavetis, L. Drougkas, G. Roussos, I. Pareés, P. Schwingenschuh, P. Katschnig, K.P. Bhatia, J.C. Rothwell, and M.J. Edwards. Diagnosis of psychogenic tremor using a smartphone. In *Movement disorders*, volume 27, pages S419–S420. WILEY-BLACKWELL, 2012.
- [226] Pierre Salamé and Alan Baddeley. Disruption of short-term memory by unattended speech: Implications for the structure of working memory. *Journal of Verbal Learning and Verbal Behavior*, 21(2):150 – 164, 1982.
- [227] Jaime Sánchez and Fernando Aguayo. Mobile messenger for the blind. In

Constantine Stephanidis and Michael Pieper, editors, *Universal Access in Ambient Intelligence Environments*, pages 369–385, Berlin, Heidelberg, 2007. Springer Berlin Heidelberg.

- [228] Jaime Sánchez and Eduardo Maureira. Subway mobility assistance tools for blind users. In Constantine Stephanidis and Michael Pieper, editors, *Universal Access in Ambient Intelligence Environments*, pages 386–404, Berlin, Heidelberg, 2007. Springer Berlin Heidelberg.
- [229] Frode Eika Sandnes and Hua-Li Jian. Pair-wise variability index: Evaluating the cognitive difficulty of using mobile text entry systems. In Stephen Brewster and Mark Dunlop, editors, *Mobile Human-Computer Interaction -MobileHCI 2004*, pages 347–350, Berlin, Heidelberg, 2004. Springer Berlin Heidelberg.
- [230] Zhanna Sarsenbayeva, Denzil Ferreira, Niels van Berkel, Chu Luo, Mikko Vaisanen, Vassilis Kostakos, and Jorge Goncalves. Vision-based happiness inference: A feasibility case-study. In *Proceedings of the 2017 ACM International Joint Conference on Pervasive and Ubiquitous Computing and Proceedings of the 2017 ACM International Symposium on Wearable Computers*, UbiComp '17, page 494–499, New York, NY, USA, 2017. Association for Computing Machinery.
- [231] Zhanna Sarsenbayeva, Jorge Goncalves, Juan García, Simon Klakegg, Sirkka Rissanen, Hannu Rintamäki, Jari Hannu, and Vassilis Kostakos. Situational impairments to mobile interaction in cold environments. In Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing, UbiComp '16, pages 85–96, New York, NY, USA, 2016. ACM.
- [232] Zhanna Sarsenbayeva, Gabriele Marini, Niels van Berkel, Chu Luo, Weiwei Jiang, Kangning Yang, Greg Wadley, Tilman Dingler, Vassilis Kostakos, and Jorge Goncalves. Does smartphone use drive our emotions or vice versa? A causal analysis. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '20, pages 36:1–36:15, New York, NY, USA, 2020. Association for Computing Machinery.
- [233] Zhanna Sarsenbayeva, Niels van Berkel, Danula Hettiachchi, Weiwei Jiang, Tilman Dingler, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. Measuring the effects of stress on mobile interaction. *Proc. ACM Interact.*

*Mob. Wearable Ubiquitous Technol.*, 3(1), March 2019.

- [234] Zhanna Sarsenbayeva, Niels van Berkel, Weiwei Jiang, Danula Hettiachchi, Vassilis Kostakos, and Jorge Goncalves. Effect of ambient light on mobile interaction. In David Lamas, Fernando Loizides, Lennart Nacke, Helen Petrie, Marco Winckler, and Panayiotis Zaphiris, editors, *Human-Computer Interaction – INTERACT 2019*, pages 465–475. Springer International Publishing, 2019.
- [235] Zhanna Sarsenbayeva, Niels van Berkel, Chu Luo, Vassilis Kostakos, and Jorge Goncalves. Challenges of situational impairments during interaction with mobile devices. In Proc. of the 29th Australian Conference on Computer-Human Interaction, pages 477–481. ACM, 2017.
- [236] Zhanna Sarsenbayeva, Niels van Berkel, Eduardo Velloso, Vassilis Kostakos, and Jorge Goncalves. Effect of distinct ambient noise types on mobile interaction. *Proc. of the ACM Interact. Mob. Wearable Ubiquitous Technol.*, 2(3), 2018.
- [237] Zhanna Sarsenbayeva, Niels van Berkel, Aku Visuri, Sirkka Rissanen, Hannu Rintamaki, Vassilis Kostakos, and Jorge Goncalves. Sensing coldinduced situational impairments in mobile interaction using battery temperature. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.*, 1(3), September 2017.
- [238] Vibha Sazawal, Roy Want, and Gaetano Borriello. The unigesture approach one-handed text entry for small devices. In Fabio Paternò, editor, *Human Computer Interaction with Mobile Devices*, pages 256–270, Berlin, Heidelberg, 2002. Springer Berlin Heidelberg.
- [239] Bastian Schildbach and Enrico Rukzio. Investigating selection and reading performance on a mobile phone while walking. In *Proceedings of the 12th International Conference on Human Computer Interaction with Mobile Devices and Services*, MobileHCI '10, pages 93–102, New York, NY, USA, 2010. ACM.
- [240] Albrecht Schmidt, Kofi Asante Aidoo, Antti Takaluoma, Urpo Tuomela, Kristof Van Laerhoven, and Walter Van de Velde. Advanced interaction in context. In Hans-W. Gellersen, editor, *Handheld and Ubiquitous Computing*, pages 89–101, Berlin, Heidelberg, 1999. Springer Berlin Heidelberg.
- [241] Stacey D. Scott, M. Sheelagh T. Carpendale, and Kori M. Inkpen. Territoriality in collaborative tabletop workspaces. In *Proceedings of the* 2004

ACM Conference on Computer Supported Cooperative Work, CSCW '04, page 294–303, New York, NY, USA, 2004. Association for Computing Machinery.

- [242] Andrew Sears, Min Lin, Julie Jacko, and Yan Xiao. When computers fade: Pervasive computing and situationally-induced impairments and disabilities. In *HCI International*, volume 2, pages 1298–1302, 2003.
- [243] Andrew Sears and Mark Young. The human-computer interaction handbook. chapter Physical Disabilities and Computing Technologies: An Analysis of Impairments, pages 482–503. L. Erlbaum Associates Inc., Hillsdale, NJ, USA, 2003.
- [244] Kristen Shinohara and Jacob O. Wobbrock. In the shadow of misperception: Assistive technology use and social interactions. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '11, pages 705–714, New York, NY, USA, 2011. ACM.
- [245] Kristen Shinohara, Jacob O. Wobbrock, and Wanda Pratt. Incorporating social factors in accessible design. In *Proceedings of the 20th International* ACM SIGACCESS Conference on Computers and Accessibility, ASSETS '18, pages 149–160, New York, NY, USA, 2018. ACM.
- [246] Katie A. Siek, Yvonne Rogers, and Kay H. Connelly. Fat finger worries: How older and younger users physically interact with pdas. In Maria Francesca Costabile and Fabio Paternò, editors, *Human-Computer Interaction - IN-TERACT 2005*, pages 267–280, Berlin, Heidelberg, 2005. Springer Berlin Heidelberg.
- [247] Timothy Sohn, Kevin A. Li, William G. Griswold, and James D. Hollan. A diary study of mobile information needs. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '08, pages 433–442, New York, NY, USA, 2008. ACM.
- [248] R. William Soukoreff and I. Scott MacKenzie. Metrics for text entry research: An evaluation of msd and kspc, and a new unified error metric. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '03, pages 113–120, New York, NY, USA, 2003. ACM.
- [249] R. William Soukoreff and I. Scott MacKenzie. Metrics for text entry research: An evaluation of msd and kspc, and a new unified error metric. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '03, page 113–120, New York, NY, USA, 2003. Association for Computing

Machinery.

- [250] Charles D. Spielberger. *State-Trait Anxiety Inventory*, pages 1–1. American Cancer Society, 2010.
- [251] Mark A. Staal. Stress, cognition, and human performance: A literature review and conceptual framework. 2004.
- [252] X. Su, H. Tong, and P. Ji. Activity recognition with smartphone sensors. *Tsinghua Science and Technology*, 19(3):235–249, June 2014.
- [253] Ke Sun, Chun Yu, Weinan Shi, Lan Liu, and Yuanchun Shi. Lip-interact: Improving mobile device interaction with silent speech commands. In Proceedings of the 31st Annual ACM Symposium on User Interface Software and Technology, UIST '18, page 581–593, New York, NY, USA, 2018. Association for Computing Machinery.
- [254] Anthony Tang, Melanie Tory, Barry Po, Petra Neumann, and Sheelagh Carpendale. Collaborative coupling over tabletop displays. In *Proceedings* of the SIGCHI Conference on Human Factors in Computing Systems, CHI '06, page 1181–1190, New York, NY, USA, 2006. Association for Computing Machinery.
- [255] Garreth W. Tigwell, David R. Flatla, and Rachel Menzies. It's not just the light: Understanding the factors causing situational visual impairments during mobile interaction. In *Proceedings of the 10th Nordic Conference on Human-Computer Interaction*, NordiCHI '18, page 338–351, New York, NY, USA, 2018. Association for Computing Machinery.
- [256] Garreth W. Tigwell, Zhanna Sarsenbayeva, Benjamin M. Gorman, David R. Flatla, Jorge Goncalves, Yeliz Yesilada, and Jacob O. Wobbrock. Addressing the challenges of situationally-induced impairments and disabilities in mobile interaction. In *Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems*, CHI EA '19, pages W30:1–W30:8, New York, NY, USA, 2019. ACM.
- [257] Shari Trewin, Simeon Keates, and Karyn Moffatt. Developing steady clicks:: A method of cursor assistance for people with motor impairments. In Proceedings of the 8th International ACM SIGACCESS Conference on Computers and Accessibility, Assets '06, pages 26–33, New York, NY, USA, 2006. ACM.
- [258] Niels van Berkel, Chu Luo, Theodoros Anagnostopoulos, Denzil Ferreira,

Jorge Goncalves, Simo Hosio, and Vassilis Kostakos. A systematic assessment of smartphone usage gaps. In *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*, CHI '16, page 4711–4721, New York, NY, USA, 2016. Association for Computing Machinery.

- [259] Gregg C. Vanderheiden. Use of audio-haptic interface techniques to allow nonvisual access to touchscreen appliances. *Proceedings of the Human Factors* and Ergonomics Society Annual Meeting, 40(24):1266–1266, 1996.
- [260] Marynel Vázquez and Aaron Steinfeld. Helping visually impaired users properly aim a camera. In *Proceedings of the 14th International ACM SIGAC-CESS Conference on Computers and Accessibility*, ASSETS '12, pages 95–102, New York, NY, USA, 2012. ACM.
- [261] N. Venetjoki, A. Kaarlela-Tuomaala, E. Keskinen, and V. Hongisto. The effect of speech and speech intelligibility on task performance. *Ergonomics*, 49(11):1068–1091, 2006. PMID: 16950722.
- [262] Aku Visuri, Zhanna Sarsenbayeva, Jorge Goncalves, Evangelos Karapanos, and Simon Jones. Impact of mood changes on application selection. In Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing: Adjunct, UbiComp '16, page 535–540, New York, NY, USA, 2016. Association for Computing Machinery.
- [263] Aku Visuri, Zhanna Sarsenbayeva, Niels van Berkel, Jorge Goncalves, Reza Rawassizadeh, Vassilis Kostakos, and Denzil Ferreira. Quantifying sources and types of smartwatch usage sessions. In *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems*, CHI '17, page 3569–3581, New York, NY, USA, 2017. Association for Computing Machinery.
- [264] Chaofan Wang, Zhanna Sarsenbayeva, Xiuge Chen, Tilman Dingler, Jorge Goncalves, and Vassilis Kostakos. Accurate measurement of handwash quality using sensor armbands: Instrument validation study. *JMIR Mhealth Uhealth*, 8(3):e17001, Mar 2020.
- [265] Johan Wannenburg and Reza Malekian. Physical activity recognition from smartphone accelerometer data for user context awareness sensing. *IEEE Transactions on Systems, Man, and Cybernetics: Systems*, 47(12):3142–3149, Dec 2017.
- [266] Tetsuya Watanabe, Manabi Miyagi, Kazunori Minatani, and Hideji Nagaoka. A survey on the use of mobile phones by visually impaired persons

in japan. In Klaus Miesenberger, Joachim Klaus, Wolfgang Zagler, and Arthur Karshmer, editors, *Computers Helping People with Special Needs*, pages 1081–1084, Berlin, Heidelberg, 2008. Springer Berlin Heidelberg.

- [267] Daniel Wigdor and Ravin Balakrishnan. Tilttext: Using tilt for text input to mobile phones. In *Proceedings of the 16th Annual ACM Symposium on User Interface Software and Technology*, UIST '03, pages 81–90, New York, NY, USA, 2003. ACM.
- [268] Jacob O. Wobbrock. The future of mobile device research in hci. In CHI 2006 workshop proceedings: what is the next generation of human-computer interaction, pages 131–134, 2006.
- [269] Jacob O. Wobbrock. Situationally aware mobile devices for overcoming situational impairments. In *Proceedings of the ACM SIGCHI Symposium on Engineering Interactive Computing Systems*, EICS '19, New York, NY, USA, 2019. Association for Computing Machinery.
- [270] Jacob O. Wobbrock. Situationally-Induced Impairments and Disabilities, pages 59–92. Springer London, London, 2019.
- [271] Jacob O. Wobbrock, Duen Horng Chau, and Brad A. Myers. An alternative to push, press, and tap-tap-tap: Gesturing on an isometric joystick for mobile phone text entry. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '07, pages 667–676, New York, NY, USA, 2007. ACM.
- [272] Jacob O. Wobbrock and Krzysztof Z. Gajos. A comparison of area pointing and goal crossing for people with and without motor impairments. In *Proceedings of the 9th International ACM SIGACCESS Conference on Computers* and Accessibility, Assets '07, pages 3–10, New York, NY, USA, 2007. ACM.
- [273] Jacob O. Wobbrock, Krzysztof Z. Gajos, Shaun K. Kane, and Gregg C. Vanderheiden. Ability-based design. *Commun. ACM*, 61(6):62–71, May 2018.
- [274] Jacob O. Wobbrock, Shaun K. Kane, Krzysztof Z. Gajos, Susumu Harada, and Jon Froehlich. Ability-based design: Concept, principles and examples. ACM Trans. Access. Comput., 3(3):9:1–9:27, April 2011.
- [275] Jacob O. Wobbrock, Brad A. Myers, and John A. Kembel. Edgewrite: A stylus-based text entry method designed for high accuracy and stability of

motion. In *Proceedings of the 16th Annual ACM Symposium on User Interface Software and Technology*, UIST '03, pages 61–70, New York, NY, USA, 2003. ACM.

- [276] T. Yamabe and K. Takahashi. Experiments in mobile user interface adaptation for walking users. In *The 2007 International Conference on Intelligent Pervasive Computing (IPC 2007)*, pages 280–284, Oct 2007.
- [277] Koji Yatani and Khai Nhut Truong. Semfeel: A user interface with semantic tactile feedback for mobile touch-screen devices. In *Proceedings of the 22Nd Annual ACM Symposium on User Interface Software and Technology*, UIST '09, pages 111–120, New York, NY, USA, 2009. ACM.
- [278] Hanlu Ye, Meethu Malu, Uran Oh, and Leah Findlater. Current and future mobile and wearable device use by people with visual impairments. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '14, pages 3123–3132, New York, NY, USA, 2014. ACM.
- [279] Yeliz Yesilada, Simon Harper, Tianyi Chen, and Shari Trewin. Smalldevice users situationally impaired by input. *Computers in Human Behavior*, 26(3):427 – 435, 2010.
- [280] Ji Soo Yi, Young Sang Choi, Julie A. Jacko, and Andrew Sears. Context awareness via a single device-attached accelerometer during mobile computing. In *Proceedings of the 7th International Conference on Human Computer Interaction with Mobile Devices & Amp; Services*, MobileHCI '05, pages 303– 306, New York, NY, USA, 2005. ACM.
- [281] A. H. Young, B. J. Sahakian, T. W. Robbins, and P. J. Cowen. The effects of chronic administration of hydrocortisone on cognitive function in normal male volunteers. *Psychopharmacology*, 145(3):260–266, Aug 1999.
- [282] Shengdong Zhao, Pierre Dragicevic, Mark Chignell, Ravin Balakrishnan, and Patrick Baudisch. Earpod: Eyes-free menu selection using touch input and reactive audio feedback. In *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, CHI '07, pages 1395–1404, New York, NY, USA, 2007. ACM.

## **University Library**



## A gateway to Melbourne's research publications

Minerva Access is the Institutional Repository of The University of Melbourne

Author/s: Sarsenbayeva, Zhanna

Title:

Quantifying the Effects of Situationally-Induced Impairments and Disabilities on Mobile Interaction

Date: 2020

Persistent Link: http://hdl.handle.net/11343/239076

File Description: Final thesis file

## Terms and Conditions:

Terms and Conditions: Copyright in works deposited in Minerva Access is retained by the copyright owner. The work may not be altered without permission from the copyright owner. Readers may only download, print and save electronic copies of whole works for their own personal non-commercial use. Any use that exceeds these limits requires permission from the copyright owner. Attribution is essential when quoting or paraphrasing from these works.